

University of Pennsylvania ScholarlyCommons

Publicly Accessible Penn Dissertations

2016

Resource-Aware Design Of Wireless Control Systems

Konstantinos Gatsis University of Pennsylvania, kgatsis@seas.upenn.edu

Follow this and additional works at: https://repository.upenn.edu/edissertations Part of the <u>Electrical and Electronics Commons</u>

Recommended Citation

Gatsis, Konstantinos, "Resource-Aware Design Of Wireless Control Systems" (2016). *Publicly Accessible Penn Dissertations*. 2300. https://repository.upenn.edu/edissertations/2300

This paper is posted at ScholarlyCommons. https://repository.upenn.edu/edissertations/2300 For more information, please contact repository@pobox.upenn.edu.

Resource-Aware Design Of Wireless Control Systems

Abstract

This work is motivated by modern monitoring and control infrastructures appearing in smart homes, urban environments, and industrial plants. These systems are characterized by multiple sensor and actuator devices at different physical locations, communicating wirelessly with each other. Desired monitoring and control performance requires efficient wireless communication, as the more information the sensors convey the more precise actuation becomes. However wireless communication is constrained by the inherent uncertainty of the wireless medium as well as resource limitations at the devices, e.g., limited power resources. The increased number of wireless devices in such environments further necessitates the management of the shared wireless spectrum with direct account of control performance. To address these challenges, the goal of this work is to provide control-aware and resource-aware communication policies. This is first examined in the fundamental problem of allocating transmit power resources for wireless closed loop control. Opportunistic online adaptation of power to plant and wireless channel conditions is shown to be essential in achieving the optimal tradeoff between control performance and power utilization. Optimal structural properties of channel access mechanisms are also considered for the problem of guaranteeing multiple control performance requirements over a shared wireless medium. This includes scheduling mechanisms implemented by central authorities, as well as decentralized mechanisms implemented independently by the wireless devices with emerging wireless interferences. Again the mechanisms exhibit an opportunistic adaptation to varying wireless channel conditions, especially designed to explore the tradeoffs between different communication links and meet control performance requirements. The structural characterization is augmented with tractable optimization algorithms to compute these channel access mechanisms. Finally, as control is naturally a dynamic task that requires a long term planning, appropriate dynamic algorithms adapting to the varying control system states are examined. Besides adapting dynamically, the proposed algorithms provide guarantees about long term control performance and resource utilization by construction.

Degree Type

Dissertation

Degree Name Doctor of Philosophy (PhD)

Graduate Group Electrical & Systems Engineering

First Advisor George J. Pappas

Keywords

Control systems, Optimal control, Resource allocation, Wireless communications

Subject Categories Electrical and Electronics

RESOURCE-AWARE DESIGN OF WIRELESS CONTROL SYSTEMS

Konstantinos Gatsis

A DISSERTATION

in

Electrical and Systems Engineering

Presented to the Faculties of the University of Pennsylvania

in

Partial Fulfillment of the Requirements for the

Degree of Doctor of Philosophy

2016

Supervisor of Dissertation

George J. Pappas, Professor of Electrical and Systems Engineering

Graduate Group Chairperson

Alejandro Ribeiro, Associate Professor of Electrical and Systems Engineering

Dissertation Committee

Babak Hassibi, Professor of Electrical Engineering, California Institute of Technology Ali Jadbabaie, Professor of Electrical and Systems Engineering

RESOURCE-AWARE DESIGN OF WIRELESS CONTROL SYSTEMS

© COPYRIGHT

2016

Konstantinos Gatsis

ACKNOWLEDGEMENT

My supervisor Prof. George J. Pappas has supported me by creating the environment for fruitful research during my graduate studies. His help has been indispensable and I am grateful. Prof. Alejandro Ribeiro's guidance and advice, technical and academic, has been influential during all my years at Penn and I would like to express my gratitude. I would also like to thank my committee members for their time invested and constructive comments on my work. I would like to thank all my lab and department colleagues for making studying at Penn such a great experience.

I would like to acknowledge the support of "Andreas Mentzelopoulos Scholarships for the University of Patras".

Finally, I would like to thank my family for all the support all these years, and I would like to especially thank my wife Anita for making our life in Philadelphia so happy and memorable.

ABSTRACT

RESOURCE-AWARE DESIGN OF WIRELESS CONTROL SYSTEMS

Konstantinos Gatsis

George J. Pappas

This work is motivated by modern monitoring and control infrastructures appearing in smart homes, urban environments, and industrial plants. These systems are characterized by multiple sensor and actuator devices at different physical locations, communicating wirelessly with each other. Desired monitoring and control performance requires efficient wireless communication, as the more information the sensors convey the more precise actuation becomes. However wireless communication is constrained by the inherent uncertainty of the wireless medium as well as resource limitations at the devices, e.g., limited power resources. The increased number of wireless devices in such environments further necessitates the management of the shared wireless spectrum with direct account of control performance. To address these challenges, the goal of this work is to provide control-aware and resource-aware communication policies. This is first examined in the fundamental problem of allocating transmit power resources for wireless closed loop control. Opportunistic online adaptation of power to plant and wireless channel conditions is shown to be essential in achieving the optimal tradeoff between control performance and power utilization. Optimal structural properties of channel access mechanisms are also considered for the problem of guaranteeing multiple control performance requirements over a shared wireless medium. This includes scheduling mechanisms implemented by central authorities, as well as decentralized mechanisms implemented independently by the wireless devices with emerging wireless interferences. Again the mechanisms exhibit an opportunistic adaptation to varying wireless channel conditions, especially designed to explore the tradeoffs between different communication links and meet control performance requirements. The structural characterization is augmented with tractable optimization algorithms to compute these channel access mechanisms. Finally, as control is naturally a dynamic task that requires a long term planning, appropriate dynamic algorithms adapting to the varying control system states are examined. Besides adapting dynamically, the proposed algorithms provide guarantees about long term control performance and resource utilization by construction.

TABLE OF CONTENTS

ACKNOWLEDGEMENT	ii
ABSTRACT in	v
LIST OF TABLES vi	ii
LIST OF ILLUSTRATIONS	1
CHAPTER 1: Introduction	1
1.1 Related Work	2 5
CHAPTER 2 : Resource-aware Wireless Control Systems	8
2.1 Control System Model	8
2.2 Wireless Channel Model	3
CHAPTER 3 : Optimal Power Management for Wireless Sensor-Actuator Systems 2	0
3.1 Problem Description	0
3.2 Information structure	3
3.3 Separation of designs	6
3.4 Optimal Communication Policy	0
3.5 Proofs	4
CHAPTER 4 : Opportunistic Control over Multiple Access Channels 44	3
4.1 Problem Description	3
4.2 Optimal scheduling and power allocation	8
4.3 Online scheduling and power allocation	3
4.4 Numerical Simulations	8
4.5 Proofs	1
CHAPTER 5: Random Access Design for Wireless Control Systems	1
5.1 Problem Formulation	1
5.2 Channel-aware Random Access Design	4
5.3 Numerical simulations	7
5.4 Proofs	0
CHAPTER 6: Adaptation to Plant States	7
6.1 Rollout Power Allocation Policies	7
6.2 Rollout Multiple Access Policies	6
6.3 Rollout Random Access Policies	8

6.4	Proofs	 •		 • •	 			 	106
CHAPT	TER 7 : Conclusion	 •	•••	 	 			 	116
7.1	Open Problems and Future Research Directions	 •		 	 • •	•	· •	 	117
APPEN	IDIX	 •		 • •	 	•		 	119
BIBLIO	GRAPHY	 •	•••	 	 			 	122

LIST OF TABLES

TABLE 1 :	System parameters & online transmission rates		60
-----------	---	--	----

LIST OF ILLUSTRATIONS

FIGURE 1 :	Wireless control system architecture. At each time step k the sensor measures and transmits the plant state x_k over a wireless channel to the controller. The selected transmit power p_k and the wireless fading state h_k determine whether the message is successfully received. The goal is to achieve a good trade-off between power consumption and closed loop performance	8
FIGURE 2 :	Complementary error function for FEC and capacity achieving codes. The probability of successful decoding <i>q</i> for a practical FEC code is a sigmoid function of the received SNR = $h p/N_0$, while for a capacity achieving code a threshold value SNR ₀ determines whether a packet is successfully received	14
FIGURE 3 :	Architecture for control over multiple access channels. Independent con- trol systems close their loops by transmitting over the shared wireless medium to a common receiver/access point. Each control system <i>i</i> expe- riences random channel conditions h_i . A centralized scheduler at the ac- cess point observes all channel states and opportunistically decides which system is scheduled to transmit and close the loop.	14
FIGURE 4 :	Random access architecture for <i>m</i> control loops over a shared wireless medium. Each sensor <i>i</i> randomly transmits with probability $\alpha_{i,k}$ at time <i>k</i> to a common access point computing the plant control inputs. If only sensor <i>i</i> transmits, the successful decoding probability depends on local channel conditions $h_{i,k}$. If other sensors transmit at the same time a collision might occur at sensor <i>i</i> 's transmission, rendering <i>i</i> 's packet lost	18
FIGURE 5 :	Wireless control system architecture. A sensor measures the plant and wireless fading channel states x_k , h_k respectively and transmits with power p_k . Messages are successfully decoded at the controller with probability q_k that depends on the channel state h_k and the power p_k . The sensor	
FIGURE 6 :	receives acknowledgments with a one-step delay Equivalent wireless control system architecture. A scheduler decides the successful decoding probability q_k and transmits the state measurement x_k with the required power $p_k = p(h_k, q_k)$. The controller receives the	21
	message with probability q_k	22

FIGURE 7 :	Wireless control system with a restricted information structure. The sensor consists of two blocks. A pre-processor computes the error ε_k given the measurement x_k and the acknowledgment γ_{k-1} . A scheduler decides q_k based on ε_k and the channel state h_k , and transmits x_k with the required power $p_k = p(h_k, q_k)$. The controller receives the message with probability	
FIGURE 8 :	q_k , computes the state estimate \hat{x}_k and provides input u_k to the plant Optimal power allocation for FEC codes with different complementary error functions. The optimal transmit power p^* is plotted as a function of the factor $R(\varepsilon)$ for a fixed channel state h using FEC codes with different q -SNR characteristics. When the q -SNR curve becomes steeper, the optimal power allocation recembles a step function	25
FIGURE 9 :	Optimal decoding probability and power allocation for a FEC code. Color intensity indicates the magnitude of optimal decoding probability q^* and optimal transmit power p^* as functions of the factor $R(\varepsilon)$ and the channel state h .	33
FIGURE 10 :	Architecture for control over multiple access channels. Independent con- trol systems close their loops by transmitting over the shared wireless medium to a common receiver/access point. Each control system <i>i</i> expe- riences random channel conditions h_i . A centralized scheduler at the ac- cess point observes all channel states and opportunistically decides which system is scheduled to transmit and close the loop.	44
FIGURE 11 :	Optimal channel-aware scheduling for the example presented in Section 4.4. System 1 has a harder Lyapunov decrease rate requirement and is scheduled to transmit for most observed channel states h_1 , h_2 . System 2 is scheduled only if its channel conditions h_2 are much more favorable that those of system 1. When both channels are very adverse, systems select zero transmit powers so scheduling is irrelevant.	57
FIGURE 12 :	Optimal channel-aware power allocation for the example presented in Sec- tion 4.4. Under adverse channel conditions systems do not transmit. The channel threshold for transmission for system 1 is lower than that of sys- tem 2 because the former has a higher Lyapunov decrease rate require-	
FIGURE 13 :	ment. System 1 also requires higher transmit power	59
FIGURE 14 :	around the optimal values μ^*	61
	decrease rates.	61

FIGURE 15 :	Random access architecture for <i>m</i> control loops over a shared wireless medium. Each sensor <i>i</i> randomly transmits with probability $\alpha_{i,k}$ at time	
	<i>k</i> to a common access point computing the plant control inputs. If only	
	sensor <i>i</i> transmits, the successful decoding probability depends on local	
	channel conditions h_{ik} . If other sensors transmit at the same time a colli-	
	sion might occur at sensor i 's transmission, rendering i 's packet lost	72
FIGURE 16 :	Evolution of dual variables during the optimization algorithm. The ele-	
	where $v(t)$ converge to the optimal values v required to	78
FICURE 17.	Channel thresholds corresponding to the access policies selected by the	70
FIGURE 17.	entimization algorithm. The channel thresholds for both songers converge	
	to their entired veloce in the limit. Concer 1 has a lower threshold is	
	to their optimal values in the limit. Sensor 1 has a lower threshold, i.e.,	
	transmits more often, since it is required to guarantee control performance	70
FICUDE 10	for a more demanding (unstable) plant.	79
FIGURE 18 :	Evolution of control systems using the optimal random access policies.	
	Both systems remain stable despite collisions and packet drops. Also their	
	long run average quadratic cost converges to the same value, since be	
	design both systems were required to have the same control performance.	79
FIGURE 19 :	The control and communication performance of a selected reference pol-	
	icy is depicted, along with the selected tradeoff line $\nu \ge 0$ between the two	
	objectives. By Theorem 6.1 the control and communication performance	
	resulting from the rollout policy is guaranteed to lie left of the selected	
	tradeoff (shaded area). The actual rollout performance for this example is	
	also depicted. The set of reference policies with constant power allocation	
	$\tilde{p}(.)$ are also depicted for comparison	91
FIGURE 20 :	Contour plot of the indefinite value-of-transmission function example	94
FIGURE 21 :	Contour plot of the positive definite value-of-transmission function example.	95
FIGURE 22 :	Control performance and average power consumption of proposed rollout	
	policies, in comparison with the reference policies	95
FIGURE 23 :	Schedule of selected transmit powers by the rollout algorithm	96
FIGURE 24 :	Simulation of reference random access policies and proposed state-aware	
	random access policies. By opportunistically adapting to system states the	
	proposed policies exhibit a significantly improved control performance.	104
FIGURE 25 :	Reference random access. Green stars indicate successful transmissions	
	for each system. Red circles indicate collisions between the two systems.	
	Successful transmissions happen in a random fashion independent of the	
	system states.	105
FIGURE 26 :	Constrained rollout random access. Green stars indicate successful trans-	
	missions for each system. Red circles indicate collisions between the two	
	systems. Successful transmissions typically occur when the systems states	
	are large, meaning that control performance is at stake.	105

Chapter 1: Introduction

Wireless sensors are abundant in modern smart infrastructures, where they are deployed to monitor and control physical processes in our homes, urban environments, and industrial plants. Essential to the development of such systems is wireless communication, allowing the separation between the physical locations where sensing and data collection is performed, and different physical locations where control operations are decided and actuated to the system. To achieve desired monitoring and control performance, efficient and reliable communication of information from sensors to actuators is required. However the latter is constrained by the uncertain stochastic nature of the wireless medium as well as by resource limitations.

Wireless sensors for example are typically battery-operated. The more information a sensor conveys the more precise decision making becomes, but the resulting increase in power consumption at the sensor leads to rapid depletion of its energy resources. Hence tradeoffs between device lifetime and overall system performance emerge. Besides power limitations, as the number of wireless sensors and actuators in these environments increases, a need to share the available wireless medium between these devices arises. Efficient communication requires frequent channel access for each sensor, but if this access is not coordinated wireless interferences arise between transmissions and performance degrades. As a result, the shared wireless medium necessitates mechanisms for maintaining desired levels of performance among the various devices/systems in the environment.

Examples of such systems are ubiquitous. Smart buildings (Nghiem and Pappas (2011); Ma et al. (2012); Oldewurtel et al. (2012)) are instrumented with sensors collecting and transmitting local information to locations where operation decisions are being made. Monitoring physical processes with a distributed or large-scale nature also requires efficient sensor communication, as illustrated in monitoring of, e.g., agricultural applications (Yiming et al. (2007); Ruiz-Garcia et al. (2009); Cardell-Oliver et al. (2004)), contagious processes (Sadilek et al. (2012); Nowzari et al. (2016)), air quality and concentration of chemical substances (Kim et al. (2009); De Vito et al. (2011)), traffic (Horvitz et al. (2012)), or water networks (Whittle et al. (2013)).

Designing wireless sensor-actuator systems with desirable characteristics involves the following challenges:

 Dynamic Physical Processes: The wireless sensors and actuators of the systems considered are deployed to collect measurements and control physical processes that are dynamically and stochastically changing over time. To characterize the performance of such systems it is indispensable to consider metrics that account for operation over a long time horizon. Hence, appropriate metrics of dynamic monitoring or control performance are required.

- Wireless Communication Uncertainties: Due to the intrinsic stochastic behavior of the wireless medium, communication between devices is not always reliable but is subject to randomness and non-determinism. The varying wireless conditions create opportunities or adversities, and their effects on the monitoring and control performance needs to be accounted for. As a result, appropriate wireless communication models which capture this stochastic behavior need to be employed.
- Adaptive Resource Allocation: Static mechanisms for allocating the available communication resources are conservative, as in order to maintain desired performance at all times they have to consider worst operation conditions. In view of the dynamic nature of the considered physical processes, as well as the variability of the wireless medium, the use of dynamic resource allocation mechanisms becomes apparent. By adapting to the system conditions during operation, wireless communication resources can be appropriately allocated only when necessary or only when opportunities arise. The development of such mechanisms that can adapt online but can as well provide long time performance guarantees emerges as a principal research challenge.

1.1. Related Work

The topics considered in this work relate to the area of networked control systems, that is, control systems implemented over communication networks. The related work can be classified with respect to the employed communication and channel models. A widely adopted paradigm, which will also be followed in this work, is that of *packet-based communication* (Hespanha et al. (2007)). In this paradigm communication is organized in packets, i.e., when a sensor collects a plant sample or measurement it transmits the acquired information to the intended receiver in a packet. Packets are assumed to be long enough so that any quantization effects from converting the collected information to, e.g., a sequence of bits, can be ignored.

Early works on packet-based networked control systems ignore the cost of conveying information, i.e., the cost of transmitting packets, and focus their analysis on the performance of control loops when various communication uncertainties are taken into account. A fundamental type of uncertainty arises when transmitting packets over erasure channels which randomly drop some of the packets. If packets are dropped in an independent and identically distributed fashion there is a maximum packet drop rate above which controlling or estimating a plant becomes impossible, i.e., the system becomes unstable. Characterizations of this maximum packet drop rate are examined by Seiler and Sengupta (2001); Sinopoli et al. (2004); Imer et al. (2006); Hespanha et al. (2007); Gupta et al. (2007). A different type of communication uncertainty arises when transmitted packets are received with unknown or random delay. Related characterizations of stability under delays are considered by Walsh et al. (2002); Hespanha et al. (2007); Heemels et al. (2010).

Besides fundamental stability characterizations, considerable attention has been given on applying

appropriate control operations to counteract these communication uncertainties, that is, random packet drops or delays. This entails the design of appropriate state estimators and controllers, e.g., linear feedback and observer gains, as discussed by Seiler and Sengupta (2001); Smith and Seiler (2003); Xu and Hespanha (2005); Imer et al. (2006); Hespanha et al. (2007); Schenato et al. (2007); Gupta et al. (2007). Overall the fact that packets/measurements are sometimes available at the receiver while other times they are not requires an appropriate analysis of the system as a switched or hybrid system.

While in the aforementioned works communication is treated as a constraint or disturbance to the overall closed loop system, more recent efforts consider communication as an active part of the design. The resulting setup typically departs from the classic periodic control paradigm, where the control loop is closed periodically after a fixed amount of time. In particular frameworks such as event-triggered sampling (Astrom and Bernhardsson (2002); Rabi et al. (2012)), event-triggered control (Tabuada (2007); Rabi et al. (2008); Heemels et al. (2012, 2013)), or self-triggered control (Anta and Tabuada (2010); Nowzari and Cortés (2012)) have been proposed to limit the amount of required communication. The underlying concept in these contributions is to prolong the time elapsed between successive sampling or input updates as long as some plant performance criterion is satisfied. Such schemes exhibit in general an average communication rate lower than periodic schemes that attain similar plant performance. However, communication costs are not explicitly accounted for in the triggering design.

Communication costs are explicitly modeled in the context of remote state estimation by Xu and Hespanha (2004); Cogill et al. (2007); Li and Lemmon (2011); Lipsa and Martins (2011); Mesquita et al. (2012). In this framework a sensor measuring the plant state decides at each time step whether to transmit its measurement to an estimator or not, and every transmission incurs a fixed cost. The overall goal is to minimize jointly a penalty on the estimation error and the communication costs aggregated over time. Alternatively a hard constraint on the number of transmissions between the sensor and the estimator is considered by Imer and Başar (2005, 2010); Rabi et al. (2012). The optimal communication is shown to be event-triggered by Xu and Hespanha (2004), similar to, e.g., Tabuada (2007); Rabi et al. (2008), meaning that transmissions are triggered when a measure of the plant state exceeds a threshold. Computing the optimal transmission-triggering sets is in general computationally hard, motivating the development of suboptimal schemes Cogill et al. (2007); Cogill (2009); Li et al. (2010); Antunes and Heemels (2014).

Related contributions consider the joint design of communication policies as well as control input policies, assuming again a fixed cost per transmission (Molin and Hirche (2009); Ramesh et al. (2013)). The problem becomes complex as the policies of the two entities, i.e., the sensor and the controller, have an information structure that is coupled over time. More specifically, the controller requires a local state estimator that is in general dependent on the communication policy. Some structural results for the joint policy design have been recently considered by Nayyar et al. (2013). Alternatively, one can impose special policy structures leading a separation principle (Ramesh et al. (2013)) so that optimal inputs and communication schedules can be found by dynamic

programming for a finite time horizon. Other structural properties of the communication policies to simplify the state estimation are considered by Han et al. (2015).

The above approaches within the packet-based communication model are related to the approach followed in this work, however it is worth noting that different lines of research emerge by other communication or channel models as follows:

- Information-theoretic approaches and Coding: An alternative communication model arises by considering a digital channel with a limited data-rate. In this case to examine the transfer of information from sources to destinations, i.e., from sensors to controllers, information-theoretic approaches are usually followed. A fundamental characterization that arises is the minimimum bit rate under which an unstable system can be stabilized; see, e.g., Nair et al. (2007); Franceschetti and Minero (2014). To achieve the communication of plant states, typically taking values in real vector spaces, an efficient coding and decoding scheme is required and, in contrast to the packet-based paradigm, quantization effects need to be taken into account (Wong and Brockett (1997); Brockett and Liberzon (2000); Tatikonda and Mitter (2004); Yuksel and Başar (2006); Ostrovsky et al. (2009); Sukhavasi and Hassibi (2016)). As the encoder and decoder policies depend on each other but are implemented at different physical locations, the joint design is complex and structural properties are explored by Walrand and Varaiya (1983); Varaiya and Walrand (1983); Borkar and Mitter (1997); Mahajan (2008). Further connections between information theoretic measures and control performance are explored by Martins and Dahleh (2008); Tanaka et al. (2015).
- *Input-ouput models of communication:* Apart from the packet-based communication paradigm and the limited data-rate channel model, communication between a sensor and a controller can also be modeled as an input-output system (Elia (2005)). The uncertainty introduced in the closed loop due to channel randomness is then treated as stochastic model uncertainty. This facilitates controller synthesis using robust control techniques (Braslavsky et al. (2007)). Communication is again not treated as a design variable in this context but a constraint to closed loop control.

Control over Shared Wireless Channels

The problem of sharing a communication medium between different sensor and actuator devices has also received considerable attention in the context of networked control systems. The prevalent approach to this problem is to design a centralized scheduling mechanism with respect to control performance objectives. The scheduling mechanisms usually examined are either static or dynamic. Typical examples of the first type are periodic protocols where the wireless devices transmit in a predefined repeating order, e.g., round-robin. Stability under such protocols can be analyzed by a switched system approach – see, e.g., Zhang et al. (2001); Hespanha et al. (2007); Schenato et al. (2007); Donkers et al. (2011). The problem of designing static schedules suitable for control applications has also been addressed. Periodic sequences leading to stability

(Hristu-Varsakelis (2001)), controllability and observability (Zhang and Hristu-Varsakelis (2006)), or minimizing linear quadratic objectives (Le Ny et al. (2011)) have been proposed. Deriving otherwise optimal scheduling sequences is recognized as a hard combinatorial problem (Meier et al. (1967); Rehbinder and Sanfridson (2004); Gupta et al. (2006)).

The second type of schedulers, the dynamic ones, do not rely on a predefined sequence but access to the communication medium is decided dynamically at each step. The decision typically depends on the current plant/control system states, i.e., informally the subsystem with the largest state discrepancy is scheduled to communicate. Examples can be found in, e.g., Walsh et al. (2002); Hristu-Varsakelis and Kumar (2002); Egerstedt and Wardi (2002); Donkers et al. (2011). Recent efforts have also focused on scheduling for event-based controllers (Cervin and Henningsson (2008); Molin and Hirche (2014)). Another approach, motivated by the problem of scheduling control tasks sharing a computation resource (CPU) rather than a communication medium, is to abstract control performance requirements in the time/frequency domain. Knowing, e.g., how often a task needs resource access to communicate and close the loop, static and dynamic schedules meeting the desired requirements can be obtained using algorithms from real-time scheduling theory (Liu (2000); Branicky et al. (2002)).

In contrast to centralized scheduling, decentralized mechanisms where sensors independently decide access to the shared wireless medium are easier to implement. They do not require predesigned sequences of how sensors access the medium, or a central authority to take scheduling decisions. The drawback however is that *packet collisions* can occur from simultaneously transmitting sensors, resulting in lost packets and control performance degradation. Control under decentralized channel access mechanisms has drawn limited attention in the literature. Comparisons between different medium access mechanisms for networked control systems and the impact of packet collisions have been considered either in numerical simulations (Liu and Goldsmith (2004); Ramesh et al. (2013)) or analytically in simple cases (Rabi et al. (2010); Blind and Allgöwer (2011)). These include random access mechanisms, where each sensor independently and randomly decides whether to transmit, and related Aloha-like schemes, where after a packet collision the involved sensors wait for a random time interval and retransmit. Stability conditions under packet collisions were examined in Zhang (2003); Tabbara and Nesic (2008). Besides closed loop control, optimal remote estimation over collision channels is considered recently in Vasconcelos and Martins (2014).

1.2. Outline and Contributions

The goal of this work is to develop mechanisms for efficient communication and resource allocation in sensor-actuator systems implemented over wireless channels. The topics considered in this context and the related contributions are presented next.

Resource-aware Wireless Control Systems

The packet-based paradigm is adopted in modeling the communication between sensors and actuators. In order to explicitly account for the allocation of transmit power resources however this model is generalized. In particular, the probability of successfully decoding the transmitted packet at the receiver is modeled as a function of the channel conditions (fading) as well as the selected transmit power. This constitutes a generalization of the independent packet success model, as here a transmitting sensor has the ability to directly control the packet success by adjusting the transmit power. This communication model allows us to mathematically formulate the problem of power allocation in wireless control systems. Moreover, it allows us to capture the uncertainties introduced by the inherent stochastic nature of the wireless medium via channel fading. Finally, this communication model is extended to the case where multiple sensors need to communicate over a shared wireless medium (multiple-access channels) and allows for allocation of resources across multiple systems. These contributions are detailed in Chapter 2.

Power Management for Wireless Sensor-Actuator Systems

Under the introduced power-aware channel model we examine a single loop control system, where a sensor transmits information over a wireless fading channel to a controller. We formulate the problem of minimizing jointly the average control performance of the plant as well as the average power consumption at the sensor. We identify a special information structure that allows the decoupling of the two designs, so that the control input policy and the power allocation policy can be designed separately. Then we derive structural properties of the optimal power allocation. We show that in order to effectively balance the tradeoff between control performance and resource utilization the sensor needs to opportunistically exploit online the channel conditions as well as the control system state. Our approach leads to novel designs as it is shown that the sensor needs to allocate a wide range of transmit powers, in contrast to the binary event-triggered approaches where a sensor decides just whether to transmit or not. In fact event-triggered policies arise as a special case in our context. These contributions are detailed in Chapter 3.

Scheduling Sensor-Actuator Systems over Shared Wireless Channels

For the case of multiple wireless control systems sharing a wireless channel we develop centralized scheduling and power allocation policies. The goal of the policies is to maintain a level of control performance for each control system, modeled via Lyapunov functions, while minimizing the total power consumption. This formulation leads to a constrained stochastic optimization problem and the structure of the optimal solution is characterized. The scheduler opportunistically exploits the channel conditions experienced by the control systems to decide which system should get access to the channel. The optimal transmit powers are shown to be decoupled among systems, allowing for an easy implementation. We further describe algorithms to find the optimal policies either offline, when the channel distributions are available, or online based on samples from the channel

states observed during operation. In the latter case we show that the algorithm converges almost surely to the optimal scheduling and power allocation rule. These contributions are detailed in Chapter 4.

Random Access Design for Wireless Sensor-Actuator Systems

When alternatively a central authority is not available to implement scheduling decisions, sensors randomly and independently decide whether to access the channel. The goal is again to guarantee closed loop control performance, however in this case a mitigation of packet collisions arising from simultaneously transmitting sensors is required. The proposed decentralized sensor access policies are derived by a constrained optimization formulation. Despite the fact that sensor policies are coupled, as each sensor is causing collisions to all other links and is subject to collisions from all other sensors, a decentralized structure emerges. Each sensor should access the channel when its local channel conditions exceed a threshold, i.e., are favorable enough. The threshold is selected in a way that balances the gains from transmitting with the losses due to packet collisions on all other systems. Decentralized algorithms to obtain the optimal mechanism are also explored. These contributions are detailed in Chapter 5.

Online Control-aware Resource Allocation

While the optimal power allocation problem examined in Chapter 3 indicates that the sensor needs to allocate its transmit power online based on plant and channel states, computing the optimal policy is computationally hard. In particular, the dynamic nature of the control system state makes it hard to account for the optimal future behavior of the system (or technically the optimal cost-to-go function). To alleviate this computational complexity we provide suboptimal yet tractable state-aware policies based on the approximate dynamic programming technique of rollout algorithms. This approach guarantees by design that the suboptimal state-aware policies have a control performance and resource utilization that is superior to other non-state-aware policies, which are hereby used as reference policies.

We further develop state-aware policies for the decentralized channel access problem. In this case the rollout policies for the sensors are explicitly constrained to prescribed average channel access rates that do not violate control performance of other systems due to collisions. The design of such constrained policies becomes tractable based on Lagrange duality arguments. These contributions are detailed in Chapter 6.

Chapter 2: Resource-aware Wireless Control Systems

The control systems considered in this work are characterized by physical separation between the sensors and controllers/actuators. The sensors are deployed to measure the state of a dynamic physical system (plant) and transmit their measurements over wireless channels to controllers/actuators. The latter are responsible for providing appropriate control inputs in order to regulate the plant given the received information. In this context the most basic architecture is presented in Fig. 1 and consists of a single sensor-actuator pair over a wireless channel. The goal of this work is to provide resource-aware and control-aware communication designs. In this chapter we discuss the employed plant and controller models, as well as the wireless communication model which explicitly takes into account the employed resources.



Figure 1: Wireless control system architecture. At each time step k the sensor measures and transmits the plant state x_k over a wireless channel to the controller. The selected transmit power p_k and the wireless fading state h_k determine whether the message is successfully received. The goal is to achieve a good trade-off between power consumption and closed loop performance.

2.1. Control System Model

The goal of the architecture in Fig. 1 is to regulate a plant. The evolution of the plants considered in this work can be described by a discrete-time linear time-invariant system given by

$$x_{k+1} = Ax_k + Bu_k + w_k, \ k \ge 0.$$
(2.1)

Here $k \ge 0$ represents discrete time steps, $x_k \in \mathbb{R}^n$ is the plant's state with x_0 being an initial given plant state, $u_k \in \mathbb{R}^m$ is the driving input, and $\{w_k, k \ge 0\}$ is the process noise which unless otherwise noted is an independent identically distributed (i.i.d) *n*-dimensional Gaussian process

 $\mathcal{N}_{0,W}$ with zero mean and covariance *W*. The process noise repeatedly perturbs the system from the desirable equilibrium operating point 0, taken without loss of generality. If the system is unstable, i.e., its spectral radius is larger than unity $\rho(A) := \max_i |\lambda_i(A)| > 1$, the plant state will grow unbounded unless appropriate feedback control inputs are provided.

Given a wireless channel model and a communication policy, which is the main part of our design and is detailed in the following section, plant state measurements from the sensor/transmitter become available to the receiver/controller. Throughout this work we assume that communication is organized in packets containing each sensor measurement x_k at each time slot (Hespanha et al. (2007)). A packet transmitted might get received at the receiver during that slot or it might get dropped. We model the packet success for each time step k as an indicator γ_k taking value $\gamma_k = 1$ when information is successfully decoded and $\gamma_k = 0$ otherwise. Variables γ_k are random due to uncertainty introduced in the wireless communication as we will see in the following section. In general they are binary random variables with a time-varying success probability $\mathbb{P}(\gamma_k = 1) \in [0, 1], k \ge 0$.

When a packet is successfully received, the plant state x_k becomes completely known to the controller – the effects of measurement quantization and transmission delays are considered negligible and are thus ignored henceforth. When a packet is dropped, the controller does not know the current state of the plant. More formally, given the packet success model $\gamma_k \in \{0, 1\}$ the controller receives an output $\gamma_k x_k$ which is the actual system state when a packet is received and 0 otherwise. We can further assume that the controller also gets the value of γ_k so that it can distinguish between the cases $x_k = 0$ and $\gamma_k = 0$. Overall the information available at the controller at time k consists of all previously received messages given by $\{\gamma_0, \ldots, \gamma_k, \gamma_0 x_0, \ldots, \gamma_k x_k\}$. A standing assumption throughout this work is that the receiver also sends lossless packet acknowledgments to the sensor/transmitter.

Given the available information, the controller is responsible for deciding what control input u_k to be applied as a feedback to the plant at each time step. The general problem of designing appropriate control input policies in this context is examined in detail in Chapter 3. In particular the goal of the control policy is to provide good closed loop control performance. The latter can be measured as a typical quadratic cost on the plant state and the control input, i.e.,

$$\limsup_{N \to +\infty} \frac{1}{N} \mathbb{E} \sum_{k=0}^{N-1} x_k^T Q x_k + u_k^T R u_k.$$
(2.2)

The limit in this expression accounts for long term performance while the expectation accounts for the randomness introduced by the process noise and the wireless communication.

Alternatively some control input policy may be fixed, i.e., the controller is pre-designed, and the remaining goal is to design the communication policy. To mathematically formulate this case we consider some examples of frequently used controller policies in networked control systems.

Example 2.1. Consider the control of a plant given by (2.1). A very simple controller is described by

$$u_k = \begin{cases} K x_k, & \text{if } \gamma_k = 1\\ 0, & \text{if } \gamma_k = 0 \end{cases}$$
(2.3)

That is, the controller applies a linear feedback with gain $K \in \mathbb{R}^{m \times n}$ when a measurement is received, otherwise does not apply any input when no measurement is received (Hadjicostis and Touri (2002)). The resulting closed loop system has two modes of operation depending on whether the measurement is received or not. In particular it can be written as a linear switched system of the form

$$x_{k+1} = \begin{cases} (A+BK) x_k + w_k, & \text{if } \gamma_k = 1\\ A x_k + w_k, & \text{if } \gamma_k = 0 \end{cases}$$
(2.4)

Intuitively the closed loop mode A + BK is chosen to be a stable system.

Example 2.2. Alternatively, as a second example consider a controller with the ability to hold the last applied input until a new measurement is received (Zhang et al. (2001); Seiler and Sengupta (2001)), that is,

$$u_k = \begin{cases} K x_k, & \text{if } \gamma_k = 1\\ u_{k-1}, & \text{if } \gamma_k = 0 \end{cases}$$
(2.5)

where K is a state feedback gain. The overall evolution of the system can be expressed as a switched system again where the plant state x_k is augmented with the last applied input u_{k-1} to rewrite the system dynamics in the form

$$\begin{bmatrix} x_{k+1} \\ u_k \end{bmatrix} = \begin{cases} \begin{bmatrix} A+BK & 0 \\ BK & 0 \end{bmatrix} \begin{bmatrix} x_k \\ u_{k-1} \end{bmatrix} + \begin{bmatrix} I \\ 0 \end{bmatrix} \begin{bmatrix} w_k \end{bmatrix}, & \text{if } \gamma_k = 1 \\ \begin{bmatrix} A & I \\ 0 & I \end{bmatrix} \begin{bmatrix} x_k \\ u_{k-1} \end{bmatrix} + \begin{bmatrix} I \\ 0 \end{bmatrix} \begin{bmatrix} w_k \end{bmatrix}, & \text{if } \gamma_k = 0 \end{cases}$$
(2.6)

This is again a switched linear time-invariant system with two modes of operation, open and closed.

Example 2.3. As a third example, consider a more sophisticated controller which may compute and apply a different control input at each time step. Since the controller does not always have access to the plant state, it is reasonable to keep a local state estimate \hat{x}_k and then apply a control input as a linear feedback with respect to the estimate, that is,

$$u_k = K \hat{x}_k. \tag{2.7}$$

An intuitive rule for updating the state estimate is as follows (Xu and Hespanha (2004); Ramesh et al. (2013))

$$\hat{x}_{k} = \begin{cases} x_{k} & \text{if } \gamma_{k} = 1, \\ A\hat{x}_{k-1} + Bu_{k-1} & \text{if } \gamma_{k} = 0 \end{cases},$$
(2.8)

In other words, when no measurement is received the plant dynamics are used in order to propagate the last estimate. We can combine the plant state with the state estimate in an augmented state vector to express the

overall dynamics as

$$\begin{bmatrix} x_{k+1} \\ \hat{x}_k \end{bmatrix} = \begin{cases} \begin{bmatrix} A+BK & 0 \\ I & 0 \end{bmatrix} \begin{bmatrix} x_k \\ \hat{x}_{k-1} \end{bmatrix} + \begin{bmatrix} I \\ 0 \end{bmatrix} \begin{bmatrix} w_k \end{bmatrix}, & \text{if } \gamma_k = 1 \\ \begin{bmatrix} A & BK (A+BK) \\ 0 & A+BK \end{bmatrix} \begin{bmatrix} x_k \\ \hat{x}_{k-1} \end{bmatrix} + \begin{bmatrix} I \\ 0 \end{bmatrix} \begin{bmatrix} w_k \end{bmatrix}, & \text{if } \gamma_k = 0 \end{cases}$$
(2.9)

This is again a switched linear time-invariant system with two modes of operation, open and closed. In Chapter 3 it will be shown that this controller becomes optimal for the wireless control system under appropriate assumptions.

As can be seen from the preceding examples a general model for the evolution of the plant and pre-designed controller is a switched linear time-invariant system of the form

$$x_{k+1} = \begin{cases} A_c x_k + w_k, & \text{if } \gamma_k = 1 \\ A_o x_k + w_k, & \text{if } \gamma_k = 0 \end{cases} .$$
(2.10)

Here $x_k \in \mathbb{R}^n$ is the overall state of the control system, with the understanding that this may include controller states as well as in the examples above. The evolution of the state has two possible linear modes of operation depending on transmission success. At a successful transmission the system dynamics are described by the matrix $A_c \in \mathbb{R}^{n \times n}$, where 'c' stands for closed-loop, and otherwise by $A_o \in \mathbb{R}^{n \times n}$, where 'o' stands for open-loop. We assume that A_c is asymptotically stable, implying that if system were to transmit at each slot its respective state evolution is stable. The open loop matrix A_o may be unstable. The terms $\{w_k, k \ge 0\}$ are again the i.i.d. Gaussian process noise with zero mean and covariance W.

For a fixed wireless control system of the form (2.10) our goal will be to design appropriate communication policies that yield good closed loop control performance, i.e., ability to regulate plant states to the desirable operating point 0 despite external disturbances or communication uncertainties. In particular we consider two different measures of control performance that will facilitate the communication design. These are presented next.

2.1.1. Lyapunov-based Control Performance Abstraction

First we capture the ability of the wireless control system to drive fast the system state to the desirable operating point 0. Convergence to 0 is typically analyzed in control systems via non-negative functions V(x) of the plant state, i.e., Lyapunov functions. The Lyapunov function of the state decreases if $V(x_{k+1}) \leq \rho V(x_k)$ for all $x_k \in \mathbb{R}^n$ for some scalar $\rho < 1$ which denotes some *desired rate of decrease*. Due to the randomness introduced by the plant disturbance and the wireless communication it is not possible to guarantee that the Lyapunov function decreases at each time step. In fact when a plant state measurement is not received the system runs in open loop and it typically increases $V(x_{k+1}) > V(x_k)$. As an alternative then we can require that the

Lyapunov function decreases in expectation with respect to the system uncertainties, that is,

$$\mathbb{E}[V(x_{k+1}) \mid x_k] \le \rho V(x_k) + \varepsilon \text{ for all } x_k \in \mathbb{R}^n \text{ and all } k \ge 0$$
(2.11)

for a desired decrease rate $\rho < 1$ and some constant ε , independent of the system state. The expectation in this expression is with respect to the random disturbance $w_k, k \ge 0$ and the random wireless communication process $\gamma_k, k \ge 0$.

For the special case of switched linear systems in (2.10) we suppose that a quadratic Lyapunov function of the form $V(x) = x^T P x$ for some positive definite matrix $P \in \mathbb{S}^n_+$ is given. In that case we can write the control performance specification in the form

$$\mathbb{E}[V(x_{k+1}) \mid x_k] \le \rho V(x_k) + \operatorname{Tr}(PW) \text{ for all } x_k \in \mathbb{R}^n \text{ and all } k \ge 0.$$
(2.12)

Here $\rho < 1$ is a given desired decrease rate. The term Tr(PW) is a constant appearing due to the linearity of the system and the additive process noise with covariance *W* (cf. (2.10)) as well as the quadratic form of the Lyapunov function.

The intuition behind condition (2.12) is as follows. If (2.12) holds for each time step k = 0, ..., N, then by taking the expectation at both sides and by iterating backwards in time we find that

$$\mathbb{E}V(x_N) \le \rho \,\mathbb{E}V(x_{N-1}) + Tr(PW) \le \dots \le \rho^N \,\mathbb{E}V(x_0) + \sum_{k=0}^{N-1} \rho^k \,Tr(PW).$$
(2.13)

Hence, system states have second moments that decay exponentially with rate ρ with respect to initial states, and in the limit remain bounded by $Tr(PW)/(1-\rho)$, since the sum in (2.13) converges due to $\rho < 1$.

The above Lyapunov control performance abstraction facilitates communication design. In Chapter 4 we will convert the above control performance requirements to equivalent explicit wireless communication requirements, in particular required packet success rates.

2.1.2. Quadratic Control Performance Abstraction

An alternative approach is to consider quadratic regulation costs as usually associated with linear systems. In particular at each time step we can consider a cost of the form

$$\begin{cases} x_k^T Q_c x_k, & \text{if } \gamma_k = 1\\ x_k^T Q_o x_k, & \text{if } \gamma_k = 0 \end{cases}.$$
(2.14)

This costs depends in general on whether transmission occurs ($\gamma_k = 1$) or not ($\gamma_k = 0$) at time k. Both $n \times n$ matrices Q_c , Q_o are assumed to be positive semidefinite and are design choices. The case where both matrices are equal is a special case. In view of the system evolution (2.10) the state x_k is a random variable affected by plant disturbance and wireless communication uncertainties. As it is of interest to examine the long term behavior of the system we consider the average expected quadratic cost

$$\limsup_{N \to \infty} \frac{1}{N} \sum_{k=0}^{N-1} \mathbb{E}[x_k^T \left(\gamma_k Q_c + (1 - \gamma_k) Q_o\right) x_k].$$
(2.15)

Average penalties on actuation, i.e., the actual control inputs, can also be captured in (2.15) as seen in the following example.

Example 2.4. Consider the control system described in Example 2.2 which is expressed with respect to an augmented state x_k, u_{k-1} . Suppose that we want to account for a joint cost on the plant state and control inputs $x_k^T Q x_k + u_k^T R u_k$ for some $n \times n$ and $m \times m$ symmetric positive definite matrices Q, R. It can be seen from (2.5) that whenever a packet is received this joint cost equals $x_k^T (Q + K^T R K) x_k$. Otherwise, when a packet is dropped, the joint cost becomes $x_k^T Q x_k + u_{k-1}^T R u_{k-1}$. In either case this is quadratic cost on the augmented state that depends on whether the current transmission is successful or not. This is of the general form (2.14) with respect to the augmented system state.

In the following section we will describe in detail the wireless communication model which determines the packet indicator variables γ_k , $k \ge 0$. Before doing so we briefly describe a difference between the two control performance abstractions presented above.

Remark 2.1. We point out that overall in this work we are interested in communication design for control performance, in contrast to determining what communication designs guarantee stability, as is commonly examined in the literature, e.g. Zhang et al. (2001); Hristu-Varsakelis (2001); Hespanha et al. (2007); Donkers et al. (2011). The two proposed control performance abstractions provide for different communication design approaches. We will show (Prop. 4.1 in Chapter 4) that the Lyapunov control performance abstraction leads to an explicit wireless communication requirement. In particular it corresponds to a convex requirement (lower bound) on the packet success rate at each time step, and leads to a design that is decoupled from the plant states over time. In contrast, the average quadratic control performance in (2.15) is in general a non-convex function of the packet success rates. The communication design in this case (Chapter 6) can exploit dynamic programming approaches and leads to solutions that are adapted online to plant states.

2.2. Wireless Channel Model

Consider again the wireless control architecture of Fig. 1. At each time step k the wireless sensor transmits the current plant state measurement with some power $p_k \in [0, p_{max}]$ which is a design variable over a wireless fading channel with coefficient h_k . Due to propagation effects the channel gain h_k changes unpredictably over time (Goldsmith, 2005, Ch. 3). We adopt the standard block fading model of wireless communications whereby channels $\{h_k, k \ge 0\}$ are modeled as i.i.d. random variables taking values in the positive reals \mathbb{R}_+ according to some distribution ϕ_h (Goldsmith, 2005, Ch. 4). The channel states are also independent of the plant process noise $\{w_k, k \ge 0\}$ (cf.(2.1) or (2.10)). To allow for transmissions adapted to the current channel conditions the transmitter has access to the channel state information h_k before transmitting at time k –



Figure 2: Complementary error function for FEC and capacity achieving codes. The probability of successful decoding *q* for a practical FEC code is a sigmoid function of the received SNR = $h p/N_0$, while for a capacity achieving code a threshold value SNR₀ determines whether a packet is successfully received.

see Remark 2.3.

At the controller side the received signal includes the information bearing signal and additive white Gaussian noise (AWGN). The noise power is denoted by N_0 and the power of the information bearing signal is the product $h_k p_k$. Assuming the receiver also has channel state information, successful decoding of the transmitted packet is determined by the signal to noise ratio (SNR) at the receiver defined as

$$\mathrm{SNR}_k := \frac{h_k \, p_k}{N_0}.\tag{2.16}$$

More precisely, given the particular type of modulation and forward error correcting (FEC) code used, the SNR determines the probability of successful decoding $\mathbb{P}(\gamma_k = 1)$. To keep the analysis general we define a generic complementary error function

$$\mathbb{P}(\gamma_k = 1) = q(h_k, p_k), \qquad (2.17)$$

mapping $\text{SNR}_k := h_k p_k / N_0$ to the packet success probability. We assume that q(h, p) is a known increasing function of the product h p - see Remark 2.2.

Considering packet decoding as a part of the communication process, we can model communication as a sequence of indicator variables γ_k taking value $\gamma_k = 1$ when information is successfully decoded and $\gamma_k = 0$ otherwise. Variables γ_k are Bernoulli distributed with time-varying success probabilities $\mathbb{P}(\gamma_k = 1)$ given by (2.17). Packet receipt acknowledgment γ_k is also sent from the controller to the sensor as provided by 802.11 and TCP protocols. We assume lossless acknowledgments, so that the sensor knows what information is received at the controller.

A fundamental problem addressed in this work is the design of the transmit powers p_k of the sensor. The design should lead to desired control performance of the system as expressed, e.g., in the preceding section. Moreover the sensor should make an efficient use of its power resources. As a result the communication policy needs to take into account the average power utilization as

measured by

$$\limsup_{N \to +\infty} \frac{1}{N} \mathbb{E} \sum_{k=0}^{N-1} p_k.$$
(2.18)

The expectation here accounts for the uncertainty introduced by the random channel fading process, the random packet success, as well as the random control system evolution. A formal problem specification for the optimal power allocation will be presented in Chapter 3.

The sensor may select transmit power per each packet as a function of the information available to the sensor, i.e., the plant state measurements, the observed channel realizations, and the receiver acknowledgments. Informally, to conserve power at the sensor side we want to transmit information only when the state x_k deviates from its desired value or when the channel realization h_k is favorable. In the first case transmission is necessary to keep the plant under control. In the latter case the transmission cost is minimal.

The present wireless channel model can be naturally extended to the case where multiple control systems are

Remark 2.2. The error profiles $1 - q(h_k, p_k)$ of particular FEC codes are difficult to determine analytically but can be measured in actual or simulated experiments (Zhang et al. (2008); Ploplys et al. (2004)). Typically $q(h_k, p_k)$ is a sigmoid function of $h_k p_k$ with exponential tails as depicted in Fig. 2. In the theoretical limit, correct decoding depends on the channel capacity $C_k = W \log_2(1 + SNR_k)$, where W is the channel bandwidth. If the packet is transmitted at a rate smaller than C_k bits per second it is almost surely successfully decoded, and it is almost surely incorrectly decoded otherwise. Thus, we can write the successful decoding probability as the indicator function

$$q(h_k, p_k) = \mathbb{I}\left(\frac{h_k p_k}{N_0} \ge SNR_0\right),\tag{2.19}$$

for some constant SNR₀. Determining the threshold SNR₀ requires specification of the sampling rate and quantization resolution of the state x_k . With α samples per second and β bits per sample we require a transmission rate of $\alpha\beta$ bits per second. The SNR threshold is then given by SNR₀ = $2^{\alpha\beta/W} - 1$. Our interest in (2.19) is conceptual as it will allow us to recover results in event-triggered communication (Xu and Hespanha (2004)) as arising from the use of capacity achieving codes – see Section 3.4.1. The form of (2.19) is shown in Fig. 2.

Remark 2.3. The assumption that channel state information (CSI) is available at the transmitter is typical in modern wireless communication setups (Goldsmith, 2005, Ch. 9). To measure the wireless channel conditions a short pilot signal of fixed power can be sent from the transmitter and then the fading characteristics can be estimated at the receiver and sent back to the transmitter by utilizing the reverse channel. Although accurate CSI is difficult to acquire at the transmitter side, our development can be modified if channel estimates are available in lieu of the actual channel value h_k with a reinterpretation of the function q(.). Further discussion is provided in Appendix A. We additionally point out that even though the pilot signals for the channel estimation incur some power consumption, we assume that practically this is much lower than the



Figure 3: Architecture for control over multiple access channels. Independent control systems close their loops by transmitting over the shared wireless medium to a common receiver/access point. Each control system *i* experiences random channel conditions h_i . A centralized scheduler at the access point observes all channel states and opportunistically decides which system is scheduled to transmit and close the loop.

power necessary for transmitting the packets of the control system, especially for large packet lengths (e.g. long headers). Hence the power for channel estimation is not included in our objective in (2.18).

Remark 2.4. There is a distinction to be made between errors that are detected by the receiver and errors that are undetected and may confuse the controller. The model here handles the former and ignores the latter. This is justified because practical communication schemes include the use of cyclic redundancy checks (CRC) for error detection that can drive the probability of undetected errors to very small values (Moon, 2005, Ch. 4). The use of simple CRCs reduces the probability of undetected errors to 10^{-3} , while longer codes can reduce this probability to 10^{-7} .

2.2.1. Multiple Access Channels

Consider now the case where multiple sensors need to communicate over a shared wireless medium in order to close their loops, as depicted in Fig. 3. Each sensor is measuring the state of an independent plant and transmits its local state information to a common access point. For example this can be a common central controller computing the control inputs to the plants. Each control system has their own independent dynamics, expressed, e.g., in the form (2.10). The access point is responsible for deciding at each time step which sensor is scheduled, i.e., gets access to the channel. The goal is to achieve control performance for all systems.

Each control loop i (i = 1, 2, ..., m) experiences different wireless fading conditions over the shared channel. In particular by $h_{i,k}$ we refer to the channel fading coefficient that system i experiences if it transmits at time slot k. Following a block fading model as in the single control loop case, channel states $\{h_{i,k}, 1 \le i \le m\}$ are modeled as constant during each transmission slot k, but independent and identically distributed across different time slots k according to some joint distribution ϕ_h on \mathcal{H}^m . We assume the channel states are available to the access point before transmission – see

Remark 2.5 for a practical implementation.

Following our power allocation model, if system *i* is scheduled to access the channel at a given time step, the corresponding sensor selects a transmit power level $p_{i,k} \in [0, p_{max}]$. As described in (2.17) the probability that the transmitted packet is successfully decoded is a function of the power level and the link's channel conditions given by a relationship of the form $q(h_{i,k}, p_{i,k})$.

The goal of the access point/scheduler is to select one system to access the channel at each time step. We denote with $\alpha_{i,k} = 1$ the decision to schedule system *i* at time *k*, and $\alpha_{i,k} = 0$ otherwise. Since at most one system is scheduled we have $\sum_{i=1}^{m} \alpha_{i,k} \leq 1$. Let us indicate with $\gamma_{i,k} \in \{0,1\}$ the event that a successful transmission occurs at time *k* for the subsystem *i*. This is a Bernoulli random variable with success probability

$$\mathbb{P}[\gamma_{i,k} = 1 \mid h_k, \alpha_k, p_k] = \alpha_{i,k} q(h_{i,k}, p_{i,k})$$
(2.20)

This expression states that the probability of a message for system *i* being successfully received equals the probability that system *i* is scheduled to transmit and the message is correctly decoded.

The scheduler to be designed can take into account current channel information when making the scheduling decisions online. Intuitively the scheduler might decide to give access to the system currently experiencing the most favorable channel conditions, as that will result in a most reliable communication. An important restriction however is posed on the scheduler design due to the physical systems. To avoid loss of stability or deterioration of control performance, the scheduler needs to guarantee that each system gets access to the shared channel at a sufficient rate. The detailed scheduler design in presented in Chapter 4.

Remark 2.5. The centralized scheduler of the multiple access channel architecture in Fig. 3 requires channel state information. The channel conditions for each system can be measured at the access point at the beginning of each time slot by short pilot signals sent from the wireless transmitters of all systems to the access point. Depending on the measured channel states the access point decides which plant is scheduled to close the loop during the time slot.

2.2.2. Decentralized (Random) Channel Access

An alternative mechanism for sharing the wireless medium is random access. In contrast to centralized scheduling, this is a decentralized mechanism where sensors independently and randomly decide whether to access the shared wireless medium and transmit plant state measurements to the access point/controller (Fig. 4). Such a mechanism is easier to implement in practical scenarios, as it does not require a central authority to take scheduling decisions. The drawback of this decentralized approach however is that packet collisions can occur from simultaneously transmitting sensors, resulting in lost packets and control performance degradation. Hence sensor access policies need to be appropriately designed to mitigate these effects.

At every slot *k* each sensor *i* transmits over the shared channel with some probability $\alpha_{i,k} \in [0, 1]$



Figure 4: Random access architecture for *m* control loops over a shared wireless medium. Each sensor *i* randomly transmits with probability $\alpha_{i,k}$ at time *k* to a common access point computing the plant control inputs. If only sensor *i* transmits, the successful decoding probability depends on local channel conditions $h_{i,k}$. If other sensors transmit at the same time a collision might occur at sensor *i*'s transmission, rendering *i*'s packet lost.

which is a decision variable. A sensor's transmission might fail due to two reasons, packet decoding errors and packet collisions. A collision might be experienced on link *i*, thereby rendering packet *i* lost, if some other sensor $j \neq i$ transmits in the same time slot. We assume that such a collision event occurs with constant probability $q_{ji} \in [0, 1]$, *given* that both sensors *i*, *j* transmit in the slot. Thus, the probability that sensor *i*'s transmission is free of collisions, i.e., that no other sensor transmits and causes collisions on link *i*, equals $\prod_{j\neq i} [1 - \alpha_{j,k} q_{ji}]$. See Remark 2.6 for details of this collision model.

We adopt once again a block fading channel model whereby $h_{i,k}$ is the channel fading coefficient that system *i* experiences if it transmits at time slot *k*. The channel fading coefficients are randomly varying over time. If sensor *i* transmits and has a collision free time slot, the success of decoding the message at the access point/receiver depends on the randomly varying channel conditions on link *i* and transmit power. Assuming for simplicity that transmit power is fixed to some value p_i in this setup, we denote by $q(h_{i,k}, p_i)$ the probability of successful transmission (cf.(2.17)).

Combining the effects of collisions and packet losses due to fading, the probability that a packet is successfully decoded at the access point can be written as

$$\mathbb{P}(\gamma_{i,k}=1) = \alpha_{i,k} q(h_{i,k}, p_i) \prod_{j \neq i} \left[1 - \alpha_{j,k} q_{ji} \right].$$
(2.21)

This expression states that the probability of system *i* in (2.10) closing the loop at time *k* equals the probability that transmission *i* is successfully decoded at the receiver, multiplied by the probability that no other sensor $j \neq i$ is causing collisions on *i*th transmission.

The goal of the random access design problem in this case it to design the sensor policies so that each involved control system has desirable control performance. One the one hand, this requires

management of the arising packet collisions between sensors. On the other hand, sensors can opportunistically exploit their local plant or channel information to decide whether to access the channel. Due to the shared wireless medium the sensor policies are coupled (cf. (2.21)). A formal problem specification and an efficient design procedure is presented in Chapter 5.

Remark 2.6. Our collision model captured by the probabilities q_{ji} subsumes: i) the conservative case where simultaneous transmissions certainly lead to collisions ($q_{ji} = 1$) usually considered in control literature, e.g., by Zhang (2003); Tabbara and Nesic (2008), ii) the case where simultaneously transmitted packets are not always lost ($q_{ji} < 1$), e.g., due to the capture phenomenon (Luo and Ephremides (2002)), and iii) the asymmetric case where different sensors j, ℓ interfere differently on link *i*, e.g., due to their spatial configuration.

Chapter 3: Optimal Power Management for Wireless Sensor-Actuator Systems

3.1. Problem Description

We consider the wireless control architecture shown in Fig. 5 deployed to control a discrete-time linear time-invariant plant described by the difference equation

$$x_{k+1} = Ax_k + Bu_k + w_k, \ k \ge 0, \tag{3.1}$$

where $x_k \in \mathbb{R}^n$ is the plant's state with x_0 given, $u_k \in \mathbb{R}^m$ the driving input, and $\{w_k, k \ge 0\}$ is the process noise composed of independent identically distributed (i.i.d) *n*-dimensional Gaussian random variables $w_k \sim \mathcal{N}_{0,W}$ with zero mean and covariance *W*. We assume the plant is unstable $(\lambda_{\max}(A) > 1)$ but that (A, B) is stabilizable.

The wireless sensor collects state measurements x_k that it communicates with power $p_k \in [0, p_{max}]$ over a wireless fading channel with coefficient h_k . At the other side of the channel the receiver/ controller uses the received information to determine a control input u_k that it feedbacks into the plant.

The channel fading state $h_k, k \ge 0$ is modeled as an i.i.d. process (block fading cf. 2.2) with some known probability distribution ϕ_h on the positive reals \mathbb{R}_+ , independent of the plant process noise $\{w_k, k \ge 0\}$. We make the technical assumption that the distribution of the channel state is absolutely continuous, i.e., has a probability density function on \mathbb{R}_+ . To allow for transmissions adapted to the current channel conditions the transmitter has access to the channel state information h_k before transmitting at time k.

Following the wireless channel modeled introduced in 2.2, the probability of successfully decoding the message at the receiver/controller at time k, denoted hereby by q_k , is a function of the current channel state h_k as well as the current transmit power p_k ,

$$q_k = q\left(h_k, p_k\right). \tag{3.2}$$

The typical form of this function is depicted in Fig. 2. We assume that q(h, p) is a known increasing function of the product h p. We denote the success of transmission at each time step k by an indicator variable $\gamma_k \in \{0, 1\}$. Variables $\gamma_k \sim \text{Bern}(q_k)$ are Bernoulli distributed with time-varying success probabilities q_k given by (3.2). With this communication model the controller receives the output of the decoding process which we model by the signal $y_k = \gamma_k x_k$. We further



Figure 5: Wireless control system architecture. A sensor measures the plant and wireless fading channel states x_k , h_k respectively and transmits with power p_k . Messages are successfully decoded at the controller with probability q_k that depends on the channel state h_k and the power p_k . The sensor receives acknowledgments with a one-step delay.

assume that the controller also gets γ_k so that it can distinguish between the cases $x_k = 0$ and $\gamma_k = 0$. Packet receipt acknowledgment γ_k is also sent to the sensor as provided by 802.11 and TCP protocols. We assume lossless acknowledgments, so that the sensor knows what information is received at the controller.

The problem addressed in this chapter is the joint design of the control inputs u_k and the transmit powers p_k . The control input u_k is determined by the received information $y_{0:k}$, $\gamma_{0:k}$. The power p_k is determined as a function of the plant state measurements $x_{0:k}$, the observed channel realizations $h_{0:k}$, and the controller acknowledgments $\gamma_{0:k}$. Informally, to conserve power at the sensor side we want to transmit information only when the state x_k deviates from its desired value or when the channel realization h_k is favorable. In the first case transmission is necessary to keep the plant under control. In the latter case the transmission cost is minimal. A formal problem specification is presented in the next section after the following remarks.

To formulate the joint design of plant controller and power management we introduce an equivalent architecture. In view of (3.2), choosing p_k is equivalent to choosing the desired probability of successful decoding q_k at time k and transmitting with the minimum required power to achieve this q_k , namely

$$p_k = p(h_k, q_k) := \inf \left\{ 0 \le p \le p_{\max} : q(h_k, p) \ge q_k \right\}.$$
(3.3)

We can therefore interpret q_k as our decision variable with $p(h_k, q_k)$ denoting the cost of selecting transmission success probability q_k . This leads to the equivalent control system architecture shown in Fig. 6 where a scheduler block responsible for deciding q_k replaces the sensor/transmitter block of Fig 5. Our formulation generalizes the simple transmit-or-not decision as considered in, e.g., Xu and Hespanha (2004).

We note for future reference that the assumed monotonicity of the function q(h, p) on the product h p implies that the power function p(h, q) is increasing in q and decreasing in h. Using maximum



Figure 6: Equivalent wireless control system architecture. A scheduler decides the successful decoding probability q_k and transmits the state measurement x_k with the required power $p_k = p(h_k, q_k)$. The controller receives the message with probability q_k .

power p_{max} , the transmitter can achieve a maximum successful decoding probability $q_{\text{max}}(h) := q(h, p_{\text{max}})$ for a given channel state *h*. Therefore, the decision variables q_k belong in the interval $[0, q_{\text{max}}(h_k)]$. We also make the following assumptions.

Assumption 3.1. The maximum achievable successful decoding probability $q_{\max}(h)$ satisfies

$$\mathbb{E}_{h}q_{\max}(h) > q_{crit} := 1 - 1/\lambda_{\max}(A)^{2}, \tag{3.4}$$

where expectation is taken over the channel distribution ϕ_h .

Assumption 3.2. For any channel realization h, the function p(h,q) in (3.3) is continuous in the successful decoding probability variable q.

Assumption 3.1 is essentially a stability condition, which as we will see in the following section states that transmitter has enough power to keep the plant state bounded in second moment, and it will be used to establish our main Theorems 3.1 and 3.2. Assumption 3.2 is of a technical nature and will be used in Theorem 3.2.

In the architecture of Fig. 6 the communication decision q_k is chosen as a function of the information available at the sensor, while the plant control signal u_k is a function of the information available at the controller. These choices are in general allowed to be randomized. The sequence $\pi := \{q_0, q_1, \ldots\}$, or equivalently the power allocation $\{p_0, p_1, \ldots\}$, is termed the communication policy, whereas the sequence $\theta := \{u_0, u_1, \ldots\}$ denotes the control policy. With fixed policies π, θ , all random variables are defined on an appropriate product space and have a measure that we denote as $\mathbb{P}^{\pi,\theta}$. We use $\mathbb{E}^{\pi,\theta}$ to denote integration with respect to $\mathbb{P}^{\pi,\theta}$, which we simplify to \mathbb{E} when not leading to confusion. We remark that sensor and controller know each other's policy.

The policy pair (π, θ) incurs a control cost and a communication cost. As a control cost we adopt

the standard linear quadratic regulator cost

$$J_{LQR}^{N}(\pi,\theta) := \mathbb{E}^{\pi,\theta} \sum_{k=0}^{N-1} x_{k}^{T} Q x_{k} + u_{k}^{T} R u_{k},$$
(3.5)

for some pair of matrices R > 0 and $Q \ge 0$, with $(A, Q^{1/2})$ detectable. The communication cost is given by the expected power consumption

$$J_{\rm PWR}^{N}(\pi,\theta) := \mathbb{E}^{\pi,\theta} \sum_{k=0}^{N-1} p(h_k, q_k).$$
(3.6)

To quantify the tradeoff between plant performance and power consumption we combine the LQR cost in (3.5) and the power cost in (3.6) into the limit aggregate cost

$$J(\pi,\theta) := \limsup_{N \to \infty} \frac{1}{N} \left[J_{\text{LQR}}^N(\pi,\theta) + \lambda J_{\text{PWR}}^N(\pi,\theta) \right],$$
(3.7)

for some positive constant $\lambda > 0$. Our goal is to design plant and power control policies θ and π respectively that minimize the joint cost (3.7). These policies depend on what information is available to the sensor and controller. The specific information structure considered is introduced next.

3.2. Information structure

Denote as O_k the information known at the controller side at time *k* just before deciding the input u_k . This information includes the given initial plant state x_0 , the history of decoding success variables $\gamma_{0:k}$ and the decoded signals $y_{0:k}$, as well as the previously chosen control inputs $u_{0:k-1}$, i.e.,

$$O_k := \{x_0, \gamma_{0:k}, y_{0:k-1}\}.$$
(3.8)

Then the control input u_k is chosen as a function of O_k , or more formally, measurable with respect to the σ -field generated by O_k .

Given the possibility of lost packets as indicated by $\gamma_k = 0$, the controller has partial information on the plant state x_k . It is then of importance to study the MMSE estimate $\mathbb{E}^{\pi,\theta}(x_k|O_k)$. This estimation is complicated by the fact that the event $\gamma_k = 0$ possibly contains information about the state x_k through the dependence of the probability q_k on the value of x_k – see Remark 3.1. To avoid this complication we discard the information given by events of the form $\gamma_k = 0$. Formally, define $\tau_k := \max\{0 \le l \le k : \gamma_l = 1\}$ as the time of the last successful transmission by time *k* and define the sequence

$$G_k := \{x_0, \gamma_{0:\tau_k}, y_{0:\tau_k}, u_{0:k-1}\}.$$
(3.9)

with $G_0 = \{x_0\}$. When $\gamma_k = 1$, G_k coincides with O_k . When $\gamma_k = 0$, G_k only contains information received till the last successful transmission which occurred at time $\tau_k < k$.

We restrict attention to control policies θ selecting inputs u_k as functions of G_k , possibly randomized, and denote the set of all such policies by Θ . Unlike $\mathbb{E}^{\pi,\theta}(x_k|O_k)$, the state MMSE estimate $\hat{x}_k := \mathbb{E}^{\pi,\theta}(x_k|G_k)$ with respect to G_k is easy to compute. When $\gamma_k = 1$ the state $x_k = y_k$ becomes known at the receiver side. When $\gamma_k = 0$ no new information becomes available and \hat{x}_k is obtained by propagating \hat{x}_{k-1} through the plant's dynamics in (3.1). Put together, we get

$$\hat{x}_{k} := \mathbb{E}^{\pi, \theta}(x_{k}|G_{k}) = \begin{cases} y_{k} & \text{if } \gamma_{k} = 1, \\ A\hat{x}_{k-1} + Bu_{k-1} & \text{if } \gamma_{k} = 0 \end{cases}$$
(3.10)

with $\hat{x}_0 = x_0$ since the initial state is given.

At the other side of the link at time *k* the sensor/transmitter has access to the channel realization h_k and the plant state x_k which allows selection of the successful transmission probability q_k to depend on the values of both of h_k , x_k . This affects the controller design however, because when the controller decides u_{k-1} to control x_k , it should consider the indirect effect on q_k . This information structure renders the joint communication and control co-design problem hard to analyze. To overcome this, we restrict transmission policies to depend on the channel state h_k and the information about plant state x_k that the controller does not know. More precisely consider the difference between the sensor measurement x_k and the controller's estimate \hat{x}_k by (3.10) if the *k*th packet is *not* successfully decoded, that is

$$\varepsilon_k := x_k - (A\hat{x}_{k-1} + Bu_{k-1}), \tag{3.11}$$

with $\varepsilon_0 := 0$. Observe that the term in the parenthesis is known to the sensor since by the acknowledgment mechanism the controller's previous estimate \hat{x}_{k-1} and input u_{k-1} can be replicated at the sensor. Alternatively the terms ε_k can be viewed as the innovations of the controller's estimate (3.10) when a new message is received.

We restrict then information at the sensor side to the set F_k defined as a collection of the channel history $h_{0:k}$, the history of innovations $\varepsilon_{0:k}$, and past decisions $q_{0:k-1}$, i.e.,

$$F_k := \{\varepsilon_{0:k}, h_{0:k}, q_{0:k-1}\}.$$
(3.12)

We also add a technical requirement that the sensor selects maximum transmit power p_{max} when the innovation ε_k gets too large, $\|\varepsilon_k\| \ge L$ for some positive constant L > 0, and the channel gain h_k is favorable, $h_k \ge h_t$ for some threshold $h_t > 0$ on channel values. We assume that a positive success probability $q_{\text{max}}(h_t) > 0$ is achieved at this threshold and also that

$$\int_{h \ge h_t} q_{\max}(h) \, d\phi_h(h) > q_{\text{crit}},\tag{3.13}$$

where q_{crit} is given in (3.4) and the integration is over the channel distribution ϕ_h . Such a channel threshold exists by Assumption 3.1. We consider then communication policies π selecting decoding success q_k as functions of F_k for each k, possibly randomized, satisfying $q_k \in Q(\varepsilon_k, h_k)$


Figure 7: Wireless control system with a restricted information structure. The sensor consists of two blocks. A pre-processor computes the error ε_k given the measurement x_k and the acknowledgment γ_{k-1} . A scheduler decides q_k based on ε_k and the channel state h_k , and transmits x_k with the required power $p_k = p(h_k, q_k)$. The controller receives the message with probability q_k , computes the state estimate \hat{x}_k and provides input u_k to the plant.

where

$$Q(\varepsilon, h) := \begin{cases} q_{\max}(h) & \text{if } \|\varepsilon\| \ge L \text{ and } h \ge h_t \\ [0, q_{\max}(h)] & \text{otherwise} \end{cases}$$
(3.14)

We denote the set of all such policies with Π . The technical power saturation requirement is inconsequential as we may pick *L* arbitrarily large, and will be used to prove Proposition 3.2 and Theorem 3.2 in the sequel. Similar requirements have been introduced in Xu and Hespanha (2004); Mesquita et al. (2012), however our setup is further complicated by the availability of the random channel states.

The proposed information structure is depicted in Fig. 7. The sensor block is split into a preprocessor and a scheduler. The pre-processor computes ε_k based on the sample x_k and the acknowledgment γ_{k-1} and feeds it to the scheduler who, upon measuring the channel h_k decides the transmission success probability q_k while incurring power cost $p(h_k, q_k)$. Our goal in this paper is to study policies $\pi \in \Pi$ and $\theta \in \Theta$ that are optimal with respect to the joint objective (3.7), that is

$$\min_{\pi \in \Pi, \theta \in \Theta} J(\pi, \theta).$$
(3.15)

In particular, the next section shows that the information structure we introduced allows optimal communication and control policies to be designed separately. The standard LQR controller is shown to be optimal and we then leverage this result to study optimal communication policies in Section 3.4.

Remark 3.1. If the controller uses the complete information O_k to estimate x_k , the optimal plant estimate is not \hat{x}_k as given by (3.10). When a packet drop $\gamma_k = 0$ is observed, and since the communication policy is known, the controller should consider the possibility that the sensor did not transmit at all, which could in general give indirect information about the value of x_k – see also Molin and Hirche (2009); Imer and Başar (2010); Lipsa and Martins (2011); Rabi et al. (2012); Ramesh et al. (2013); Nayyar et al. (2013) for further discussion on this issue. The restriction to G_k in (3.9) allows to overcome this issue and obtain linear dynamics of the estimation error e_k and the related ε_k as described next in (3.16),(3.17), but it is not needed for the separation result of Prop. 3.1 in the following section to hold. An alternative approach to overcome this issue, followed by other works, is to restrict the communication policy design to transmission policies with suitable symmetries facilitating estimation (Imer and Başar, 2010; Rabi et al., 2012; Ramesh et al., 2013; Han et al., 2015).

3.3. Separation of designs

In this section we show that with the imposed restrictions on the information available at sensor and controller the control law $\theta \in \Theta$ and the communication policy $\pi \in \Pi$ can be designed separately. In particular the control policy has no effect on the estimation process at the receiver and by utilizing a separation principle the optimal controller becomes the standard linear quadratic one.

Let us denote the difference between the plant state and the estimate kept at the controller by $e_k := x_k - \hat{x}_k$ and its covariance as seen at the controller by $\Sigma_k := \mathbb{E}^{\pi} \left[e_k e_k^T \mid G_k \right]$. The estimation error dynamics can be found by subtracting (3.10) from the system dynamics (3.1) to get

$$e_k = (1 - \gamma_k)(Ae_{k-1} + w_{k-1}), \tag{3.16}$$

with $e_0 = 0$ since x_0 is given. Stabilizability of estimation error is guaranteed by Assumption 3.1. Indeed if transmitter were to use maximum power all the time the dynamics in (3.16) become a jump linear system since γ_k are Bernoulli with constant probability equal to the left hand side of (3.4). Then condition (3.4) is sufficient for bounded second moment as, e.g., in (Hespanha et al., 2007, Theorem 2). It is also tight in the sense that estimation error becomes unstable if $\mathbb{E}_h q_{max}(h) < q_{crit}$.

Turning our attention to the innovation substituting x_k by (3.1) in the definition of ε_k in (3.11) gives $\varepsilon_k = Ae_{k-1} + w_{k-1}$. The term e_{k-1} equals $(1 - \gamma_{k-1})\varepsilon_{k-1}$ as seen by (3.16), therefore ε_k evolves according to

$$\varepsilon_k = (1 - \gamma_{k-1})A\varepsilon_{k-1} + w_{k-1}, \qquad (3.17)$$

with initial value $\varepsilon_0 = 0$. The following proposition establishes a separation principle in our restricted information structure setup, stating that the control action has no effect on the quality of the future estimates at the controller.

Proposition 3.1. Consider any communication policy π selecting successful decoding probabilities q_k as functions of F_k given in (3.12), possible randomized, with ε_k defined in (3.11) and channel states h_k independently drawn from a distribution ϕ_h . Then at any step k the distributions of the future processes $\{\varepsilon_\ell, q_\ell, \gamma_\ell, e_\ell, \ell > k\}$ given G_k do not depend on the chosen control policy $\theta \in \Theta$.

Proof. First note that the processes $\{w_k, h_k, k \ge 0\}$ are by assumption independent of any other process. Then we follow an induction argument to prove the claim. At k = 0, ε_0 is equal to 0,

 q_0 depends only on h_0 and ε_0 , γ_0 is an independent Bernoulli with success q_0 , and e_0 is also 0 since x_0 is initially known. Consider then a time k with a given G_k , the corresponding estimation error e_k given G_k having zero mean and covariance Σ_k , and a control input u_k that is a function of G_k as described by the control policy θ . The term ε_{k+1} equals $Ae_k + w_k$, as indicated by the arguments preceding (3.17), which given G_k has mean 0 and covariance $A\Sigma_k A^T + W$. The choice $q_{k+1} \in F_{k+1}$ by construction depends on past variables in F_k which by causality do not depend on the action u_k , as well as the new variables ε_{k+1} , h_{k+1} which are also independent of u_k . Also the distribution of $\gamma_{k+1} \sim \text{Bern}(q_{k+1})$ only depends on the distribution of q_{k+1} , and the same holds for e_{k+1} , which equals $(1 - \gamma_{k+1})\varepsilon_{k+1}$ again by the arguments preceding (3.17). To sum up all variables ε_{k+1} , q_{k+1} , γ_{k+1} , e_{k+1} given G_k do not depend on u_k .

The intuition behind this proposition is that the effect of control inputs is subtracted from x_k when forming the innovation terms ε_k in (3.11) that are fed to the communication policy π . Similar separation results based on innovation terms have been utilized in other communication/control design problems (Molin and Hirche (2009); Ramesh et al. (2013); Nair et al. (2007)). The above proposition restates the separation principle for our power allocation problem under channel state information.

Since the power cost $J_{PWR}^N(\pi, \theta)$ in (3.6) only depends on pairs (q_k, h_k) , the above proposition shows that the control policy θ has no effect on the power cost. Thus we can rewrite the objective in (3.7) as

$$J(\pi,\theta) = \limsup_{N \to \infty} \frac{1}{N} J_{\text{LQR}}^N(\pi,\theta) + \lambda \limsup_{N \to \infty} \frac{1}{N} J_{\text{PWR}}^N(\pi).$$
(3.18)

This means that the optimal control policy $\theta \in \Theta$ for a given communication policy $\pi \in \Pi$ is the one minimizing the limit LQR cost. It turns out that the form of the optimal controller does not depend on the communication policy, leading to a stronger separability than what follows from (3.18).

Indeed by the above separation principle standard dynamic programming arguments show that the optimal control law for a finite horizon is given by the standard LQR one, as in, e.g., Molin and Hirche (2009); Ramesh et al. (2013). We are interested however in the infinite horizon problem. Our setup differs from the standard LQG/Kalman filtering problem with state observations y_k containing Gaussian noise, where the estimation error covariance Σ_k converges to some limit and the system is assumed to start at time k = 0 with this limit estimation error. In our setup whenever a packet is received the estimation error is reset to zero otherwise it grows (cf. (3.16)), so for the general communication policies $\pi \in \Pi$ under consideration it is not clear whether some limit covariance exists. Alternatively the following proposition shows that estimation errors admit a uniform bound in second moment.

Proposition 3.2. Suppose Assumption 3.1 holds. Then there exists a finite positive constant M such that for any communication policy $\pi \in \Pi$ selecting successful decoding probabilities q_k with respect to F_k given

in (3.12), possibly randomized, satisfying the additional restriction $q_k \in Q(\varepsilon_k, h_k)$ given by (3.14), and for every k = 0, 1, ..., it holds that

$$\mathbb{E}^{\pi} e_k^T e_k \le M. \tag{3.19}$$

With this bound on expected magnitude of estimation error established, uniform over *k* and over any policy $\pi \in \Pi$, the following theorem determines the optimal control law for the average infinite horizon problem (3.15).

Theorem 3.1 (Optimal control policy). Consider the wireless control system of Fig. 7 with any communication policy $\pi := \{q_0, q_1, \ldots\} \in \Pi$ selecting successful decoding probabilities q_k as functions of F_k given in (3.12), possibly randomized, with innovation terms ε_k as defined in (3.11) and channel states h_k independently drawn from a distribution ϕ_h , satisfying the additional restriction $q_k \in Q(\varepsilon_k, h_k)$ given by (3.14). Suppose Assumption 3.1 holds. Then for any control policy $\theta := \{u_0, u_1, \ldots\} \in \Theta$ composed of inputs u_k as possibly randomized functions of G_k in (3.9) such that

$$\lim_{N \to \infty} \frac{1}{N} \mathbb{E}^{\pi, \theta} x_N^T x_N = 0, \tag{3.20}$$

the joint objective $J(\pi, \theta)$ described by (3.5) - (3.7) satisfies

$$J(\pi,\theta) \ge Tr(PW) + \limsup_{N \to \infty} \frac{1}{N} \mathbb{E}^{\pi} \sum_{k=0}^{N-1} e_k^T \tilde{P} e_k + \lambda p(h_k, q_k)$$
(3.21)

where P is the solution to the standard algebraic Riccati equation $P = A^T P A + Q - A^T P B (R + B^T P B)^{-1} B^T P A$ for the system in (3.1) and the linear quadratic regulator cost (LQR) in (3.5), and the matrix \tilde{P} is defined as

$$\tilde{P} := A^T P A + Q - P. \tag{3.22}$$

Moreover, the minimum value in (3.21) is achieved for the control policy

$$u_k = K \hat{x}_k, \tag{3.23}$$

with \hat{x}_k the state estimate described in (3.10) and the steady state LQR gain $K := -(R + B^T P B)^{-1} B^T P A$.

The theorem determines the optimal control policy θ of problem (3.15) as the conventional LQR controller in (3.23), shown in Fig. 7. The optimal cost given in (3.21) equals a constant Tr(PW) and a limit average sum term that only depends on the communication policy $\pi \in \Pi$. This term shows that the optimal communication policy needs to balance the power expenditures with a weighted version of the estimation error at the controller.

Observe that as per (3.10) and (3.16) it holds that $e_k = (1 - \gamma_k)\varepsilon_k$. Also $\mathbb{E}^{\pi}[\gamma_k|F_k] = \mathbb{P}^{\pi}[\gamma_k = 1|F_k] = q_k$ and $\varepsilon_k \in F_k$. So we can write

$$\mathbb{E}^{\pi}[e_k^T \tilde{P}e_k | F_k] = \mathbb{E}^{\pi}[(1 - \gamma_k)\varepsilon_k^T \tilde{P}\varepsilon_k | F_k] = (1 - q_k)\varepsilon_k^T \tilde{P}\varepsilon_k,$$
(3.24)

and taking the expectation in both sides gives

$$\mathbb{E}^{\pi}[e_k^T \tilde{P}e_k] = \mathbb{E}^{\pi}[(1-q_k)\varepsilon_k^T \tilde{P}\varepsilon_k].$$
(3.25)

Substituting the expression (3.25) into the second summand of (3.21) it follows that the optimal communication policy $\pi \in \Pi$ of problem (3.15) is the one achieving the infimum cost

$$J_{\text{COMM}}^* := \inf_{\pi \in \Pi} \lim_{N \to \infty} \frac{1}{N} \mathbb{E}^{\pi} \sum_{k=0}^{N-1} c(\varepsilon_k, h_k, q_k),$$
(3.26)

where we define

$$c(\varepsilon, h, q) := (1 - q)\varepsilon^T \tilde{P}\varepsilon + \lambda p(h, q).$$
(3.27)

The difference between the sum in (3.21) and the objective in (3.26) is that in the former e_k is not known at the sensor at time k, while ε_k in the latter is. This way (3.26) takes the form of a Markov decision process (MDP) problem with an infinite horizon average cost criterion. The state of the problem at time k is the pair (ε_k , h_k) $\in \mathbb{R}^n \times \mathbb{R}_+$, the available action is $q_k \in Q(\varepsilon_k, h_k)$ by (3.14), and the cost-per-stage is $c(\varepsilon_k, h_k, q_k)$. The state transition probabilities can be obtained from (3.17) and are given by

$$\mathbb{P}(\varepsilon_+, h_+|\varepsilon, h, q) = [q \mathcal{N}_{0,W}(\varepsilon_+) + (1-q) \mathcal{N}_{A\varepsilon,W}(\varepsilon_+)] \phi_h(h_+).$$
(3.28)

Here ε , h and ε_+ , h_+ denote the current and next states respectively, and q the current action. When q is chosen at state (ε , h), a variable $\gamma \sim \text{Bern}(q)$ is drawn. By (3.17) on the event $\gamma = 1$, $\varepsilon_+ = w \sim \mathcal{N}_{0,W}$, while on the event $\gamma = 0$, $\varepsilon_+ = A\varepsilon + w$ with $w \sim \mathcal{N}_{0,W}$, which is equivalent to $\varepsilon_+ \sim \mathcal{N}_{A\varepsilon,W}$. Since h_+ is independent of ε , h, ε_+ , its distribution ϕ_h appears as a product in (3.28). We denote $\mathbb{E} [\varepsilon_+, h_+ | \varepsilon, h, q]$ the integration with respect to the above transition probability measure.

To sum up, we have exploited the proposed decoupling information structure to determine the optimal control policy as the standard LQR control input. We proceed in the following section to show that an optimal communication policy exists and we characterize its main features in the case of general FEC codes and in the special case of capacity achieving codes.

Remark 3.2. The technical condition (3.20) for the controller in Theorem 3.1 can be viewed as an additional stability condition requiring that the norm of the plant state grows at a sub-linear rate. Such conditions appear in general average cost optimal control problems, see e.g. (Bertsekas, 2005, Vol.II, p.254-5), and have also been used in average LQG problems (Bertsekas, 2005, Vol.II, p.272-3). This technical condition may potentially be relaxed under a different proof technique.

3.4. Optimal Communication Policy

Exploiting the MDP formulation of (3.26) we can show that optimal communication policies for the co-design problem in (3.15) exist. This existence result provides a characterization of these policies from which we infer the general features of optimal transmit powers p_k and corresponding successful decoding probabilities q_k as a function of innovation terms ε_k and channel realizations h_k .

The existence of optimal policies for average infinite-horizon MDPs on general state spaces requires some technical conditions (Hernández-Lerma and Lasserre (1996)). In our case restriction to communication policies $\pi \in \Pi$ that uniformly satisfy (3.14) guarantee existence, as the following theorem shows.

Theorem 3.2 (Optimal communication policy). Consider the Markov decision process with optimal cost as in (3.26), state transition probabilities as in (3.28), and actions restricted to $q_k \in Q(\varepsilon_k, h_k)$ with $Q(\varepsilon, h)$ abiding to (3.14). If Assumptions 3.1 and 3.2 hold true there exists a function $V : \mathbb{R}^n \times \mathbb{R}_+ \to \mathbb{R}$ such that for all $\varepsilon \in \mathbb{R}^n$ and $h \in \mathbb{R}_+$ it satisfies

$$V(\varepsilon, h) = \min_{q \in Q(\varepsilon, h)} \left\{ c(\varepsilon, h, q) - J_{COMM}^* + \mathbb{E} \left[V(\varepsilon_+, h_+) \, \big| \, \varepsilon, h, q \right] \right\}.$$
(3.29)

The optimal communication cost can be written as $J_{COMM}^* = \mathbb{E}_{w,h}V(w,h)$, where $\mathbb{E}_{w,h}$ denotes integration with respect to the product measure $\mathcal{N}_{0,W} \times \phi_h$. The optimal communication policy π^* achieving the minimum cost can be written as a function of the error and channel states at time k, $q_k^* = q^*(\varepsilon_k, h_k)$, and is the one achieving the minimum in the right hand side of (3.29), i.e.

$$q^{*}(\varepsilon, h) := \underset{q \in Q(\varepsilon, h)}{\operatorname{argmin}} \left\{ c(\varepsilon, h, q) - J_{COMM}^{*} + \mathbb{E} \left[V(\varepsilon_{+}, h_{+}) \, \big| \, \varepsilon, h, q \right] \right\}.$$
(3.30)

The theorem states that the optimal communication policy exists, is deterministic, and also stationary in the sense that q_k^* adapts only to the current state (ε_k , h_k) and not the complete history F_k in (3.12). The optimal policy is described by (3.30) in terms of a function $V(\varepsilon, h)$ that solves (3.29). Note that this function is unique up to a constant. Related characterizations of optimal communication policies when the decision is whether to transmit or not appear in Xu and Hespanha (2004); Mesquita et al. (2012). Our setup however differs since the decision is on the transmit power and this depends on the random wireless channel state. The proof of the theorem relies on constructing a Lyapunov-like function that is common for all policies $\pi \in \Pi$, and applying the MDP results of Vega-Amaya (2003). This methodology has been used in Mesquita et al. (2012), however a refined construction is required here to account for the random channel states.

An informal interpretation of condition (3.29) based on finite state spaces (Bertsekas, 2005) is

the following. Suppose the constant J^*_{COMM} , corresponding to the optimal cost of (3.26), was known. Subtracting this constant from the cost-per-stage in problem (3.26) does not change the optimal policy, and gives a relative cost per stage $c(\varepsilon, h, q) - J^*_{COMM}$ indicating how far we are from the optimal average cost. Then equation (3.29) has exactly the form of a standard Bellman equation for a non-averaged infinite horizon problem with this relative cost per stage (Bertsekas, 2005, Vol.I, Ch.7). The function $V(\varepsilon, h)$ captures the expected future relative cost of following the optimal policy when starting from state (ε, h) , and is termed the relative value function. Bellman's equation (3.29) states that the optimal choice q at every step minimizes the sum of the current-stage relative cost $c(\varepsilon, h, q) - J^*_{COMM}$ and the expected future relative cost $\mathbb{E}[V(\varepsilon_+, h_+) | \varepsilon, h, q]$. The minimization over the current action q gives again the value $V(\varepsilon, h)$ of the current state at the left hand side of (3.29).

In principle one can find $V(\varepsilon, h)$ using value iteration or policy iteration algorithms which involve iterative application of (3.29) (Hernández-Lerma and Lasserre, 1996). This procedure is, however, computationally onerous as each iteration requires minimizing the right hand side of (3.29) for all possible state pairs $(\varepsilon, h) \in \mathbb{R}^n \times \mathbb{R}_+$. Nevertheless, (3.29) still gives qualitative information on the optimal policy.

Let us ignore the case $\|\varepsilon\| \ge L, h \ge h_t$ in (3.14) as it is irrelevant for the following discussion. Integrating $V(\varepsilon, h)$ with respect to the transition (3.28) gives

$$\mathbb{E}\left[V(\varepsilon_+, h_+)|\varepsilon, h, q\right]$$

= $q\mathbb{E}_{w,h_+}V(w, h_+) + (1-q)\mathbb{E}_{w,h_+}V(A\varepsilon + w, h_+).$ (3.31)

We substitute this, the cost-per-stage $c(\varepsilon, h, q)$ defined by (3.27), and the expression $J^*_{\text{COMM}} = \mathbb{E}_{w,h}V(w,h)$ provided by the theorem in the minimization of (3.30), and upon reordering terms, the optimal communication policy can be written as

$$q^*(\varepsilon,h) = \underset{q \in [0,q_{\max}(h)]}{\operatorname{argmin}} \lambda p(h,q) + (1-q)R(\varepsilon),$$
(3.32)

where for convenience we defined the function

$$R(\varepsilon) := \mathbb{E}_{w,h} \left[V(A\varepsilon + w, h) - V(w, h) \right] + \varepsilon^T \tilde{P}\varepsilon,$$
(3.33)

which can be thought as a penalty function on the error ε . The optimal policy $q^*(\varepsilon, h)$ depends on the shape of the function p(h, q) and takes values anywhere in the interval $[0, q_{\max}(h)]$. The optimal power allocation can be found by converting (3.32) to power by (3.2), (3.3), and is described by

$$p^*(\varepsilon, h) := \operatorname*{argmin}_{p \in [0, p_{\max}]} \lambda p + (1 - q(h, p)) R(\varepsilon).$$
(3.34)

Despite the fact that $V(\varepsilon, h)$ and $R(\varepsilon)$ are hard to compute, the above expression is an important characterization of the optimal power allocation. It provides a tool for qualitative analysis of



Figure 8: Optimal power allocation for FEC codes with different complementary error functions. The optimal transmit power p^* is plotted as a function of the factor $R(\varepsilon)$ for a fixed channel state h using FEC codes with different q-SNR characteristics. When the q-SNR curve becomes steeper, the optimal power allocation resembles a step function.

different FEC codes in wireless control systems. We illustrate this in Fig. 8 where we examine how the *q*-SNR relationship of a FEC code affects the optimal power allocation. For simplicity we fix the channel state *h* and plot p^* in Fig. 8 as a function of $R(\varepsilon)$. In all cases, when the error penalty $R(\varepsilon)$ is below some threshold, the best option is to not transmit. Above the threshold, the optimal transmit power increases as the error penalty $R(\varepsilon)$ gets larger. For powerful FEC codes characterized by a steep *q*-SNR relationship, close to the theoretical limit in (2.19), the optimal power allocation resembles a step function, since the probability of successful decoding becomes practically one for large powers. For fat *q*-SNR tails, this behavior deteriorates as the sensor needs to transmit with higher power to achieve a larger *q*.

Then in Fig. 9 we present qualitative plots of the optimal decoding probability q^* and optimal transmit power p^* as functions of both the factor $R(\varepsilon)$ and the channel state h for a given q-SNR characteristic. The blue region indicates the event where no transmission occurs. This happens if channel gain h is low, where transmission is costly, or if error ε has a low penalty, meaning that there is no need to update the receiver's estimate. This no-transmission region becomes larger for a higher penalty λ on power in (3.7). Outside this region a transmission occurs and transmit power adapts to both channel and error states. In principle when channel gain h is high, a small amount of power suffices. For intermediate values of channel h power takes a wide range of values depending on the error as well.

Overall this optimal power management displays different features than the standard "0-1" eventtriggered transmission paradigm of, e.g., Xu and Hespanha (2004) or Tabuada (2007). It can be though as a 'soft" version of these policies since the power decision ranges between $[0, p_{max}]$, or equivalently the decoding *q* between $[0, q_{max}(h)]$. Finally we note that the transmit power/remote estimation problem has also been studied in the very recent works of Quevedo et al. (2012); Leong and Dey (2012), which however consider only power adaptation to the channel and the packet drop processes, not the plant state/error observed online. Hence the qualitative characterization we discuss here and the connections with the event-triggered paradigm were not apparent.



Figure 9: Optimal decoding probability and power allocation for a FEC code. Color intensity indicates the magnitude of optimal decoding probability q^* and optimal transmit power p^* as functions of the factor $R(\varepsilon)$ and the channel state h.

Since computing the optimal power allocation becomes a computationally hard problem, tractable suboptimal power allocation policies are developed in Chapter 6. In the following chapter we consider the problem of multiple wireless control systems sharing a wireless medium and design appropriate resource-aware and control-aware scheduling policies.

3.4.1. Optimal solution for capacity achieving codes

Consider now the case of capacity achieving codes. By (2.19), at time *k* the transmitter needs to use either $p_k = 0$, i.e. not transmitting, or $p_k = p_0/h_k$ with $p_0 := N_0 \text{SNR}_0$, which certainly guarantees correct packet delivery. Any other power allocation is unfavorable. However the instantaneous power is bounded by $p_k \leq p_{\text{max}}$, so the sensor can transmit only when $p_0/h_k \leq p_{\text{max}}$, or equivalently when the channel state exceeds $h_k \geq p_0/p_{\text{max}}$.

In this case we are looking again for a randomized policy, i.e. a distribution on the two power options $\{0, p_0/h_k\}$ when $h_k \ge p_0/p_{max}$. With a slight abuse of notation we denote $q_k \in [0, 1]$ the probability of choosing power p_0/h_k . Then when $h_k \ge p_0/p_{max}$ the transmitter draws independent $\gamma_k \sim \text{Bern}(q_k)$ and transmits with power $p_k = \gamma_k p_0/h_k$. The decoding success at the receiver is given by the same γ_k . The expected power consumption becomes

$$\mathbb{E}\sum_{k=0}^{N-1} p_k = \mathbb{E}\sum_{k=0}^{N-1} q_k \frac{p_0}{h_k} \mathbb{I}\left(h_k \ge \frac{p_0}{p_{\max}}\right).$$
(3.35)

Observe that this is of the same form as the expected power consumption of the original problem given in (3.6) with the function p(h,q) substituted with $q p_0/h \mathbb{I}$ ($h \ge p_0/p_{max}$). Then the statements of the results so far hold for the capacity achieving codes as well. For this special case of p(h,q) however the minimization in (3.32) becomes linear in q, and the optimal communication policy is deterministic,

$$q^{CA}(\varepsilon,h) := \begin{cases} 0 & \text{if } h R(\varepsilon) \le \lambda p_0 \text{ or } h \le p_0 / p_{\max} \\ 1 & \text{otherwise} \end{cases}$$
(3.36)

or in terms of power

$$p^{CA}(\varepsilon,h) := \begin{cases} 0 & \text{if } h R(\varepsilon) \le \lambda p_0 \text{ or } h \le p_0/p_{\max} \\ p_0/h & \text{otherwise} \end{cases}$$
(3.37)

This is an event-triggered transmission scheme along the lines of, e.g., Xu and Hespanha (2004), except that now the decision is also affected by the current channel state h apart from the error ε . This deterministic policy was expected as the limit behavior of powerful FEC codes in Fig. 8. The region of the plant/channel state space $\mathbb{R}^n \times \mathbb{R}_+$ where it is optimal to transmit is described in (3.36) as $h R(\varepsilon) > \lambda p_0$ and $h \ge p_0/p_{\text{max}}$. Intuitively condition $h R(\varepsilon) > \lambda p_0$ states that when channel gain is large, transmitting is worthy as it does not cost much, while when an error penalty $R(\varepsilon)$ is large, it is necessary to transmit in order to reset it to zero. This region gets smaller when the constant $p_0 = N_0 \text{SNR}_0$ (cf. (2.19)) increases, since transmission then requires more power, or when λ increases, since power then is penalized more in the objective (3.7).

3.5. Proofs

3.5.1. Proof of Proposition 3.2

First note that, by the same arguments we use to derive (3.25) later, conditioned on F_k we can rewrite

$$\mathbb{E}^{\pi} e_k^T e_k = \mathbb{E}^{\pi} (1 - q_k) \varepsilon_k^T \varepsilon_k.$$
(3.38)

The bound in (3.19) will be shown by an equivalent bound on the innovation process { $\varepsilon_k, k \ge 0$ }. By Proposition 3.1 for any communication policy $\pi \in \Pi$ this process is independent of the control policy $\theta \in \Theta$ and its evolution is given by (3.17). This evolution can be described more formally along with the i.i.d. channel process $h_k \sim \phi_h$ by a stochastic transition kernel given the values of ε, h and decision q at each step as

$$\mathbb{P}(\varepsilon_+, h_+|\varepsilon, h, q) = [q \mathcal{N}_{0,W}(\varepsilon_+) + (1-q) \mathcal{N}_{A\varepsilon,W}(\varepsilon_+)] \phi_h(h_+).$$
(3.39)

This expression is included again in (3.28), where its derivation is explained in detail.

The following technical lemma shows that under Assumption 3.1 one can construct a Lyapunovlike function common for all communication policies, satisfying explicitly the technical requirements of (Vega-Amaya, 2003, Assumptions 3.1, 3.2). The uniform bound (3.19) will be a direct consequence of these requirements. We note also that the lemma will be subsequently used to prove Theorem 3.2 based on the results of Vega-Amaya (2003).

Lemma 3.1. Suppose Assumption 3.1 holds and consider the innovation and channel processes $\{\varepsilon_k, h_k, k \ge 0\}$ described by the transition (3.39), with communication decisions satisfying $q_k \in Q(\varepsilon_k, h_k)$ given in (3.14). Then there exists a measurable function W on $\mathbb{R}^n \times \mathbb{R}_+$ bounded below by a constant $\gamma > 0$ such that

$$(1-q)\varepsilon^{T}\varepsilon + c \le KW(\varepsilon, h), \tag{3.40}$$

where $c \ge 0$ is some constant, for all $\varepsilon, h \in \mathbb{R}^n \times \mathbb{R}_+$, $q \in Q(\varepsilon, h)$, for some positive K. Moreover there exists a non-trivial measure v on $\mathbb{R}^n \times \mathbb{R}_+$, a non-negative measurable function $r(\varepsilon, h, q) \ge 0$ for $\varepsilon, h \in \mathbb{R}^n \times \mathbb{R}_+$, $q \in Q(\varepsilon, h)$, and a positive constant $\mu < 1$ such that

$$\begin{aligned} (i) \ \nu(W) &:= \int W(\varepsilon, h) d\nu(\varepsilon, h) < \infty, \\ (ii) \ \mathbb{P}(\varepsilon_+ \in B_1, h_+ \in B_2 | \varepsilon, h, q) \ge \nu(B_1, B_2) r(\varepsilon, h, q) \\ for all measurable subsets (B_1, B_2) &\in \mathcal{B}(\mathbb{R}^n \times \mathbb{R}_+), \\ (iii) \ \mathbb{E} \left[W(\varepsilon_+, h_+) | \varepsilon, h, q \right] \le \mu W(\varepsilon, h) + r(\varepsilon, h, q) \nu(W) \\ (iv) \ \int r(\varepsilon, h, q) d\nu(\varepsilon, q) > 0 \ for \ all \ q \in Q(\varepsilon, h). \end{aligned}$$

Proof. The proof is constructive. Let

$$\nu := \mathcal{N}_{0,W} \times \phi_h \text{ and } r(\varepsilon, h, q) := q.$$
(3.41)

Let us denote the set where the choice of q is free as

$$S := \{ (\varepsilon, h) \in \mathbb{R}^n \times \mathbb{R}_+ : \|\varepsilon\| < L \text{ or } h < h_t \}.$$
(3.42)

We choose $\mu < 1$ such that

$$\mu > 1 - q_{\max}(h_t) + q_{\max}(h_t)\nu(S), \tag{3.43}$$

$$\mu > (1 - \bar{q}) \lambda_{\max}(A)^2, \tag{3.44}$$

where \bar{q} denotes the integral introduced in (3.13),

$$\bar{q} := \int_{h_t}^{+\infty} q_{\max}(h) d\phi_h(h). \tag{3.45}$$

The right hand side of (3.43) is less than 1 because the event *S* under the measure ν happens with probability less than 1 and we have assumed $q_{\max}(h_t) > 0$. The right hand side of (3.44) is also less than 1 because of Assumption 3.1 and by the choice for h_t that satisfies (3.13).

For future reference note that by construction of the set $Q(\varepsilon, h)$, for any L > 0, when $\|\varepsilon\| \ge L$ we can upper bound

$$1 - q \le 1 - q_{\max}(h) \mathbb{I}(h \ge h_t) =: \psi(h), \tag{3.46}$$

where we named the quantity on the right $\psi(h)$ to be used within this proof. This inequality holds because when $h < h_t$, we have $q \ge 0$, and when $h \ge h_t$, we choose $q = q_{\max}(h)$.

Finally we pick

$$W(\varepsilon, h) := \psi(h)\varepsilon^{T}H\varepsilon + \beta \mathbb{I}(\varepsilon, h \in S) + \gamma, \qquad (3.47)$$

where β , $\gamma > 0$ are appropriate positive constants that will be designed next, and $H \succ 0$ is a positive definite matrix satisfying

$$(1 - \bar{q})A^T H A - \mu H = -\Theta, \qquad (3.48)$$

for some positive definite matrix $\Theta > 0$. This Lyapunov equation is feasible by our choice of μ that satisfies (3.44).

Next we show that the conditions of the lemma are satisfied for the constructed quantities. First observe that $W(\varepsilon, h) \ge \gamma > 0$ by construction. Then we check (3.40). When $\|\varepsilon\| < L$,

$$(1-q)\varepsilon^{T}\varepsilon + c \le L^{2} + c \le K(\beta + \gamma) \le KW(\varepsilon, h),$$
(3.49)

for a sufficiently large *K*, where the last inequality follows from the form of $W(\varepsilon, h)$ on $\|\varepsilon\| < L$. On the other hand if $\|\varepsilon\| \ge L$, we may use (3.46) to upper bound

$$(1-q)\varepsilon^{T}\varepsilon + c \leq \psi(h)\varepsilon^{T}\varepsilon + c$$

$$\leq K(\psi(h)\varepsilon^{T}H\varepsilon + \gamma) \leq KW(\varepsilon, h), \qquad (3.50)$$

for a sufficiently large *K*, by our choice for the function $W(\varepsilon, h)$ when $\|\varepsilon\| \ge L$.

We proceed to show that parts (i)-(iv) in the statement of the lemma also hold. Part (i) holds because the integral of $W(\varepsilon, h)$ with our chosen measure ν equals

$$\nu(W) = (1 - \bar{q})Tr(HW) + \nu(S)\beta + \gamma < \infty.$$
(3.51)

Part (ii) holds because the transition probability in (3.39) gives

$$\mathbb{P}(\varepsilon_{+} \in B_{1}, h_{+} \in B_{2}|\varepsilon, h, q)
= [q \mathcal{N}_{0,W}(B_{1}) + (1-q) \mathcal{N}_{A\varepsilon,W}(B_{1})] \phi_{h}(B_{2})
\geq q \mathcal{N}_{0,W}(B_{1}) \phi_{h}(B_{2}) = r(\varepsilon, h, q)\nu(B_{1}, B_{2}).$$
(3.52)

Part (iv) follows by our choice $r(\varepsilon, h, q) = q$ and the construction of the set $Q(\varepsilon, h)$ in (3.14) because

$$\int r(\varepsilon, h, q) d\nu(\varepsilon, h) \ge \int_{\varepsilon, h \in S^c} q_{\max}(h) d\nu(\varepsilon, h)$$

= $\bar{q} \int_{\|\varepsilon\| \ge L} d\mathcal{N}_{0, W}(\varepsilon) > 0.$ (3.53)

The remainder of the proof shows that (iii) also holds. First observe that by the transition defined in (3.39) and our choices for the measure ν and the function r(.) we have

$$\mathbb{E}\left[W(\varepsilon_{+},h_{+})|\varepsilon,h,q\right] = r(\varepsilon,h,q)\nu(W) + (1-q)\int W(\varepsilon_{+},h_{+})d\mathcal{N}_{A\varepsilon,W}(\varepsilon_{+}) d\phi_{h}(h_{+}).$$
(3.54)

Substituting (3.54) in (iii), we only need to show that

$$(1-q)\int W(\varepsilon_{+},h_{+})d\mathcal{N}_{A\varepsilon,W}(\varepsilon_{+})\,d\,\phi_{h}(h_{+})\leq \mu W(\varepsilon,h).$$
(3.55)

Plugging the expression of $W(\varepsilon, h)$ given by (3.47) in the integral of the left hand side, condition (3.55) becomes

$$(1-q) \left\{ (1-\bar{q}) \left[\varepsilon^{T} A^{T} H A \varepsilon + Tr(HW) \right] + \beta \mathcal{N}_{A\varepsilon,W} \times \phi_{h}(S) + \gamma \right\} \leq \mu W(\varepsilon, h).$$
(3.56)

We can bound $\mathcal{N}_{A\varepsilon,W} \times \phi_h(S) \leq \nu(S)$ for any $\varepsilon \in \mathbb{R}^n$, and also $(1-q)(1-\bar{q})Tr(HW) \leq Tr(HW)$. So a sufficient condition for (3.56) is to show that

$$(1-q)\left\{(1-\bar{q})\varepsilon^{T}A^{T}HA\varepsilon + \beta\nu(S) + \gamma\right\} + Tr(HW)$$

$$\leq \mu W(\varepsilon, h)$$
(3.57)

holds for every choice of $q \in Q(\varepsilon, h)$. To show this we examine cases.

Case $\|\varepsilon\| \ge L$. Using (3.46) to upper bound $1 - q \le \psi(h)$, and upon substituting $W(\varepsilon, h)$ in (3.57) and rearranging terms, we need to show equivalently that

$$\psi(h) \left\{ \varepsilon^{T} \left[(1 - \bar{q}) A^{T} H A - \mu H \right] \varepsilon + \beta \nu(S) + \gamma \right\} + Tr(HW) \leq \mu \left\{ \beta \mathbb{I} \left(h < h_{t} \right) + \gamma \right\}$$
(3.58)

By the choice of *H* in (3.48) the quadratic on the left hand side is negative definite equal to to $-\varepsilon^T \Theta \varepsilon$. And since $\|\varepsilon\| \ge L$ we can upper bound $-\varepsilon^T \Theta \varepsilon \le -\lambda_{\min}(\Theta)L^2 \le 0$. After these, a sufficient condition for (3.58) is

$$Tr(HW) + \psi(h) \left\{ \beta \nu(S) + \gamma \right\} \le \mu \left\{ \beta \mathbb{I} \left(h < h_t \right) + \gamma \right\}$$
(3.59)

Now consider two cases for *h*. If $h < h_t$ condition (3.59) becomes

$$Tr(HW) + \beta \nu(S) + \gamma \le \mu(\beta + \gamma).$$
(3.60)

On the other hand if $h \ge h_t$ we have that $q_{\max}(h) \ge q_{\max}(h_t)$ by monotonicity of q_{\max} , so we may bound $\psi(h) = 1 - q_{\max}(h) \le 1 - q_{\max}(h_t)$. Condition (3.59) becomes

$$Tr(HW) + (1 - q_{\max}(h_t)) \left\{ \beta \nu(S) + \gamma \right\} \le \mu \gamma.$$
(3.61)

We pick a γ > to satisfy (3.61) with equality, that is

$$\gamma = \frac{(1 - q_{\max}(h_t))\nu(S)\beta + Tr(HW)}{\mu - (1 - q_{\max}(h_t))}$$
(3.62)

where the denominator is positive by the choice of μ in (3.43). We will show that condition (3.60) also holds by an appropriate choice for $\beta > 0$.

Case $\|\varepsilon\| < L$. In this case $q \ge 0 \Rightarrow 1 - q \le 1$ and it is sufficient for (3.57) to show that

$$\sup_{\|\varepsilon\| < L} \varepsilon^{T} (1 - \bar{q}) A^{T} H A \varepsilon + Tr(HW) + \beta \nu(S) + \gamma$$

$$\leq \mu(\beta + \gamma)$$
(3.63)

holds, where on the right hand side we lower bounded the quadratic term of *W* by 0. This is of the general form

$$C_2 + \beta \nu(S) + \gamma \le \mu(\beta + \gamma) \tag{3.64}$$

for some constant C_2 , and recall that the left over condition (3.60) is of the same form. Plugging the chosen γ by (3.62) in (3.64) leads to a condition of the form

$$C_{3} \leq \left(1 - \frac{q_{\max}(h_{t})\,\nu(S)}{\mu - (1 - q_{\max}(h_{t}))}\right)\,\mu\beta,\tag{3.65}$$

for some constant C_3 . We want to be able select $\beta > 0$ that satisfies (3.65) for any value of the constant C_3 . Hence we require the coefficient of β to be *strictly* positive. This turns out to be equivalent to $\mu > 1 - q_{\max}(h_t) + q_{\max}(h_t)\nu(S)$, which corresponds to our choice of μ in (3.43). Therefore we conclude that part (iii) of the lemma holds as well and this completed the proof.

Returning to the proof of Proposition 3.2, combining (3.38) with condition (3.40) of the above Lemma we have that $\mathbb{E}^{\pi} e_k^T e_k \leq K \mathbb{E}^{\pi} W(\varepsilon_k, h_k)$, so it suffices for (3.19) to show that a uniform bound on the expected value of $W(\varepsilon_k, h_k)$ exists.

By condition (ii) of the above lemma for $(B_1, B_2) = (\mathbb{R}^n, \mathbb{R}_+)$ we have that $r(\varepsilon, h, q) \leq 1/\nu(\mathbb{R}^n, \mathbb{R}_+)$.

Plugging this in (iii) leads to

$$\mathbb{E}\left[W(\varepsilon_+, h_+)|\varepsilon, h, q\right] \le \mu W(\varepsilon, h) + \nu(W)/\nu(\mathbb{R}^n, \mathbb{R}_+)$$
(3.66)

Iterated applications of this inequality across some policy $\pi \in \Pi$ yields

$$\mathbb{E}^{\pi}W(\varepsilon_k, h_k) \le \mu^k \mathbb{E} W(\varepsilon_0, h_0) + \frac{\nu(W)}{(1-\mu)\nu(\mathbb{R}^n, \mathbb{R}_+)}$$
(3.67)

Thus since $\mu < 1$ a uniform bound on $\mathbb{E}^{\pi}W(\varepsilon_k, h_k)$ exists and this completes the proof.

3.5.2. Proof of Theorem 3.1

First note that since $\Sigma_k := \mathbb{E}^{\pi} \left[e_k e_k^T \mid G_k \right]$ we have that

$$\mathbb{E}^{\pi}[Tr(\Sigma_k)] = \mathbb{E}^{\pi} e_k^T e_k.$$
(3.68)

Then under Assumption 3.1, Proposition 3.2 states that for any $\pi \in \Pi$ condition (3.19) holds and guarantees that both quantities in (3.68) are bounded uniformly over *k*.

To establish the optimality of the proposed control law we use the fact that the Bellman-like equation

$$V(G_k) + Tr(PW) + Tr(\tilde{P}\Sigma_k) =$$

$$\min_{u_k} \mathbb{E}^{\pi} \left[x_k^T Q x_k + u_k^T R u_k + V(G_{k+1}) | G_k, u_k \right], \qquad (3.69)$$

is satisfied for the function

$$V(G_k) = \mathbb{E}^{\pi} \left[x_k^T P x_k \, \big| \, G_k \right], \tag{3.70}$$

with $V(G_0) = x_0^T P x_0$, where *P* is the solution to the standard algebraic Riccati equation and \tilde{P} is given by (3.22). The existence of *P* is guaranteed by the stabilizability of (A, B) and detectability of $(A, Q^{1/2})$.

Indeed observe that we can use the tower property to rewrite the term on the right hand side of (3.69) as

$$\mathbb{E}^{\pi} [V(G_{k+1})|G_k, u_k] = \mathbb{E}^{\pi} \left[x_{k+1}^T P x_{k+1} |G_k, u_k \right]$$

= $\mathbb{E}^{\pi} \left[(Ax_k + Bu_k)^T P (Ax_k + Bu_k) |G_k, u_k \right] + Tr(PW),$ (3.71)

where the last equality follows by substituting x_{k+1} from the system equation (3.1). The quadratic minimization over u_k at the right hand side of (3.69) takes the standard form appearing in LQR problems with partial state information - see e.g. (Bertsekas, 2005, Vol. II, Section 5.2). The argument of the minimization in (3.69) is given by the control law (3.23). Straightforward substitutions

show that the optimal value of the minimization equals the left hand side of (3.69).

The equation (3.69) can be used to show that the optimal control policy is (3.23). First iterate (3.69) for k = 0, ..., N - 1 across some control policy $\theta \in \Theta$ to get

$$V(G_0) + N \operatorname{Tr}(PW) + \mathbb{E}^{\pi} \sum_{k=0}^{N-1} \operatorname{Tr}(\tilde{P}\Sigma_k)$$

$$\leq J_{LQR}^N(\pi, \theta) + \mathbb{E}^{\pi, \theta} V(G_N)$$
(3.72)

Dividing (3.72) by N and taking the limit as $N \to \infty$, the term on the left hand side tends to

$$\limsup_{N \to \infty} 1/N \left[x_0^T P x_0 + N \operatorname{Tr}(PW) + \mathbb{E}^{\pi} \sum_{k=0}^{N-1} \operatorname{Tr}(\tilde{P}\Sigma_k) \right]$$
$$= \operatorname{Tr}(PW) + \limsup_{N \to \infty} 1/N \mathbb{E}^{\pi} \sum_{k=0}^{N-1} e_k^T \tilde{P} e_k$$
(3.73)

where we used (3.68) to convert Σ_k to e_k .

Then consider the term on the right hand side of (3.72). Any control policy $\theta \in \Theta$ satisfying (3.20) also satisfies

$$\lim_{N \to \infty} \frac{1}{N} \mathbb{E}^{\pi, \theta} V(G_N) = \lim_{N \to \infty} \frac{1}{N} \mathbb{E}^{\pi, \theta} x_N^T P x_N = 0$$
(3.74)

by the form of *V* given in (3.70). Thus taking the limit as $N \to \infty$, by (3.74) the term on the right hand side of (3.72) tends to the average LQR cost. The inequality in (3.72) then shows that the average LQR cost of θ is larger or equal to the limit of the left hand side which was given in (3.73). The result (3.21) of the theorem follows by including the power cost that depends only on the communication policy π as suggested by (3.18).

The final step of the proof is to show that the control policy θ^* defined by (3.23) gives exactly the LQR cost given in (3.73). This policy satisfies (3.69) with equality, so (3.72) also holds with equality for θ^* . Dividing by *N* and taking the limit as before would prove the desired result if condition (3.74) also holds for θ^* . We next verify that this is the case.

Indeed use $u_k^* = K \hat{x}_k$ and $x_k = \hat{x}_k + e_k$ to rewrite the closed loop system equation (3.1) under θ^* as

$$x_{k+1} = (A + BK)\hat{x}_k + Ae_k + w_k. \tag{3.75}$$

Then denoting $\rho := \lambda_{\max}(A + BK)$ which is stable, $\rho < 1$, we can upper bound (3.71) under θ^* by

$$\mathbb{E}^{\pi,\theta^*} \left[V(G_{k+1}) | G_k \right]$$

$$\leq \rho^2 \hat{x}_k^T P \hat{x}_k + Tr(A^T P A \Sigma_k) + Tr(PW)$$

$$= \rho^2 V(G_k) + Tr((A^T P A - \rho^2 P) \Sigma_k) + Tr(PW)$$
(3.76)

where in the last equality we used the fact that $\hat{x}_k \hat{x}_k^T = \mathbb{E}^{\pi} [x_k x_k^T | G_k] - \Sigma_k$. Taking expectation on both sides of (3.76) we have that

$$\mathbb{E}^{\pi,\theta^*} V(G_{k+1}) \leq \rho^2 \mathbb{E}^{\pi,\theta^*} V(G_k) + Tr((A^T P A - \rho^2 P) \mathbb{E}^{\pi,\theta^*} \Sigma_k) + Tr(PW)$$
(3.77)

But (3.68) and (3.19) imply that $\mathbb{E}^{\pi,\theta^*}\Sigma_k$ is uniformly bounded over k so the term on the second line of (3.77) is bounded by some constant $\delta < \infty$. Iterating the above inequality (3.77) across θ^* up to k = N yields

$$\mathbb{E}^{\pi,\theta^*}\left[x_N^T P x_N\right] \le \rho^{2N} x_0^T P x_0 + \delta/(1-\rho^2)$$
(3.78)

which guarantees the limit (3.74) since $\rho < 1$.

3.5.3. Proof of Theorem 3.2

The proof of the theorem is a direct application of the theorems contained in Vega-Amaya (2003). For these we need to show that (Vega-Amaya, 2003, Assumptions 3.1, 3.2, 3.4) hold in our case. In particular (Vega-Amaya, 2003, Assumption 3.1) requires that the cost per stage is bounded $|c(\varepsilon, h, q)| \leq KW(\varepsilon, h)$ by a positive measurable function *W*. This is a consequence of (3.40) of Lemma 3.1, since

$$|c(\varepsilon, h, q)| \le (1 - q)\lambda_{\max}(\tilde{P})\varepsilon^{T}\varepsilon + \lambda p_{\max}$$
(3.79)

which is of the same form as (3.40). Also (Vega-Amaya, 2003, Assumption 3.2) requires exactly the conditions given in (i)-(iv) of Lemma 3.1. Finally (Vega-Amaya, 2003, Assumption 3.4) requires the following conditions for the functions *W* and *r* satisfying Lemma 3.1.

Assumption 3.3. *For every* $\varepsilon \in \mathbb{R}^n$ *,* $h \in \mathbb{R}_+$

- (*i*) $Q(\varepsilon, h)$ is compact,
- (ii) $c(\varepsilon, h, q)$ is lower semi-continuous in $q \in Q(\varepsilon, h)$,
- (iii) $\mathbb{P}(\varepsilon_+, h_+|\varepsilon, h, q)$ is strongly continuous ¹ in $q \in Q(\varepsilon, h)$,
- (iv) the mapping $q \to \mathbb{E}[W(\varepsilon_+, h_+)|\varepsilon, h, q]$ is continuous,
- (v) $r(\varepsilon, h, q)$ is continuous in $q \in Q(\varepsilon, h)$.

Part (i) is trivial, and (ii) is a consequence of the continuity of p(h, q) by Assumption 3.2. Strong continuity in (iii) is guaranteed by the fact that the transition kernel given in (3.28) has a probability density function. Part (iv) holds because the transition (3.28) is linear in q, and (v) is trivial.

¹i.e. for every bounded measurable function Ψ on $\mathbb{R}^n \times \mathbb{R}_+$, the mapping $q \mapsto \mathbb{E}[\Psi(\varepsilon_+, h_+)|\varepsilon, h, q]$ is continuous

Having established (Vega-Amaya, 2003, Assumptions 3.1, 3.2, 3.4), then (Vega-Amaya, 2003, Theorems 3.5, 3.6) state that in our case the infimum J^*_{COMM} in (3.26) exists, there exists a function $V(\varepsilon, h)$ that satisfies (3.29), and the optimal policy is the minimizer of the right hand side of (3.29) as given in (3.30).

Finally note that (3.29) holds if we add any constant to $V(\varepsilon, h)$, so without loss of generality we may take $V(0, \hat{h}) = 0$ for some \hat{h} . Then for $\varepsilon = 0, h = \hat{h}$ (3.29) gives

$$V(0,\hat{h}) = 0 = \min_{q \in Q(0,\hat{h})} \left\{ c(0,\hat{h},q) - J_{\text{COMM}}^* + \mathbb{E} \left[V(\varepsilon_+,h_+) \, \big| \, 0, \hat{h},q \right] \right\}.$$
(3.80)

Note that by (3.28), $\mathbb{P}(\varepsilon_+, h_+|0, \hat{h}, q) = \mathcal{N}_{0,W}(\varepsilon_+) \phi_h(h_+)$, and also $c(0, \hat{h}, q) = \lambda p(\hat{h}, q)$ by (3.27), so (3.80) becomes

$$0 = \min_{q \in Q(0,\hat{h})} \left\{ \lambda p(\hat{h}, q) - J_{\text{COMM}}^* + \mathbb{E}_{w,h} V(w, h) \right\}.$$
(3.81)

The minimizer is q = 0, giving the optimal value $J^*_{\text{COMM}} = \mathbb{E}_{w,h}V(w,h)$ provided in the statement of the theorem.

Chapter 4: Opportunistic Control over Multiple Access Channels

4.1. Problem Description

In this chapter we consider a wireless control architecture with multiple control systems over a shared wireless medium. The architecture, introduced in Section 2.2.1, is repeated here as well in Fig. 10. The setup consists of *m* independent networked control systems. Each control loop *i* (i = 1, 2, ..., m) includes a wireless transmitter communicating to a common receiver/access point. For example this can be a wireless sensor transmitting plant measurements to a common controller computing the control inputs to the plants. A centralized scheduler, implemented at the access point, decides which control system is given access to the shared wireless channel.

We denote the state of system *i* at each time *k* by $x_{i,k} \in \mathbb{R}^{n_i}$. The evolution of each system at time *k* depends on whether a transmission occurs at time *k* or not for link/system *i*, hereby denoted as $\gamma_{i,k} \in \{0,1\}$. We assume then that each system is described by a known switched linear time-invariant system of the form

$$x_{i,k+1} = \begin{cases} A_{c,i} x_{i,k} + w_{i,k}, & \text{if } \gamma_{i,k} = 1\\ A_{o,i} x_{i,k} + w_{i,k}, & \text{if } \gamma_{i,k} = 0 \end{cases}$$
(4.1)

This model was introduced in Section 2.1 where examples of such control systems were presented. We assume that the closed-loop matrix $A_{c,i}$ is asymptotically stable for each system *i*, implying that if system *i* were to transmit at each slot its respective state evolution is stable. The open loop matrices $A_{o,i}$ may be unstable. The additive terms $w_{i,k}$ model an independent (both across time *k* for each plant *i*, and across plants) identically distributed (i.i.d.) noise process with mean zero and covariance $W_i \succeq 0$.

Let us now describe the wireless communication system and model how it determines the packet transmission success $\gamma_{i,k}$. Suppose there are f different frequencies that each system may use to communicate to the access point and let the wireless channel fading conditions for a system i and frequency j at time slot k be denoted as $h_{ij,k}$. Channel conditions take values in a subset $\mathcal{H} \subseteq \mathbb{R}_+$ of the positive reals. We adopt a block fading model whereby channel states $\{h_{ij,k}, 1 \leq i \leq m, 1 \leq j \leq f\}$ are modeled as constant during each transmission slot k, but independent and identically distributed across different time slots k according to some joint distribution ϕ on $\mathcal{H}^{m \times f}$. They are also independent of the plant process noise $w_{i,k}$. We assume the channel states are available to the access point before transmission – see Remark 2.5 for a practical implementation. We also make



Figure 10: Architecture for control over multiple access channels. Independent control systems close their loops by transmitting over the shared wireless medium to a common receiver/access point. Each control system *i* experiences random channel conditions h_i . A centralized scheduler at the access point observes all channel states and opportunistically decides which system is scheduled to transmit and close the loop.

the following technical assumption on their joint distribution to exclude the possibility of channel states becoming degenerate random variables.

Assumption 4.1. The joint distribution ϕ of channel states $\{h_{ij,k}, 1 \leq i \leq m, 1 \leq j \leq f\}$ is absolutely continuous, i.e., has a probability density function on $\mathcal{H}^{m \times f}$.

If system *i* transmits at time *k* over frequency *j* it selects a transmit power level $p_{ij,k}$ taking values in $[0, p_{max}]$. Following the wireless channel model introduced in Section 2.2, the probability of successfully decoding the message at the receiver depends on the received power level expressed by the product $h \cdot p$ of channel fading and the allocated transmit power. The probability of success is given by a known relationship of the form $q(h_{ij,k} \cdot p_{ij,k})$ – see also Fig. 2 where this relationship is depicted. The following technical assumption on the form of the function q(hp) will be helpful in the subsequent sections.

Assumption 4.2. The function q(.) as a function of the product r = h p for $r \ge 0$ satisfies:

(a) q(0) = 0,

(b) q(r) is continuous, and strictly increasing when q(r) > 0, i.e., for r' > r it holds that q(r') > q(r) > 0,

(c) for any $\mu \in \mathbb{R}_+$ and for almost all values $h \in \mathcal{H}$ the set $\operatorname{argmin}_{0 is a singleton.$

Parts (a) and (b) of this assumption state that the probability of successful decoding q(hp) is zero when the received power level hp is small, and it becomes positive q(hp) > 0 and strictly increasing for larger values of hp. These properties are verified for cases of practical interest as shown in Fig. 2. Part (c) is more stringent but not restrictive in practice. The function q(hp) typically has a sigmoid form with exponential tails as shown in Fig. 2. This verifies that the power

minimizer in (c) is unique for almost all channel gains h, and is either equal to zero or belongs to the strictly concave exponential tail. We also note that the minimizer set in (c) exists by the continuity assumption in (b). These properties are assumed for technical reasons that will become clear later (see the discussion before Theorem 4.1).

We are interested in a centralized mechanism that selects which system accesses each of the available frequencies at the channel, i.e., which system is *scheduled* to transmit. We denote with $\alpha_{ij,k} = 1$ the decision to schedule system *i* on frequency *j* at time *k*, and $\alpha_{ij,k} = 0$ otherwise. To avoid packet collisions we let at most one system transmit on each frequency *j*, that is $\sum_{i=1}^{m} \alpha_{ij,k} \leq 1$. We allow each system *i* to transmit on at most one frequency, that is $\sum_{j=1}^{f} \alpha_{ij,k} \leq 1$. Mathematically we denote the set $\Delta_{m,f}$ of all feasible scheduling decisions $\alpha_{ij,k}$ at each time *k* as

$$\Delta_{m,f} = \left\{ \alpha \in \{0,1\}^{m \times f} : \begin{array}{l} \sum_{i=1}^{m} \alpha_{ij} \leq 1, \ 1 \leq j \leq f, \\ \sum_{j=1}^{f} \alpha_{ij} \leq 1, \ 1 \leq i \leq m \end{array} \right\}.$$
(4.2)

For compactness we group channel states, scheduling decisions, and power allocations of the communication model at time *k* into matrices $h_k \in \mathcal{H}^{m \times f}$, $\alpha_k \in \Delta_{m,f}$, and $p_k \in [0, p_{\max}]^{m \times f}$ respectively. We can then model the transmission event $\gamma_{i,k}$ of system *i* at time *k* given scheduling variables, power allocation, and channel state, as a Bernoulli random variable with success probability

$$\mathbb{P}[\gamma_{i,k} = 1 \mid h_k, \alpha_k, p_k] = \sum_{j=1}^f \alpha_{ij,k} \, q(h_{ij,k}, p_{ij,k})$$
(4.3)

This expression states that the probability of a message for system i being successfully received equals the probability that the message is correctly decoded if system i is scheduled to transmit on any of the f available frequencies. Note that, by design of the scheduling variables, system i uses at most one frequency, and we make the implicit assumption that no interferences arise from transmissions on different frequencies.

Our goal is to design the communication variables of the shared wireless control system, i.e., the scheduling and power allocation. Since the randomly varying channel affects the communication process, we are interested in selecting scheduling and power variables that adapt to channel states h_k in order to counteract or exploit these effects. Overall we express the scheduling and power decisions at time k by mappings $\alpha_k = \alpha(h_k)$ and $p_k = p(h_k)$ selected from the sets

$$\mathcal{A} = \{ \alpha : \mathcal{H}^{m \times f} \to \Delta_{m,f} \},$$

$$\mathcal{P} = \{ p : \mathcal{H}^{m \times f} \to [0, p_{\max}]^{m \times f} \}.$$
(4.4)

Since channel states h_k are i.i.d. over time k these mappings do not need to change over time. Substituting the scheduling and power allocation mappings $\alpha(.)$, p(.) in our communication model described by (4.3) the probability of successful transmission for each system i at any given slot k becomes

$$\mathbb{P}(\gamma_{i,k} = 1) = \mathbb{E}_{h_k} \left\{ \mathbb{P}[\gamma_{i,k} = 1 \mid h_k, \alpha(h_k), p(h_k)] \right\}$$
$$= \mathbb{E}_h \sum_{j=1}^f \alpha_{ij}(h) q(h_{ij}, p_{ij}(h)).$$
(4.5)

Here the expectation is with respect to the joint distribution ϕ of the channel realization h_k which we assumed to be identical for any time k, hence we drop the index k. Note also that the communication process modeled by the sequence $\{\gamma_{i,k}, 1 \le i \le m, k \ge 0\}$ depends only on variables related to the wireless communication counterpart of the overall system, and is in particular independent of the system evolutions $\{x_{i,k}, 1 \le i \le m, k \ge 0\}$.

Our primary goal in designing the communication variables of the system is to guarantee a level of closed loop control performance for each subsystem. To formalize the problem description we consider Lyapunov-like performance requirements for each control system, as introduced in Section 2.1.1. In particular suppose that for each system i a quadratic Lyapunov function of the form

$$V_i(x_i) = x_i^T P_i x_i, \ x_i \in \mathbb{R}^{n_i}, \tag{4.6}$$

with positive definite matrix $P_i \in S_{++}^{n_i}$ is given. A Lyapunov-like requirement then states that *for all systems i* at time *k* the Lyapunov functions at the next time step decrease at the desired rates $\rho_i < 1$ *in expectation,* that is

$$\mathbb{E}\left[V_i(x_{i,k+1}) \mid x_{i,k}\right] \le \rho_i \, V_i(x_{i,k}) + Tr(P_i W_i) \tag{4.7}$$

for any possible value of the current plant states $x_{i,k} \in \mathbb{R}^{n_i}$. The expectation over the next system state $x_{i,k+1}$ on the left hand side accounts via (4.1) for the randomness introduced by the process noise $w_{i,k}$ as well as the transmission success $\gamma_{i,k}$. The latter is expressed in (4.5) and depends on the observed channel state h_k as well as the communication decisions α_k , p_k .

On the other hand, apart from control performance requirements an efficient communication design should make an efficient use of the available power resources at the devices. The induced overall expected power consumption on each slot k is given by

$$\mathbb{E}_{h_k} \sum_{i=1}^m \sum_{j=1}^f \alpha_{ij,k}(h_k) p_{ij,k}(h_k),$$
(4.8)

summing up the transmit power of each system *i* and frequency *j* if the system is scheduled to transmit. The expectation here is with respect to the joint distribution ϕ of channels h_k . We design scheduling and power allocation (cf. (4.4)) that are control-performance aware (cf. (4.11)) and also energy-efficient (cf. (4.8)) through a stochastic optimization framework that we present next.

We formulate the problem of designing scheduling and power allocation in an optimization frame-

work as follows.

Problem 4.1 (Optimal Scheduling and Power Allocation Design). Consider a shared wireless control architecture with f frequencies and m systems of the form (4.1), quadratic Lyapunov performance require*ments by* (4.7), *channel states* $h_k \in \mathcal{H}^{m \times f}$ *i.i.d. with distribution* ϕ *, and communication modeled by* (4.3). *The design of optimal scheduling and power allocation as functions of the current channel states* $\alpha_k = \alpha(h_k)$ and $p_k = p(h_k)$ respectively is posed as

$$\min_{\substack{\alpha, p \in (\mathcal{A}, \mathcal{P})}} \mathbb{E}_{h_k} \sum_{i=1}^m \sum_{j=1}^f \alpha_{ij,k}(h_k) p_{ij,k}(h_k)$$
(4.9)
$$\operatorname{subject to} \mathbb{E}\left[V(x_{i-1}) \mid x_{i-1} \right] \leq o V(x_{i-1}) + \operatorname{Tr}(\mathcal{B}W)$$

subject to

$$\mathbb{E}\left[V_i(x_{i,k+1}) \mid x_{i,k}\right] \le \rho_i V_i(x_{i,k}) + \operatorname{Tr}(P_i W_i)$$

for all $x_{i,k} \in \mathbb{R}^{n_i}, \ i = 1, \dots, m.$ (4.10)

In other words, at each time step we seek to minimize the total expected power consumption (4.8) of the design while satisfying the Lyapunov requirements (4.7) for all systems *i* and for any value of the current plant states $x_{i,k} \in \mathbb{R}^{n_i}$, since scheduling and power allocation adapt to channel states but are independent of the plant states.

The Lyapunov control performance abstraction can be equivalently translates to desired packet success rates for each link. This important observation will permit the design of control-aware scheduling policies. It is formally stated in the following proposition.

Proposition 4.1 (Lyapunov Control Performance Abstraction). Consider a switched linear system described by (4.1) where $\gamma_{i,k}$, $k \ge 0$ is a sequence of random binary variables, and a quadratic function $V_i(x_i) = x_i^T P_i x_i, x_i \in \mathbb{R}^{n_i}$ with a positive definite matrix $P_i \succ 0$. Further suppose that the function $V_i(x_i)$ *is a Lyapunov function for the closed loop mode of the system, i.e.,* $A_{c,i}^T P_i A_{c,i} \prec \rho_i P_i$. Then the function $V_i(x_i)$ decreases with an expected rate $\rho_i < 1$ at each step, i.e., we have

$$\mathbb{E}\left[V_i(x_{i,k+1}) \mid x_{i,k}\right] \le \rho_i \, V_i(x_{i,k}) + Tr(P_i W_i) \quad \text{for all } x_{i,k} \in \mathbb{R}^{n_i}, \, k \ge 0, \tag{4.11}$$

if and only if the packet success rate satisfies

$$\mathbb{P}(\gamma_{i,k}=1) \ge c_i, \text{ for all } k \ge 0, \tag{4.12}$$

where c_i is computed by the semidefinite program

$$c_{i} = \min\{\theta \ge 0: \ \theta \ A_{c,i}^{T} P_{i} A_{c,i} + (1-\theta) \ A_{o,i}^{T} P A_{o,i} \preceq \rho_{i} P_{i}\}.$$
(4.13)

This proposition is important as it transforms control performance requirements (4.11) to wireless communication requirements, i.e., desired packet success rates (4.12) for each link. In particular there exist easily computed non-negative constants $c_i \ge 0$ such that (4.7) is equivalent to requiring $\mathbb{P}(\gamma_{i,k} = 1) \ge c_i$. This makes explicit how the resource allocation functions $\alpha(.)$, p(.) appear in the optimization problem via (4.5).

As a result the optimization problem (4.9) can be equivalently written as

$$\begin{array}{ll} \underset{\alpha,p\in(\mathcal{A},\mathcal{P})}{\text{minimize}} & \mathbb{E}_{h}\sum_{i=1}^{m}\sum_{j=1}^{f}\alpha_{ij}(h)p_{ij}(h) & (4.14) \\ \\ \text{subject to} & c_{i} \leq \mathbb{E}_{h}\sum_{j=1}^{f}\alpha_{ij}(h)q(h_{ij},p_{ij}(h)), \ i = 1,\ldots,m \end{array}$$

Here we have dropped the time indices k from the variables h_k since they are identically distributed over time. Finally we make a constraint qualification assumption that is typical in optimization theory, i.e., that a strictly feasible solution exists.

Assumption 4.3. There exist variables $\alpha' \in A$ and $p' \in P$ that satisfy the constraints of the optimization problem (4.14) with strict inequality, i.e.,

$$c_i < \mathbb{E}_h \sum_{j=1}^f \alpha'_{ij}(h) q(h_{ij}, p'_{ij}(h)), \quad i = 1, \dots, m$$
 (4.15)

By the equivalence between problems (4.9) and (4.14), condition (4.15) can be interpreted as a feasibility/schedulability assumption for the shared wireless control system. It requires that there exist some channel-aware scheduling and power allocation such that the control performance requirements (4.7) of *all* control systems are met. This assumption however does not provide any information on how to find such a solution.

In the rest of the chapter we examine problem (4.14), which is equivalent to the optimal scheduling and power allocation for the shared wireless control architecture (Problem 4.1). Since this problem is feasible (Assumption 4.3) we denote the optimal value by *P* and an optimal solution pair by $\alpha^*(.)$, $p^*(.)$. In the following section we characterize the form of the optimal solution and describe a methodology to obtain it.

4.2. Optimal scheduling and power allocation

In this section we examine how the optimal scheduling and power allocation for the wireless control system can be recovered by considering the optimization problem in the dual domain. This allows us to develop an offline algorithm to solve the problem and provides an explicit characterization of the form of the optimal solution.

First let us derive the Lagrange dual problem of (4.14). Consider non-negative dual variables $\mu \in \mathbb{R}^m_+$ corresponding to each one of the *m* constraints of (4.14). The Lagrangian then is defined

as

$$L(\alpha, p, \mu) = \mathbb{E}_{h} \sum_{i=1}^{m} \sum_{j=1}^{f} \alpha_{ij}(h) p_{ij}(h) + \sum_{i=1}^{m} \mu_{i} \left[c_{i} - \mathbb{E}_{h} \sum_{j=1}^{f} \alpha_{ij}(h) q(h_{ij}, p_{ij}(h)) \right],$$
(4.16)

while the dual function is defined as

$$g(\mu) = \inf_{\alpha, p \in (\mathcal{A}, \mathcal{P})} L(\alpha, p, \mu).$$
(4.17)

For future reference we also denote the set of functions $\alpha(.)$, p(.) that minimize the Lagrangian at μ by

$$(\mathcal{A}, \mathcal{P})(\mu) = \underset{\alpha, p \in (\mathcal{A}, \mathcal{P})}{\operatorname{argmin}} L(\alpha, p, \mu),$$
(4.18)

whenever the minimizers exist. This set might contain in general multiple solutions and we denote with $\alpha(\mu)$, $p(\mu)$ an arbitrary element pair of the set. Since the pair itself is a function on $\mathcal{H}^{m \times f}$ (cf. (4.4)), we denote the value of the pair at a point $h \in \mathcal{H}^{m \times f}$ by $\alpha(\mu; h)$, $p(\mu; h)$.

The Lagrange dual problem is defined as follows.

$$D = \sup_{\mu \in \mathbb{R}^m_+} g(\mu).$$
(4.19)

According to Lagrange duality theory the optimal dual value *D* is a lower bound on the optimal cost *P* of problem (4.14). The following proposition however establishes a strong duality result (D = P) for the problem under consideration and provides a relationship between the optimal primal and dual variables.

Proposition 4.2. Let Assumptions 4.1 and 4.3 hold. Let P be the optimal value of the optimization problem (4.14) and (α^*, p^*) be an optimal solution, and let D be the optimal value of the dual problem (4.19) and μ^* be an optimal solution. Then

(a) P = D (strong duality)

(b)
$$\mu_{i}^{*} \left[c_{i} - \mathbb{E}_{h} \sum_{j=1}^{f} \alpha_{ij}^{*}(h) q(h_{ij}, p_{ij}^{*}(h)) \right] = 0$$
 for all $i = 1, ..., m$ (complementary slackness)
(c) $(\alpha^{*}, p^{*}) \in (\mathcal{A}, \mathcal{P})(\mu^{*})$

This proposition states that strong duality holds even though the original problem is not convex, regardless also of the form of the function q(h, p) (Assumption 4.2 is not imposed). More importantly, part (c) suggests the possibility of finding the optimal primal variables α^* , p^* by solving first for the optimal point μ^* of the dual problem, and then searching for primal variables the minimize the Lagrangian function at μ^* (cf. (4.18)). As we present next, this direction provides a

significant advantage. The design of infinite-dimensional scheduling and power allocation policies that meet the control performance specifications in Problem 4.1 is reduced to the problem of determining finite-dimensional optimal dual variables. A technical caveat of Proposition 4.2(c) is that the optimal policies are included in a set which could in general contain other irrelevant policies. As we show next, Assumption 4.2 helps overcome this issue.

4.2.1. Dual subgradient method

To solve the dual problem in (4.19), that is, to maximize the dual function $g(\mu)$, we employ a dual projected subgradient algorithm (Bertsekas et al., 2003, Ch. 8). We first note that function $g(\mu)$ is concave, as a pointwise infimum over functions linear in μ (cf. (4.17)). A subgradient direction for $g(\mu)$ at any point $\mu \in \mathbb{R}^m_+$ is a vector, denoted here as $s(\mu) \in \mathbb{R}^m$, that satisfies

$$g(\mu') - g(\mu) \le (\mu' - \mu)^T s(\mu) \quad \text{for all} \mu' \in \mathbb{R}^m_+.$$
(4.20)

If we pick $\alpha(\mu)$, $p(\mu) \in (\mathcal{A}, \mathcal{P})(\mu)$ by (4.18) then a subgradient $s(\mu)$ can be found as the constraint slack of the primal problem (4.14) evaluated at these points, i.e.,

$$s_i(\mu) = c_i - \mathbb{E}_h \sum_{j=1}^f \alpha_{ij}(\mu; h) \, q(h_{ij}, p_{ij}(\mu; h)).$$
(4.21)

To show this observe that for any $\mu' \in \mathbb{R}^m_+$ in general we have $g(\mu') \leq L(\alpha(\mu), p(\mu), \mu')$ by the definition of the dual function in (4.17). Subtracting $g(\mu) = L(\alpha(\mu), p(\mu), \mu)$ from both sides of this inequality and expanding the terms of the Lagrangian as in (4.16) we get

$$g(\mu') - g(\mu) \le \sum_{i=1}^{m} (\mu'_i - \mu_i) \left[c_i - \mathbb{E}_h \sum_{j=1}^{f} \alpha_{ij}(\mu; h) \, q(h_{ij}, p_{ij}(\mu; h)) \right].$$
(4.22)

Comparing this with the property of the subgradient in (4.20), we verify that (4.21) indeed gives a subgradient direction. We also note for future reference that for any μ the subgradients are bounded because at the right hand side of (4.21) the term c_i is bounded and the term in the expectation corresponds to a probability (cf.(4.5)).

A projected dual subgradient ascent method to maximize the concave dual function $g(\mu)$ consists of the following steps:

1. At iteration t given $\mu(t)$ find primal optimizers of the Lagrangian at $\mu(t)$ according to (4.18),

$$p(\mu(t)), \alpha(\mu(t)) \in (\mathcal{A}, \mathcal{P})(\mu(t))$$
(4.23)

2. Evaluate the subgradient vector $s(\mu(t))$ by (4.21) and update the dual variables by an ascent

step

$$\mu(t+1) = [\mu(t) + \varepsilon(t)s(\mu(t))]_+$$
(4.24)

where $[]_+$ denotes the projection on the non-negative orthant and $\varepsilon(t) > 0$ is the stepsize.

The stepsizes are selected to be square summable but not summable, i.e.,

$$\sum_{t\geq 1} \varepsilon(t)^2 < \infty, \ \sum_{t\geq 1} \varepsilon(t) = \infty.$$
(4.25)

Before stating the convergence properties of the algorithm, we note that in order to implement it we need an efficient way to compute primal Lagrange optimizers in (4.23) that solve (4.18). This problem also relates to our capability of finding the optimal primal variables of interest α^* , p^* as we have argued by Proposition 4.2(c). Hence we turn our focus to problem (4.18). A more convenient expression for the Lagrangian defined in (4.16) can be obtained by rearranging terms to get

$$L(\alpha, p, \mu) = \mu^{T} c + \mathbb{E}_{h} \sum_{i=1}^{m} \sum_{j=1}^{f} \alpha_{ij}(h) \left[p_{ij}(h) - \mu_{i}q(h_{ij}, p_{ij}(h)) \right].$$
(4.26)

This form provides a useful separation structure for the primal Lagrangian optimizers that we exploit in the following proposition.

Proposition 4.3. For any $\mu \in \mathbb{R}^m_+$ the following hold true:

(a) Solutions $\alpha(\mu)$, $p(\mu) \in (\mathcal{A}, \mathcal{P})(\mu)$ of problem (4.18) can be obtained at each $h \in \mathcal{H}^{m \times f}$ as

$$p_{ij}(\mu;h) = p_{ij}(\mu_i;h_{ij}) = \underset{0 \le p \le p_{\max}}{\operatorname{argmin}} p - \mu_i q(h_{ij},p)$$
(4.27)

for any i = 1, ..., m and j = 1, ..., f, and

$$\alpha(\mu;h) = \underset{\alpha \in \mathbb{R}^{m \times f}_{+}}{\operatorname{argmin}} \qquad \sum_{i=1}^{m} \sum_{j=1}^{f} \alpha_{ij} \,\xi(h_{ij},\mu_i) \qquad (4.28)$$

subject to
$$\sum_{i=1}^{m} \alpha_{ij} \leq 1, \ \sum_{j=1}^{f} \alpha_{ij} \leq 1$$

where

$$\xi(h_{ij},\mu_i) = \min_{0 \le p \le p_{\max}} p - \mu_i q(h_{ij},p).$$
(4.29)

(b) If Assumptions 4.1 and 4.2 hold, then for any solution $\alpha(\mu)$, $p(\mu) \in (\mathcal{A}, \mathcal{P})(\mu)$ the vector $s(\mu)$ defined in (4.21) has a unique value.

The first part of the proposition provides in (4.27) and (4.28) a method to obtain primal Lagrange optimizers that can be used in step (4.23) of the subgradient algorithm. Interestingly, the minimizing scheduling and power allocation decisions can be computed separately at each channel state value $h \in \mathcal{H}^{m \times f}$, hence significantly simplifying the computation. A further separability for the power allocation across systems and frequencies is revealed – see Remark 4.1.

The second part of the proposition relies on the properties of the function q(h, p) by Assumption 4.2 in order to establish that the subgradient vector $s(\mu)$ takes a unique value. The proof relies on the fact that the expected value in $s_i(\mu)$ (cf. (4.21)) conditioned on the channel state h is almost surely either zero, because system i should not be scheduled, or unique, because the optimal Lagrange minimizers $\alpha(\mu;h)$, $p(\mu;h)$ are unique. For example, Assumption 4.2(c) ensures that the power minimizer $p(\mu;h)$ in (4.27) is (almost surely) unique. Proposition 4.3(b) is important because it allows a stronger and more explicit characterization of the optimal scheduling and power allocation than the set-characterization of Proposition 4.2(c). This is established in the following theorem.

Theorem 4.1 (Optimal Scheduling and Power Allocation). Consider the design of channel-aware scheduling and power allocation variables in Problem 4.1 for the shared wireless control architecture of Fig. 10, and let Assumptions 4.1, 4.2, 4.3 hold. Then optimal scheduling α^* and power allocation p^* are obtained by (4.27)-(4.29) at any point $\mu^* \in \mathbb{R}^m_+$ that is an optimal solution of the dual problem (4.19). Moreover a point μ^* can be obtained by iterating (4.23)-(4.24), i.e., $\mu(t) \rightarrow \mu^*$, for stepsizes satisfying (4.25).

The theorem characterizes the optimal scheduling and power allocation that meet the control performance specifications in our shared wireless control architecture – see Remarks 4.1, 4.2 for more details about the form of the optimal policy. It is worth noting that the optimal policy need not be unique. More precisely, there might be many optimal dual solutions μ^* , each corresponding to a different scheduling and power allocation policy according to the theorem. However all such policies will have the same objective value in (4.9).

The theorem also establishes a methodology to find the optimal communication policy by iterating (4.23)-(4.24). This can be viewed as an offline algorithm, and requires knowledge of the channel distribution. In the next section we develop an online algorithm that solves for the optimal communication policy based instead only on a random sequence of channel realizations observed during system execution.

Remark 4.1. According to Theorem 4.1, the optimal power allocation can be obtained at each channel value h by solving (4.27) at the point μ^* . In particular $p_{ij}^*(h)$ depends on the variables μ_i^* and h_{ij} pertinent only to system i and frequency j and not on the whole vectors μ^* or h. This implies a decentralized power allocation among systems and frequencies, made explicit in (4.27) by the notation $p_{ij}(\mu_i; h_{ij})$. Similar separability results are also known in the context of resource allocation for wireless communication networks Liu et al. (2003); Georgiadis et al. (2006); Ribeiro (2012) even though in those works the goal is to maximize user utility (capacity, data rates, etc.) while the goal here is to meet closed loop control performance. The

separability can be intuitively understood from the shared wireless control architecture of Fig. 10, since each transmitter experiences different channel conditions and is responsible for an independent control task. Moreover, this optimal power allocation can be easily implemented in practice. The transmitter of each control system i can store its value μ_i^* and adapt transmit power, whenever scheduled, based on the channel conditions it currently experiences. The optimal scheduling $\alpha^*(h)$ in (4.28), on the other hand, is centralized since it depends on the whole vector μ^* and all channel states h.

Remark 4.2. The optimal scheduling decision in (4.28) is posed as a linear program by relaxing the integer constraints of $\Delta_{m,f}$ in (4.2), hence the policy is computationally efficient. As mentioned in the proof of the theorem there is no loss in the relaxation, as the optimal solution to the linear program is integer. It is worth noting that (4.28) solves a standard assignment problem² which opportunistically tries to match control systems to frequencies at each time slot. A numerical example of this opportunistic behavior is shown in Fig. 11 of Section 4.4. Besides the linear program presented here, integer programming algorithms with complexity polynomial in the number of systems m and frequencies f exist (Bertsimas and Tsitsiklis, 1997, Ch. 7). In the special case of a single frequency (f = 1) the complexity of scheduling in (4.28) is linear in the number of systems (O(m)), since the scheduler looks for and schedules the system i with the minimum value $\xi(h_i, \mu_i)$.

4.3. Online scheduling and power allocation

The algorithm presented in the previous section to obtain optimal scheduling and power allocation for the shared wireless control system of Problem 4.1 is hard to implement in practice. In the primal step (4.23) one needs to obtain a solution pair $\alpha(h)$, p(h) for a continuum of channel variables $h \in \mathcal{H}^{m \times f}$, while for the dual step in (4.24) one needs to compute the subgradient direction $s(\mu)$ in (4.21) by integrating over the channel distribution ϕ . A practical implementation would require drawing a large number of samples from ϕ and solving for primal variables at these samples to obtain an estimate of the actual subgradient direction. This is computationally intensive, does not scale for a large number of systems *m* and frequencies *f*, while also in most cases of practical interest the channel distribution is not available.

These drawbacks motivate us to develop an *online* algorithm to solve Problem 4.1. The algorithm is a stochastic version of the primal/dual steps (4.23), (4.24) of the offline subgradient method and does not rely on availability of the channel distribution. In particular, suppose that at time k a channel realization h_k is observed, and the current power and scheduling decision are selected by solving (4.27)-(4.28) at the current h_k , i.e.,

$$p_{ij,k} = p_{ij}(\mu_{i,k}; h_{ij,k}), \quad i = 1, \dots, m, \ j = 1, \dots, f,$$

$$\alpha_k = \alpha(\mu_k; h_k). \tag{4.30}$$

Then in contrast to updating the dual variables μ_k by (4.24) after computing the vector (4.21), sup-

²Technically the standard assignment problem requires equal number of systems and frequencies. This can be accomplished by introducing dummy systems or frequencies with zero values $\xi(h_{ij}, \mu_i)$.

Algorithm 4.1 Online Scheduling and Power Allocation

Input: *m*, *f*, *c* \in $[0, 1]^m$, *q* : $\mathcal{H} \times [0, p_{\max}] \mapsto [0, 1]$, $\varepsilon_k \in \mathbb{R}_+, k \ge 0$ 1: Initialize $\mu_0 \in \mathbb{R}^m_+, k \leftarrow 0$

2: **loop**

- 3: At time *k* observe channel state h_k
- 4: Compute power allocation for all systems *i* and frequencies *j* by

$$p_{ij,k} \leftarrow \underset{0 \le p \le p_{\max}}{\operatorname{argmin}} p - \mu_{i,k} q(h_{ij}, p)$$
(4.33)

$$\xi_{ij,k} \leftarrow \min_{0 \le p \le p_{\max}} p - \mu_{i,k} q(h_{ij}, p)$$
(4.34)

5: Decide scheduling by solving

$$\alpha_k \leftarrow \underset{\alpha \in \Delta_{m,f}}{\operatorname{argmin}} \sum_{i=1}^m \sum_{j=1}^f \alpha_{ij} \,\xi_{ij,k} \tag{4.35}$$

6: Compute for all i = 1, ..., m

$$s_{i,k} \leftarrow c_i - \sum_{j=1}^f \alpha_{ij,k} \, q(h_{ij,k}, p_{ij,k})$$
 (4.36)

7: Update dual variables by $\mu_{k+1} \leftarrow [\mu_k + \varepsilon_k s_k]_+$ 8: end loop

pose only the current channel measurement and power/scheduling choices are used. In particular, suppose we compute

$$s_{i,k} = c_i - \sum_{j=1}^f \alpha_{ij,k} q(h_{ij,k}, p_{ij,k}), \quad i = 1, \dots, m,$$
 (4.31)

and update the variables μ_k by

$$\mu_{k+1} = [\mu_k + \varepsilon_k s_k]_+ \tag{4.32}$$

where $[]_+$ is the projection on the non-negative orthant and $\varepsilon_k > 0$ is the stepsize.

To emphasize that this is an online algorithm we have explicitly indexed the variables with k corresponding to real time slots. This procedure, summarized in Algorithm 4.1, gives scheduling and power variables $\{\alpha_k, p_k, k \ge 0\}$ as well as dual variables $\{\mu_k, k \ge 0\}$ which are random because they depend on the random observed channel sequence $\{h_k, k \ge 0\}$. The main difference compared to the subgradient algorithm of the previous section is that it follows random directions s_k in (4.31) instead of the exact subgradient directions $s(\mu_k)$ by (4.21). Comparing these two expressions it is immediate that the expected value of s_k coincides with the subgradient $s(\mu_k)$, so it is reasonable to conjecture that the online algorithm is expected to move towards the maximum of the dual function, as the subgradient method does. The following proposition indeed establishes convergence in a strong sense.

Proposition 4.4. Consider the optimization problem (4.14) and its dual derived in (4.19) and let Assumption 4.3 hold. Let a sequence $\mu_k, k \ge 0$ be obtained by steps (4.30)-(4.32) based on a sequence $\{h_k, k \ge 0\}$ of *i.i.d.* random variables with distribution ϕ , and stepsizes ε_k satisfying (4.25). Then almost surely we have that

$$\lim_{k \to \infty} \mu_k = \mu^*, \text{ and } \lim_{k \to \infty} g(\mu_k) = D$$
(4.37)

where μ^* is an optimal solution of the dual problem and D is the optimal value of the dual problem.

The proposition states that the stochastic online algorithm yields a random sequence of dual variables μ_k that converges to the optimal point μ^* almost surely for any sequence of channel realizations that is observed. However the real problem of interest is the primal problem (4.14), or equivalently Problem 4.1. This is the problem of optimal design of scheduling and power allocation policies that satisfy the given Lyapunov performance requirements (4.11) for each control system *i*, while also minimizing the expected power expenditures of the communication process. Hence, in the following theorem, we characterize how the control systems would actually perform if the communication variables are selected according to the proposed online algorithm.

Theorem 4.2 (Online Scheduling and Power Allocation). Consider a shared wireless control architecture composed of m systems of the form (4.1), f frequencies, and communication modeled by (4.3) depending on channel states $h_k \in \mathcal{H}^{m \times f}$ which are i.i.d. with distribution ϕ , and scheduling and power allocation variables $\alpha_k \in \Delta_{m,f}$, $p_k \in [0, p_{\max}]^{m \times f}$. Also consider given quadratic Lyapunov performance requirements (4.11) for each system and let Assumptions 4.1, 4.2, 4.3 hold. If α_k , p_k are chosen according to (4.30)-(4.32), then almost surely with respect to the channel sequence $\{h_k, k \ge 0\}$ the control performances for all systems $i = 1, \ldots, m$ satisfy

$$\limsup_{k \to \infty} \mathbb{E}[V_i(x_{i,k+1}) \mid x_{i,k} = x_i, h_0, \dots, h_{k-1}]$$

$$\leq \rho_i V_i(x_i) + Tr(P_i W_i), \qquad (4.38)$$

for any state values $x_i \in \mathbb{R}^{n_i}$. In addition, the power consumption almost surely satisfies

$$\limsup_{k \to \infty} \mathbb{E}\left[\sum_{i=1}^{m} \sum_{j=1}^{f} \alpha_{ij,k} p_{ij,k} \middle| h_0, \dots, h_{k-1}\right] \le P$$
(4.39)

where *P* is the optimal value of the optimization problem (4.9).

According to the theorem the scheduling and power allocation variables selected by the online algorithm lead in the limit to the desired Lyapunov requirements for all control systems and to the optimal power expenditure, for almost all channel sequences. We can also establish the following corollary.

Corollary 4.1. Consider the setup of Theorem 4.2. Then for any positive constant $\delta > 0$ there exists a time

step N such that for all times $k \ge N$ we have that

$$\mathbb{E}\left[V_i(x_{i,k+1}) \mid x_{i,k}\right] \le (\rho_i + \delta) \ V_i(x_{i,k}) + Tr(P_i W_i) \tag{4.40}$$

for any possible value of plant states $x_{i,k} \in \mathbb{R}^{n_i}$ and for all systems i = 1, ..., m.

Recall that we initially asked for a communication design that guarantees expected control performance requirements at each time step k in (4.11). According to the above corollary our online algorithm approximately satisfy this. After a sufficiently long time horizon the expected decrease rates of all Lyapunov functions get arbitrarily close to the desired ones. Before proceeding to simulations of the stochastic online algorithm, we present an intuitive interpretation of the algorithm from an economic resource allocation point of view.

4.3.1. Pricing interpretation of online scheduling and power allocation algorithm

In this section we provide an interpretation of the problem variables as well at the online Algorithm 4.1 in economic terms. In particular we may view each transmitter in the wireless control architecture as an agent that utilizes some scarce resource, namely transmit power, to produce some 'good', namely the probability of successfully transmitting and closing the corresponding control loop. Our development shows that each closed loop has a Lyapunov control performance requirement (cf. (4.11)) that can be translated as requiring c_i units of good (cf. (4.14)). Under this view, the dual variables μ_i can be interpreted as the 'unit price' at which each agent can 'sell' the produced good. In this context the role of Algorithm 4.1 is to determine unit prices such that all demand levels c_i are met and in the most profitable manner from the agents' perspective.

More specifically, consider a time step *k* where prices are set to μ_k and the current channel conditions are described by h_k . If agent *i* gets access to the channel at frequency *j*, the agent can spend an amount $p_{ij,k}$ to produce $q(h_{ij,k}, p_{ij,k})$ units of good, which can be sold at a price of $\mu_{i,k}$ per unit. In this case the total profit for the agent can be expressed as

$$\mu_{i,k} q(h_{ij,k}, p_{ij,k}) - p_{ij,k}, \tag{4.41}$$

i.e., the difference between the total revenue $\mu_{i,k} q(h_{ij,k}, p_{ij,k})$ and the total cost $p_{ij,k}$. The optimal resource allocation $p_{ij,k}$ is the one maximizing the profit (4.41), matching exactly the optimization over power provided in (4.33). The optimal profit if agent *i* gets access to the channel at frequency *j* under conditions $h_{ij,k}$ equals $-\xi_{ij,k}$ given in (4.34).

Then the role of the scheduler is to opportunistically assign agents to the available frequencies in a way that maximizes the total aggregated profit. In particular the scheduler observes current conditions $h_{ij,k}$ for all agents *i* and frequencies *j*, computes the possible profit $-\xi ij, k$ of all agent/frequency pairs, and searches for the scheduling $\alpha \in \Delta_{m,f}$ defined by (4.2) that maximizes the total profit

$$\sum_{i,j} \alpha_{ij}(-\xi ij,k) \tag{4.42}$$



Figure 11: Optimal channel-aware scheduling for the example presented in Section 4.4. System 1 has a harder Lyapunov decrease rate requirement and is scheduled to transmit for most observed channel states h_1 , h_2 . System 2 is scheduled only if its channel conditions h_2 are much more favorable that those of system 1. When both channels are very adverse, systems select zero transmit powers so scheduling is irrelevant.

aggregated over all agents. This optimal scheduling matches the one implemented by Algorithm 4.1 (cf. line (4.35)).

After the current scheduling α_k and power p_k decisions have been made, the unit prices μ_{k+1} for the next step are adjusted depending on the current production levels. If the production for system *i* exceeds the required level c_i , i.e., $s_{i,k} < 0$ in (4.31), then the unit price for system *i* is reduced to $\mu_{i,k} + \varepsilon_k s_{i,k}$ (cf. line 7 in Algorithm 4.1). If on the other hand the production for system *i* does not meet c_i , i.e., $s_{i,k} > 0$, then the unit price *i* increases to $\mu_{i,k} + \varepsilon_k s_{i,k}$.

According to Theorem 4.2 the online algorithm converges almost surely to the optimal prices μ^* , under which the expected production meets demand. Moreover the expected total production cost (the objective of problem (4.14)) becomes optimal in the limit.

We note that Theorem 4.2 does not provide theoretical guarantees on how fast the solution converges to the optimal one. We discuss this issue along with other limitations of the algorithm in Section VI. In the following section we present simulations verifying our theoretical results, and also indicating that the convergence of the algorithm is relatively fast so that online control performance is not severely affected.

4.4. Numerical Simulations

We first illustrate through simulations the opportunistic nature of the communication mechanism for wireless control systems obtained in Section 4.2, in particular how scheduling and power decisions adapt appropriately to channel conditions to meet the control performance goals. Moreover we compare the resulting performance with other simple non channel-adaptive mechanisms. Recall that control systems with vector states are converted to scalar constraints in optimization problem (4.14) by introducing the constants c_i . Hence without loss of generality we consider scalar control systems.

Consider a heating system application controlling the temperature in two independent rooms of a building. Assuming the wireless control architecture of Fig. 10 with m = 2, wireless sensors transmit the temperatures of each room to a central location (the access point in Fig. 10) responsible for adjusting the heating in the rooms. For simplicity suppose both systems have identical dynamics of the form (4.1) with state $x_{i,k}$ denoting the difference between current and some desired temperature for room *i*. When system *i* transmits ($\gamma_{i,k} = 1$), heating is activated for system *i* and results in stable dynamics $A_{c,i} = 0.4$. Otherwise if $\gamma_{i,k} = 0$ heating is deactivated and the system is open loop unstable with $A_{o,i} = 1.1$.

For simplicity we assume there is one (f = 1) available frequency and for symmetry let channel states $h_{1,k}$ and $h_{2,k}$ be independent for each system, both having an exponential distribution with mean 1. The function q(h, p) is shown in Fig. 2. For these scalar systems it suffices to consider Lyapunov functions $V_i(x) = x^2$. We require then that system 1 guarantees a high Lyapunov decrease $\rho_1 = 0.75$ rate according to (4.11), while system 2 only requires $\rho_2 = 0.90$. For these choices we get a higher required success of transmission $c_1 \approx 0.44$ for system 1 (solving (4.45) of Prop. 4.1), compared to a lower $c_2 \approx 0.30$ of system 2.

Using the offline subgradient method of Section 4.2 to solve problem (4.14), the optimal channelaware scheduling and power allocation variables are depicted in Fig. 11 and Fig. 12 respectively. We observe in Fig. 11 that System 1, which requires higher transmission success c_1 , is scheduled to transmit for most values of the channel states h_1 , h_2 . System 2, which has a lower requirement, is scheduled only if its channel h_2 is sufficiently favorable *and* system 1 experiences an adverse channel h_1 . This illustrates how the scheduler exploits opportunistically the channel conditions to select which system will transmit to close the loop, in order to meet the Lyapunov constraints in a power efficient manner. Note also that when both systems experience very adverse channels scheduling is irrelevant because, as we will see in Fig. 12, the optimal transmit powers then are zero (no transmission).

The optimal power allocation is decentralized as we noted in Remark 4.1, i.e., the transmit power p_i for system *i* depends only on the channel h_i that system *i* experiences. Thus we plot in Fig. 12 the power allocation for both systems on same axes. For both systems, when the channel conditions are adverse it is not worth to spend transmit power. System 1, which has a more demanding control constraint, requires in general higher transmit power since, as we saw in Fig. 11, it is



Figure 12: Optimal channel-aware power allocation for the example presented in Section 4.4. Under adverse channel conditions systems do not transmit. The channel threshold for transmission for system 1 is lower than that of system 2 because the former has a higher Lyapunov decrease rate requirement. System 1 also requires higher transmit power.

scheduled to transmit even under adverse channel conditions. This is also captured in the expected power consumption of each system computed numerically as $\mathbb{E}_h \alpha_1^*(h) p_1^*(h_1) \approx 11 mW$ and $\mathbb{E}_h \alpha_2^*(h) p_2^*(h_2) \approx 6.5 mW$. Hence the minimum total power budget required to meet the control objectives is 17.5mW.

To demonstrate the power savings obtained by the opportunistic mechanism we compare with a simple non-channel-aware mechanism. In particular suppose that at each step a system is scheduled randomly to access the channel/frequency. With a slight abuse of notation systems 1 and 2 are chosen with probabilities α_1 and $\alpha_2 = 1 - \alpha_1$ respectively. When a system is selected, we suppose it transmits with a constant power level p_c . The control performance requirements (cf. (4.14)) in this case become $\alpha_i \mathbb{E}_{h_i}q(h_i p_c) \ge c_i$ for i = 1, 2 and the total power cost is $(\alpha_1 + \alpha_2) p_c = p_c$. We are interested then in selecting the minimum constant power p_c that would satisfy both control requirements. It can be easily argued that a necessary and sufficient condition for the requirements is $\sum_{i=1,2} \alpha_i \mathbb{E}_{h_i}q(h_i p_c) \ge c_1 + c_2$, which is equivalent to $\mathbb{E}_{h_i}q(h_i p_c) \ge c_1 + c_2$ since $\sum_{i=1,2} \alpha_i = 1$. We compute then numerically the minimum constant p_c that satisfies this equivalent requirement, which is $p_c \approx 73mW$. Note that this transmit power is higher than the optimal opportunistic power policy in Fig. 12. Moreover, in this example, the optimal opportunistic scheduling and power allocation achieves almost 80% decrease in power consumption compared to an optimal not channel-aware randomized schedule scheme.

4.4.1. Stochastic online scheduling and power allocation

Next we implement the stochastic online algorithm of Section 4.3 in a setup with three (m = 3) control loops sharing two (f = 2) frequencies. For example consider again the room heating system of the previous section including three rooms/systems with identical dynamics, $A_{o,i} = 1.1$ and $A_{c,i} = 0.4$ as before. The chosen desired Lyapunov decrease rates are shown in Table 1, implying that system 1 is the most demanding. We assume channel states h_{ij} are independent

	Control	Mean	Mean	Transmit Transmit	
	objec-	Fad-	Fad-	Rate	Rate
	tive	ing	ing	at	at
	ρ	h_{i1}	h_{i2}	Freq.	Freq.
				1	2
Plant 1	0.8	1	1	0.46	0.40
Plant 2	0.92	1	1	0.34	0.29
Plant 3	0.92	1	2	0.16	0.29

Table 1: System parameters & online transmission rates

across systems *i* and frequencies *j*, and have exponential distributions with means given in Table 1. In particular we model that system 3 experiences better channel quality (higher channel fading gain) in the second frequency.

The evolution of the dual variables μ_k during Algorithm 4.1 is shown in Fig. 13. After a number of iterations (time *k* in this example corresponds to seconds) they remain in a small neighborhood around the optimal μ^* , as anticipated by the theoretical a.s. convergence in Prop. 4.4. Consequently, the scheduling and power allocation decisions taken online are almost feasible for the constraints of problem (4.14) after a number of iterations. We observe that the dual variable corresponding to system 1 is the largest, consistent with the fact that it has a harder control requirement to meet. Using the economic interpretation of Section 4.3.1 about the dual variables, the price at which agent 1 can sell its produced good is higher, giving the incentive to schedule agent 1 to produce more often. On the other hand, systems 2 and 3 have the same control requirements but the dual variable for system 2 is larger. The reason is that system 2 experiences worse channel conditions than system 3 (cf. Table 1), which imply higher required transmit power, or in economic terms a higher production cost in (4.41). By setting a higher selling price μ_2 , system 2 becomes profitable enough so that it is scheduled to produce at a sufficient rate to meet the requirement.

In Table 1 we show the average transmission rates selected by the online algorithm, i.e., the average number of time slots where each system *i* was selected to transmit (with a positive power level) at each frequency *j* as $1/N\sum_{k=1}^{N} \alpha_{ij,k} \mathbb{I}\left(p_{ij,k} > 0\right)$. System 3 was scheduled mainly at frequency 2, exploiting its better channel quality. This forced systems 1 and 2 to use frequency 1 more often. Also system 1, which has higher control requirement, transmitted more often than the other systems. This behavior resulted from the online algorithm using only an observed channel sequence, not any prior knowledge on the channel quality distribution.

Finally, we examine the evolution of the three heating control systems when the online algorithm is employed for scheduling and power decisions. Suppose that for all systems *i* the states x_i , which measure deviations from reference room temperatures, are perturbed by disturbances $w_{i,k}$ as in (4.1), which we model as independent Gaussian with mean zero and variance $W_i = 1$. We plot in Fig. 14 the evolution of the empirical quadratic averages $1/N\sum_{k=1}^{N} x_{i,k}^2$. Recall that when the Lyapunov condition (4.11) is satisfied, we get from (2.13) of Section 2.1.1 that the expected limit quadratic costs are bounded by $W_i/(1 - \rho_i)$. We observe from Fig. 14 that after some initial


Figure 13: Evolution of dual variables μ_k over time *k* using the online algorithm. After a number of steps the dual variables μ_k remain in a neighborhood around the optimal values μ^* .



Figure 14: Average quadratic costs during the online scheduling and power allocation algorithm. The stochastic algorithm keeps the average quadratic cost of each control system close to the upper bound of the limit expected cost, shown with dashed lines, induced theoretically by the required Lyapunov decrease rates.

transient the online communication algorithm keeps the empirical average quadratic costs close to the theoretical upper bounds.

4.5. Proofs

4.5.1. Proof of Proposition 4.1

For simplicity we drop the indices *i* within the proof. The expectation over the next system state x_{k+1} on the left hand side of (4.11) accounts via (4.1) for the randomness introduced by the process noise w_k and the random packet success γ_k . In particular we have that

$$\mathbb{E}\left[V(x_{k+1}) \mid x_k\right] = \mathbb{P}(\gamma_k = 1) \ x_k^T A_c^T P A_c x_k + \mathbb{P}(\gamma_k = 0) \ x_k^T A_o^T P A_o x_k + Tr(PW).$$
(4.43)

Here we used the fact that the random variable γ_k is independent of the system state x_k and depends just on the communication model. Plugging (4.43) at the left hand side of (4.11) we get

$$\mathbb{P}(\gamma_k = 1) \ x_k^T A_c^T P A_c x_k + \mathbb{P}(\gamma_k = 0) \ x_k^T A_o^T P A_o x_k \le \rho x_k^T P x_k.$$
(4.44)

Since condition (4.11) needs to hold for all $x_k \in \mathbb{R}^n$ we can rewrite (4.44) as a linear matrix inequality (Boyd and Vandenberghe (2009))

$$\mathbb{P}(\gamma_k = 1)A_c^T P A_c + (1 - \mathbb{P}(\gamma_k = 1))A_o^T P A_o \leq \rho P,$$
(4.45)

The values $\mathbb{P}(\gamma_k = 1)$ that satisfy this linear matrix inequality belong in some closed convex set. We know by assumption that the case $\mathbb{P}(\gamma_k = 1) = 1$ belongs in this set by the assumption that V(x) is a Lyapunov function for the closed loop mode of the system. As a result there is a minimum value *c*, given by the semidefinite program (4.13), such that condition (4.45) is equivalent to $\mathbb{P}(\gamma_k = 1) \ge c$.

4.5.2. Proof of Proposition 4.2

Statement (a) under assumptions 4.1 and 4.3 follows immediately from (Ribeiro, 2012, Theorem 1) where a similar optimization setup is examined. The proof is omitted due to space limitations.

To show (b) observe that, by definition of the dual function in (4.17), at the point μ^* we have that

$$g(\mu^*) \le L(\alpha^*, p^*, \mu^*)$$
 (4.46)

Since μ^* is optimal for (4.19) and using part (a) we have for the left hand side of (4.46) that $g(\mu^*) = D = P$. On the other hand, the right hand side of (4.46), by the definition of the Lagrangian at (4.16), equals

$$L(\alpha^{*}, p^{*}, \mu^{*}) = P + \sum_{i=1}^{m} \mu_{i}^{*} \left[c_{i} - \mathbb{E}_{h} \sum_{j=1}^{f} \alpha_{ij}^{*}(h) q(h_{ij}, p_{ij}^{*}(h)) \right],$$
(4.47)

because the objective of (4.14) at the optimal solution (α^* , p^*) equals the optimal value *P*. These expressions for the left and right hand sides of the inequality in (4.46) therefore give

$$P \le P + \sum_{i=1}^{m} \mu_i^* \left[c_i - \mathbb{E}_h \sum_{j=1}^{f} \alpha_{ij}^*(h) \, q(h_{ij}, p_{ij}^*(h)) \right].$$
(4.48)

This implies that the sum on the right hand side is non-negative. However all summands are non-positive, because $\mu^* \ge 0$ by dual feasibility (4.19), and the terms in the brackets in (4.48) are non-positive because (α^*, p^*) are feasible for the primal problem (4.14). The only possibility is that all summands in (4.48) are identically zero, which proves statement (b).

We have established that (4.48) holds with equality, so by tracing back our steps, we have that (4.46) holds with equality too, which, by the definition of the dual function on (4.17) translates to

$$\inf_{\alpha, p \in (\mathcal{A}, \mathcal{P})} L(\alpha, p, \mu^*) = L(\alpha^*, p^*, \mu^*).$$
(4.49)

This verifies statement (c).

4.5.3. Proof of Proposition 4.3

We first show part (a) of the proposition. Consider the problem of minimizing the Lagrangian as given at the form (4.26) over variables $\alpha(.)$, p(.) for some $\mu \in \mathbb{R}^m_+$. Since $\mu^T c$ is constant the problem is equivalent to ³

$$\inf_{\alpha, p \in (\mathcal{A}, \mathcal{P})} \mathbb{E}_h \sum_{i,j} \alpha_{ij}(h) \left[p_{ij}(h) - \mu_i q(h_{ij}, p_{ij}(h)) \right].$$
(4.50)

Without loss of generality we can exchange the expectation over *h* and the minimization over functions $\alpha(.)$, p(.) in (4.50) to equivalently solve for each $h \in \mathcal{H}^{m \times f}$

$$\inf_{\substack{\alpha(h)\in\Delta_{m,f}\\p(h)\in[0,p_{\max}]^{m\times f}}}\sum_{i,j}\alpha_{ij}(h)\left[p_{ij}(h)-\mu_iq(h_{ij},p_{ij}(h))\right]$$
(4.51)

This step is valid because any pair of functions α , p that does not minimize the objective in (4.51) on a set of values of variables h with ϕ -positive measure must yield a strictly larger expected value in the objective of (4.50). In other words, the minimizers of (4.50) can only differ from the minimizers of (4.51) at a set of values for h with measure zero.

Then note that at any $h \in \mathcal{H}^{m \times f}$ and any choice for the variable $\alpha(h)$ we have that $\alpha_{ij}(h) \ge 0$. Hence the optimization over p(h) in (4.51) can be rearranged to

$$\inf_{\substack{\alpha(h)\in\Delta_{m,f}}}\sum_{i,j}\alpha_{ij}(h) \\
\inf_{\substack{p_{ij}(h)\in[0,p_{\max}]}}p_{ij}(h) - \mu_i q(h_{ij}, p_{ij}(h)).$$
(4.52)

The optimization over power variables $p_{i,j}(h)$ in this expression corresponds exactly to (4.27). Using the notation introduced in (4.29), the minimization over scheduling variables $\alpha(h)$ in (4.52) becomes

$$\inf_{\alpha(h)\in\Delta_{m,f}}\sum_{i,j}\alpha_{ij}(h)\,\xi(h_{ij},\mu_i).$$
(4.53)

This is an integer programming problem over $\alpha_{ij} \in \{0, 1\}$ according to the set $\Delta_{m,f}$ (cf.(4.2)). The expression given in (4.28) is a linear programming relaxation of (4.53) by assuming $\alpha_{ij} \ge 0$. The

³Within this proof we denote $\sum_{i=1}^{m} \sum_{j=1}^{f}$ as $\sum_{i,j}$ for compactness.

relaxation is tight (see, e.g., (Bertsimas and Tsitsiklis, 1997, Th. 7.5)), meaning that the optimal solution of (4.28) will be integer and feasible with respect to $\Delta_{m,f}$, hence optimal for (4.53) too.

Now let us prove part (b) of the proposition. We need to show that any pair $\alpha(\mu)$, $p(\mu)$, which are functions of h, that solves (4.50) gives a unique evaluation of $s(\mu)$ given in (4.21). Since $s_i(\mu)$ involves integrating the term

$$\sum_{j=1}^{f} \alpha_{ij}(\mu; h) \, q(h_{ij}, p_{ij}(\mu; h)) \tag{4.54}$$

with respect to the distribution ϕ of $h \in \mathcal{H}^{m \times f}$, it suffices to show that (4.54) is unique ϕ -a.s.

By the argument presented already, minimizing (4.50) is a.s. equivalent to minimizing (4.51). The latter is again equivalent to the problem (4.52) since all $\alpha_{ij}(h) \ge 0$. Note that the only case where the optimizers in (4.51) can differ from the ones obtained in (4.52) is if $\alpha_{ij}(\mu; h) = 0$ for some i, j is optimal at some values $h \in \mathcal{H}^{m \times f}$ and the power minimizer $p_{ij}(\mu; h)$ in (4.51) can be chosen arbitrarily. But this does not affect the computation of $s_i(\mu)$ since (4.54) will equal zero. Hence we only need to show that the minimizers $\alpha(\mu; h), p(\mu; h)$ in (4.52) imply a.s. uniqueness of (4.54).

For values of *h* where the minimizers $\alpha(\mu; h)$, $p(\mu; h)$ of problem (4.52) are unique it is immediate that (4.54) has a unique value, hence we only need to consider *h* where the minimizers are not unique. By Assumption 4.2(c) the minimizer $p(\mu; h)$, which is given in (4.27), is unique for almost all *h*, therefore we only need to focus on the set of values for *h* where the minimizer $\alpha(\mu; h)$, described by (4.28), is not unique.

Let us denote by *E* the set of interest, i.e., the set of $h \in \mathcal{H}^{m \times f}$ where $\alpha(\mu; h)$ in (4.28) is not unique. By considering all possible pairs of multiple solutions $\alpha' \neq \alpha''$ in the finite set $\Delta_{m,f}$, we can rewrite *E* as a union $E = \bigcup_{\alpha' \neq \alpha'' \in \Delta_{m,f}} E_{\alpha',\alpha''}$ where $E_{\alpha',\alpha''} \subseteq \mathcal{H}^{m \times f}$ such that

$$h \in E_{\alpha',\alpha''} \Leftrightarrow \alpha', \alpha'' \in \operatorname*{argmin}_{\alpha \in \Delta_{m,f}} \sum_{i,j} \alpha_{ij} \,\xi(h_{ij},\mu_i). \tag{4.55}$$

In other words, the set $E_{\alpha',\alpha''}$ is the set of values *h* where both α', α'' are optimal for (4.28). The rest of the proof shows that on any $E_{\alpha',\alpha''}$ the value of (4.54) is almost surely unique.

The set $E_{\alpha',\alpha''}$ depends on the shape of the function ξ defined in (4.29), so next we point out two properties of $\xi(h_{ij}, \mu_i)$.

Fact 1: For almost all h_{ij} where the optimal value of problem (4.29) is $\xi(h_{ij}, \mu_i) = 0$, the optimal solution is unique and equals $p_{ij}(\mu; h) = 0$.

Proof of Fact 1: First we note that for any h_{ij} , the choice p = 0 is feasible for problem (4.29) and by Assumption 4.2(a) it gives an objective $p - \mu_i q(h_{ij}, p) = 0$. So whenever the optimal value of problem (4.29) is 0, then p = 0 is an optimal solution. This optimal solution is unique for almost all h_{ij} because of Assumption 4.2(c).

Fact 2: If at some h_{ij} the optimal value of problem (4.29) is $\xi(h_{ij}, \mu_i) < 0$, then for $h'_{ij} > h_{ij}$ we have that $\xi(h'_{ij}, \mu_i) < \xi(h_{ij}, \mu_i)$.

Proof of Fact 2: First note that at the given h_{ij} it must be that the optimal solution $p_{ij}(\mu;h)$ of problem (4.29) satisfies $q(h_{ij}, p_{ij}(\mu;h)) > 0$. This is true because otherwise $q(h_{ij}, p_{ij}(\mu;h)) = 0$ implies $\xi(h_{ij}, \mu_i) = p_{ij}(\mu;h) \ge 0$. Second by Assumption 4.2(b) when q(.) > 0, it is strictly increasing in its argument. Thus we have

$$\xi(h_{ij},\mu_i) = p_{ij}(\mu;h) - \mu_i q(h_{ij}, p_{ij}(\mu;h)) > p_{ij}(\mu;h) - \mu_i q(h'_{ij}, p_{ij}(\mu;h)) \ge \xi(h'_{ij},\mu_i)$$
(4.56)

for $h'_{ij} > h_{ij}$.

Let us now fix some $\alpha' \neq \alpha'' \in \Delta_{m,f}$ and consider the set $E_{\alpha',\alpha''}$. Pick indices i, j where α', α'' differ, i.e., without loss of generality, $\alpha'_{i,j} = 1, \alpha''_{i,j} = 0$. Consider first the case of $h \in E_{\alpha',\alpha''}$ where $\xi(h_{i,j}, \mu_i) = 0$. By Fact 1 above we know that this implies $p_{i,j}(\mu; h) = 0$ is almost surely the unique optimizer of (4.27). But in that case $q(h_{i,j}, p_{i,j}(\mu; h)) = 0$, and the choice of $\alpha_{i,j}(h)$ does not affect the value of (4.54), which is zero.

Second, we examine the set $h \in E_{\alpha',\alpha''}$ where $\xi(h_{ij}, \mu_i) < 0$. We will show that this event happens with ϕ -probability zero. In particular by Assumption 4.1 ϕ has a probability density function on $\mathcal{H}^{m \times f}$, or more formally ϕ is absolutely continuous with respect to the Lebesgue measure on $\mathcal{H}^{m \times f}$. Hence to show that the discussed event has ϕ -measure zero, it suffices to show that it has Lebesgue measure zero. Note that we can upper bound the set as follows

$$E_{\alpha',\alpha''} \bigcap \{h : \xi(h_{ij},\mu_i) < 0\}$$

$$\subseteq \{h : \sum_{i,j} (\alpha''_{ij} - \alpha'_{ij}) \xi(h_{ij},\mu_i) = 0, \quad \xi(h_{ij},\mu_i) < 0\}$$

$$= \{h : \sum_{i \neq i, j \neq j} (\alpha''_{ij} - \alpha'_{ij}) \xi(h_{ij},\mu_i) = \xi(h_{ij},\mu_i) < 0\}$$
(4.57)

The subset in the first step is justified from the fact that, in contrary to the definition of $E_{\alpha',\alpha''}$ in (4.55), we do not take α', α'' to be optimal for problem (4.28). We only require that they yield the same objective in the problem. The second step follows by the appropriately selected indices ι , ι .

We will now argue that the last set in (4.57) has Lebesgue measure zero. If we fix the values of all the variables/coordinates h_{ij} , $i \neq i, j \neq j$, there is at most one value for the variable/coordinate h_{ij} that belongs in the set. The reason is that for values of the h_{ij} coordinate where $\xi(h_{ij}, \mu_i) < 0$, Fact 2 above states that $\xi(h_{ij}, \mu_i)$ is strictly monotonic in h_{ij} . Hence there can be at most one value h_{ij} that equals the sum within the last set of (4.57). This means that the last set in (4.57) can be equivalently described by a mapping from an $m \cdot f - 1$ dimensional space to the space $\mathcal{H}^{m \times f}$, or in other words it is a lower-dimensional subset of $\mathcal{H}^{m \times f}$. Hence it has Lebesgue measure zero. This implies that the first set in (4.57) has Lebesgue (and ϕ) measure zero as well.

The above procedure can be iterated for any pair α', α'' to conclude that in their union set *E* the value of the subgradient vector is almost surely unique.

4.5.4. Proof of Theorem 4.1

Let μ^* be an optimal solution of the dual problem (4.19). First, we argue that every pair $\alpha(\mu^*)$, $p(\mu^*)$ chosen from the set of Lagrangian minimizers $(\mathcal{A}, \mathcal{P})(\mu^*)$ at the point μ^* (cf. (4.18)) is an optimal solution to primal Problem 4.1 (equivalently (4.14)). Under Assumptions 4.1 and 4.2, Proposition 4.3(b) states that the vector $s(\mu^*)$ in (4.21) has the same value at any chosen pair $\alpha(\mu^*)$, $p(\mu^*)$. Since $s(\mu^*)$ is also the constraint slack of the chosen pair in the primal problem (4.14), then any Lagrange optimizers $\alpha(\mu^*)$, $p(\mu^*)$ have the same constraint slack. Moreover, under Assumptions 4.1 and 4.3, Proposition 4.2(c) states that the optimal primal variables α^* , p^* are one such pair of Lagrange optimizers at μ^* , and by definition they have a feasible constraint slack. Hence all Lagrange optimizers $\alpha(\mu^*)$, $p(\mu^*)$ have the same feasible constraint slack as α^* , p^* . Additionally all optimizers $\alpha(\mu^*)$, $p(\mu^*)$ yield the same minimum Lagrangian value $L(\alpha(\mu^*), p(\mu^*), \mu^*)$. By the form of the Lagrangian in (4.16) it follows that all optimizers $\alpha(\mu^*)$, $p(\mu^*)$ also give the same primal objective in (4.14) as the point α^* , p^* , i.e., the minimum P. Hence any optimizer pair $\alpha(\mu^*)$, $p(\mu^*)$ is primal optimal. The first statement of the theorem follows because the scheduling and power allocation obtained by (4.27)-(4.29) at μ^* describe one pair of Lagrange optimizers at μ^* , i.e., are optimal solutions to Problem 4.1.

The convergence of iterations (4.23)-(4.24) to the optimal dual variable μ^* for stepsizes in (4.25) relies on the boundedness of the subgradient vectors (as mentioned after (4.22)) and follows from a standard subgradient method argument – for a proof see, e.g., (Bertsekas et al., 2003, Prop. 8.2.6).

4.5.5. Proof of Proposition 4.4

We begin by noting that at every time *k* the vector s_k computed by (4.31) is a stochastic subgradient for the dual function $g(\mu)$ at the point μ_k , i.e.,

$$g(\mu') - g(\mu_k) \le (\mu' - \mu_k)^T \mathbb{E}[s_k \mid \mu_k] \quad \text{for all} \mu' \in \mathbb{R}^m_+.$$
(4.58)

To show this fact compare equations (4.30)-(4.31) of the online algorithm with (4.21) to conclude that $\mathbb{E}[s_k \mid \mu_k] = s(\mu_k)$ because h_k is i.i.d for every k. Inequality (4.58) then follows directly from (4.20).

Then note that by Assumption 4.3 there exists a strictly feasible primal solution α' , p'. Call P' the resulting objective value (4.14) at this point, and let a positive constant $\varepsilon' > 0$ denote the constraint slack of (4.15) at this point, i.e., $c_i + \varepsilon' \leq \mathbb{E}_h \sum_{i=1}^f \alpha'_{ii}(h) q(h_{ij}, p'_{ij}(h))$. Then we may bound the dual

function (4.17) at the optimal μ^* by

$$D = g(\mu^{*}) \le L(\alpha', p', \mu^{*}) = P' + \sum_{i=1}^{m} \mu_{i}^{*} \left[c_{i} - \mathbb{E}_{h} \sum_{j=1}^{f} \alpha'_{ij}(h) q(h_{ij}, p'_{ij}(h)) \right] \le P' - \sum_{i=1}^{m} \mu_{i}^{*} \varepsilon'$$

Rearranging the above inequality, and since $\mu^* \ge 0$, it follows that $\mu_{\ell}^* \le \sum_{i=1}^m \mu_i^* \le (P' - D)/\varepsilon'$ for every ℓ , i.e., the optimal dual variables are finite.

Since the optimal dual variables are finite, the distance $||\mu_k - \mu^*||$ between any random μ_k obtained by Algorithm 4.1 and the set of optimal dual variables μ^* is well-defined and bounded. The following lemma gives an upper bound on this distance. Here recall that as we commented after (4.21) the subgradients $s(\mu)$ are always bounded in our problem.

Lemma 4.1. Let *D* be the optimal value of the dual problem (4.19), μ^* be an optimal solution, and *S* be the bound on the subgradient $||s(\mu)|| \leq S$ for any $\mu \in \mathbb{R}^m_+$. Then at each step *k* of Algorithm 4.1 the update of μ_{k+1} satisfies

$$\mathbb{E}[\|\mu_{k+1} - \mu^*\|^2 \,|\mu_k] \le \|\mu_k - \mu^*\|^2 + \varepsilon_k^2 S^2 - 2\varepsilon_k (D - g(\mu_k)) \tag{4.59}$$

Proof. First use the expression $\mu_{k+1} = [\mu_k + \varepsilon_k s_k]_+$ in Algorithm 4.1 to write

$$\|\mu_{k+1} - \mu^*\| = \|[\mu_k + \varepsilon_k s_k]_+ - \mu^*\| \le \|\mu_k + \varepsilon_k s_k - \mu^*\|,$$
(4.60)

where the last inequality holds because when projecting on the positive orthant the distance from a point μ^* in the orthant can only decrease. Taking expectation on both sides given μ_k and expanding the square norm of the right hand side, we get

$$\mathbb{E}[\|\mu_{k+1} - \mu^*\|^2 |\mu_k] \le \|\mu_k - \mu^*\|^2 + \varepsilon_k^2 S^2 + 2\varepsilon_k (\mu_k - \mu^*)^T \mathbb{E}[s_k \mid \mu_k]$$
(4.61)

where we bounded $\|\mathbb{E}[s_k \mid \mu_k]\|^2 < S^2$. Then (4.59) follows from (4.61) by applying inequality (4.58) with $\mu' = \mu^*$.

Based on (4.59), we will use a supermartingale convergence argument to show that $\|\mu_k - \mu^*\|^2 \to 0$ almost surely. Note first that at any $\mu_k \in \mathbb{R}^m_+$ the dual function is smaller than the optimal value (cf. (4.19)), so $D - g(\mu_k) \ge 0$. Hence (4.59) can be simplified to

$$\mathbb{E}[\|\mu_{k+1} - \mu^*\|^2 \,|\mu_k] \le \|\mu_k - \mu^*\|^2 + \varepsilon_k^2 S^2.$$
(4.62)

Then consider the sequence of random variables

$$a_k = \|\mu_k - \mu^*\|^2 + \sum_{\ell \ge k} \varepsilon_l^2 S^2.$$
(4.63)

This stochastic process is: i) measurable with respect to the sequence (filtration) $\mathcal{F}_k = \{\mu_{0:k}\}$, ii) non-negative, iii) integrable because μ_k generated by Algorithm 4.1 is bounded at every *k* and stepsizes are square summable, iv) satisfies $\mathbb{E}[a_{k+1} \mid \mathcal{F}_k] \leq a_k$ as can be seen by definition (4.63) and (4.62).

Such a stochastic process is called a supermartingale (Durrett, 2010, Ch. 5), and a non-negative supermartingale converges almost surely to some limit random variable (Durrett, 2010, Th. 5.2.9). Observe that the second summand $\sum_{\ell=k}^{\infty} \varepsilon_l^2 S^2$ of a_k in (4.63) is deterministic and converges to 0 because of square summability of the stepsizes. Hence the random variable $\|\mu_k - \mu^*\|^2$ converges almost surely (to some limit random variable).

To arrive at a contradiction suppose the limit random random variable is not identically zero. Equivalently, with probability $\delta > 0$ we have $\|\mu_k - \mu^*\|^2 \ge \varepsilon$ for some $\varepsilon > 0$ for all sufficiently large *k*. This implies that μ_k and $g(\mu_k)$ are bounded away from the optimal μ^* and *D* respectively, hence the following expected value diverges,

$$\mathbb{E}\sum_{k\geq 0} 2\varepsilon_k (D - g(\mu_k)) = +\infty.$$
(4.64)

However taking expectation at both sides of (4.59) and iterating for k = 0, ..., N - 1 implies

$$\mathbb{E} \|\mu_N - \mu^*\|^2 \le \|\mu_0 - \mu^*\|^2 + \sum_{k=0}^{N-1} 2\varepsilon_k^2 S^2 - \mathbb{E} \sum_{k=0}^{N-1} 2\varepsilon_k (D - g(\mu_k)).$$
(4.65)

The left hand side is non-negative, but (4.64) implies that in the limit as $N \to \infty$ the right hand side becomes negative. This is a contradiction. Therefore it must be that $\|\mu_k - \mu^*\|^2$ converges to zero with probability 1.

By continuity of the concave dual function $g(\mu)$ we also have that $g(\mu_k)$ converges to $g(\mu^*) = D$ a.s.

4.5.6. Proof of Theorem 4.2 and Corollary 4.1

The proofs of the two results are presented together. First we will show that the statements (4.38) of Theorem 4.2 and (4.40) of Corollary 4.1 hold, by converting them into equivalent ones involving variables relating to the dual problem (4.19). Imitating the steps leading from problem (4.9) to problem (4.14), the statement (4.38) is equivalent to

$$\limsup_{k \to \infty} c_i - \mathbb{E}_{h_k} \left[\sum_{j=1}^f \alpha_{ij,k} \, q(h_{ij,k}, p_{ij,k}) \, \middle| \, \mu_k \right] \le 0 \tag{4.66}$$

holding a.s. with respect to the channel sequence $\{h_k, k \ge 0\}$. Here we exploited the fact that by the online algorithm the variables α_k , p_k conditioned on the value of μ_k are independent of the observed channel history (but μ_k does depend on the whole history). Similarly we see that the statement (4.40) is equivalent to

$$\limsup_{k \to \infty} c_i - \mathbb{E} \sum_{j=1}^f \alpha_{ij,k} q(h_{ij,k}, p_{ij,k}) \le 0$$
(4.67)

The term inside the limit is the expected (i.e., unconditioned) value of the term in (4.66) with respect to the random sequence $\{h_k, k \ge 0\}$.

Condition (4.66) is equivalent to $\limsup_{k\to\infty} \mathbb{E}_{h_k}[s_k \mid \mu_k] \leq 0$ using the expression of s_k given in (4.31). Also we already argued in the proof of Prop. 4.4 that $\mathbb{E}_{h_k}[s_k \mid \mu_k] = s(\mu_k)$ where $s(\mu_k)$ is given by (4.21) and is a subgradient of the dual function g at μ_k . To sum up, (4.38) is equivalent to

$$\limsup_{k \to \infty} s(\mu_k) \le 0 \quad \text{a.s.} \tag{4.68}$$

Similarly, condition (4.40) via (4.67) is equivalent to $\limsup_{k\to\infty} \mathbb{E}s(\mu_k) \leq 0$. But the latter is a consequence of (4.68) because $\limsup_{k\to\infty} \mathbb{E}s(\mu_k) \leq \mathbb{E} \limsup_{k\to\infty} s(\mu_k) \leq 0$. The first inequality follows by applying Fatou's lemma (Durrett, 2010, Thm. 1.6.5) to the bounded below (as we comment after (4.21)) random variable $-s(\mu_k)$. The second inequality follows by monotonicity of expectation. Hence, to prove the statements (4.38) and (4.40) it suffices to show (4.68) which we do next.

Under Assumption 4.3 we have established in Proposition 4.4 that a.s. $\mu_k \to \mu^*$. Then we note a convex analysis fact by (Bertsekas et al., 2003, Prop. 4.2.3). If *g* is concave, and $\mu_k \to \mu^*$, and $s(\mu_k)$ is selected as a subgradient of *g* at μ_k , then every limit point of $s(\mu_k)$ is a subgradient of *g* at μ^* . Hence for the sequence of μ_k obtained by the online algorithm we have that $s(\mu_k)$ converges a.s. to a subgradient of *g* at μ^* .

Also, as follows from Danskin's theorem (Bertsekas et al., 2003, Prop. 4.5.1), the subgradients of the dual function g at any point μ belong in the convex hull of the vectors $s(\mu)$ obtained in (4.21). Hence the sequence $s(\mu_k)$ converges a.s. to the convex hull of the vectors $s(\mu^*)$. But under Assumptions 4.1, 4.2, and 4.3, as we argued in the proof of Theorem 4.1, the vectors $s(\mu^*)$ take a unique value that satisfies $s(\mu^*) \leq 0$. Hence for the sequence of μ_k obtained by the online algorithm we have that $\limsup_{k\to\infty} s(\mu_k) \leq 0$ a.s., which is exactly what we set out to prove in (4.68).

Finally let us prove (4.39). Recall that the dual function equals $g(\mu) = L(\alpha(\mu), p(\mu), \mu)$ where $\alpha(\mu), p(\mu)$ are chosen as Lagrange optimizers at μ according to (4.18). Using the definition of the Lagrangian at (4.16) and the interpretation of the subgradient $s(\mu)$ at (4.21) as the constraint slack,

we have that for any μ_k

$$g(\mu_k) = L(\alpha(\mu_k), p(\mu_k), \mu_k)$$

= $\mathbb{E}_h \sum_{i=1}^m \alpha_i(\mu_k; h) p_i(\mu_k; h) + \mu_k^T s(\mu_k)$ (4.69)

Now observe that the expectation in (4.39) equals the expectation given in (4.69) because by design of Algorithm 4.1 the primal variables α_k , p_k are selected as Lagrange optimizers at μ_k . Therefore to show that (4.39) holds a.s. it suffices to show that the expectation in (4.69) converges a.s. to *P* which equals *D* by strong duality.

Proposition 4.4 establishes that the left hand side of (4.69) converges to $g(\mu_k) \to D$, and also that $\mu_k \to \mu^*$ a.s. We have also already argued that $s(\mu_k) \to s(\mu^*)$ a.s. Therefore also $\mu_k^T s(\mu_k) \to \mu^{*T} s(\mu^*)$ a.s. But by Prop. 4.2(b) $\mu^{*T} s(\mu^*) = 0$. This shows that the expectation at the right hand side of (4.69) converges to D, which completes the proof.

Chapter 5: Random Access Design for Wireless Control Systems

5.1. Problem Formulation

In this chapter a decentralized channel access method is employed for multiple control systems sharing a wireless medium. The architecture was introduced in Section 2.2.2 and is depicted again in Fig. 15. Each sensor i (i = 1, 2, ..., m) in the architecture transmits measurements of plant i to an access point responsible for computing the plant control inputs. Packet collisions might arise on the shared medium between simultaneously transmitting sensors. We are interested in designing a mechanism for each sensor to independently decide whether to access the medium (random access) in a way that guarantees desirable control performance for all control systems.

We denote by $\gamma_{i,k} \in \{0,1\}$ the success of the transmission at time *k* for link/system *i*, determined by the communication algorithm to be designed. As introduced in Section 2.1 the evolution of each system *i* depends on the success of transmission and is described by a switched linear timeinvariant system of the form

$$x_{i,k+1} = \begin{cases} A_{c,i} x_{i,k} + w_{i,k}, & \text{if } \gamma_{i,k} = 1, \\ A_{o,i} x_{i,k} + w_{i,k}, & \text{if } \gamma_{i,k} = 0. \end{cases}$$
(5.1)

Here $x_{i,k} \in \mathbb{R}^{n_i}$ denotes the state of control system *i* at each time *k*. At a successful transmission the system dynamics are described by the asymptotically stable closed-loop matrix $A_{c,i} \in \mathbb{R}^{n_i \times n_i}$, and otherwise by the unstable open-loop matrix $A_{o,i} \in \mathbb{R}^{n_i \times n_i}$. The additive terms $w_{i,k}$ model an independent (both across time *k* for each system *i*, and across systems) identically distributed (i.i.d.) noise process with mean zero and covariance $W_i \succeq 0$. Example of such control systems were presented in Section 2.1.

According to the random access communication mechanism, introduced in Section 2.2.2, at each time step *k* a sensor *i* transmits over the shared wireless medium with a probability $\alpha_{i,k} \in [0, 1]$ which is a design variable. A sensor's transmission might fail due to two reasons, packet decoding errors and packet collisions. A collision might be experienced on link *i*, thereby rendering packet *i* lost, if some other sensor $j \neq i$ transmits in the same time slot. We assume that such a collision event occurs with constant probability $q_{ji} \in [0, 1]$, given that both sensors *i*, *j* transmit in the slot. Thus, the probability that sensor *i*'s transmission is free of collisions, i.e., that no other sensor transmits and causes collisions on link *i*, equals $\prod_{j\neq i} [1 - \alpha_{j,k} q_{ji}]$.



Figure 15: Random access architecture for *m* control loops over a shared wireless medium. Each sensor *i* randomly transmits with probability $\alpha_{i,k}$ at time *k* to a common access point computing the plant control inputs. If only sensor *i* transmits, the successful decoding probability depends on local channel conditions $h_{i,k}$. If other sensors transmit at the same time a collision might occur at sensor *i*'s transmission, rendering *i*'s packet lost.

If sensor *i* transmits and has a collision free time slot, the success of decoding the message at the access point/receiver depends on the randomly varying channel conditions on link *i*. Employing the block fading wireless channel model introduced in Section 2.2 we denote by $h_{i,k} \in \mathbb{R}_+$ the current fading state for link *i* at time *k*, which is assumed i.i.d. across time with distribution ϕ_i . We also assume channel states are independent among systems *i*, a common assumption in the literature Adireddy and Tong (2005); Qin and Berry (2006); Hu and Ribeiro (2011), as well as independent of the plant process noise $w_{i,k}$. The probability of successful decoding, in absence of collisions, depends on the current channel fading and transmit power of the sensor. For simplicity in this section we assume the transmit power for each sensor *i* is fixed to some constant value p_i . Then the decoding probability is given by a relationship of the form $q(h_{i,k}, p_i)$. An illustration of this function is depicted in Fig. 2. Since power p_i is fixed in this setup (not a design variable) it is omitted and the decoding probability is denoted as $q(h_{i,k})$ in the rest of this chapter. The function $q : \mathbb{R}_+ \to [0, 1]$ is assumed to be continuous and strictly increasing, i.e., higher channel fading states imply higher packet success probability.

Combining the effects of collisions and packet losses due to fading, the probability that a packet is successfully decoded at the access point can be written as

$$\mathbb{P}(\gamma_{i,k}=1) = \alpha_{i,k} q(h_{i,k}) \prod_{j \neq i} \left[1 - \alpha_{j,k} q_{ji} \right].$$
(5.2)

This expression states that the probability of system *i* in (5.1) closing the loop at time *k* equals the probability that transmission *i* is successfully decoded at the receiver, multiplied by the probability that no other sensor $j \neq i$ is causing collisions on *i*th transmission.

Channel states reveal information about how easy it is for each sensor to successfully communicate, assuming no other sensor transmits. We assume that before deciding whether to transmit each sensor has access to its respective channel state and may adapt accordingly. For example sensor *i* may transmit with higher or lower rate $\alpha_{i,k}$ under favorable or unfavorable channel states $h_{i,k}$ respectively. Hence we design policies that are measurable functions of the form $\alpha_{i,k} = \alpha_i(h_{i,k})$. Since channel states are i.i.d. over time we restrict attention to stationary policies, and drop the time index when not necessary. The set of all access policies for sensor *i* is then

$$\mathcal{A}_i = \{ \alpha_i : \mathbb{R}_+ \to [0, 1] \} \tag{5.3}$$

and the vector $\alpha(.)$ of access policies for all sensors belongs in the Cartesian product space $\mathcal{A} = \mathcal{A}_1 \times ... \mathcal{A}_m$. For fixed sensor access policies, the probability of successful transmission on link *i* can be expressed as

$$\mathbb{P}(\gamma_{i,k}=1) = \mathbb{E}_{h_i}[\alpha_i(h_i) q(h_i)] \prod_{j \neq i} \left[1 - \mathbb{E}_{h_j}[\alpha_j(h_j)] q_{ji}\right].$$
(5.4)

This expression follows from (5.2) by taking expectation with respect to the channel states and using the independence of channels among systems. The expectation is well-defined as both functions $\alpha(.)$, q(.) are measurable and bounded in [0, 1] hence integrable.

We make the following technical assumption on the probability distribution of channel states, which holds true for practically considered models (Goldsmith, 2005, Ch. 3).

Assumption 5.1. The distributions ϕ_i of channel states $\{h_{i,k}, k \ge 0\}$ for all i = 1, ..., m are absolutely continuous, i.e., have a probability density function on \mathbb{R}_+ .

The random packet success on link *i* modeled by (5.4) causes each control system *i* in (5.1) to switch in a random fashion between the two modes of operation (open and closed loop). As a result, the sensor access policies $\alpha(.)$ to be designed affect the performance of all control systems. We account for control performance via a Lyapunov-function-based abstraction, as introduced in Section 2.1.1. In particular we assume that a quadratic function $V_i(x_i) = x_i^T P_i x_i$, $x_i \in \mathbb{R}^{n_i}$ with a positive definite matrix $P_i \succ 0$ is provided for each system. To achieve desired control performance *every function* $V_i(x_i)$ *is required to decrease with a desired rate* $\rho_i < 1$ *in expectation at each step*, i.e.,

$$\mathbb{E}\left[V_i(x_{i,k+1}) \mid x_{i,k}\right] \le \rho_i \, V_i(x_{i,k}) + Tr(P_i W_i) \tag{5.5}$$

for all $x_{i,k} \in \mathbb{R}^{n_i}$. The expectation in this expression accouns for the randomness introduced by the wireless communication as well as the plant process noise.

By Proposition 4.1 of Section 2.1.1 the above Lyapunov control performance requirements are equivalent to guaranteeing minimum packet success rates for each link *i*, i.e.,

$$\mathbb{P}(\gamma_{i,k}=1) \ge c_i,\tag{5.6}$$

for some non-negative constants $c_i \ge 0$ that are easily computed (by the semidefinite program (4.45)). Hence to achieve desired control performance we need to ensure that (5.6) holds for all links *i*.

Besides control performance, it is desired that the sensors' channel access mechanism makes an efficient use of their power resources. Since sensor *i* transmits with a fixed power $p_i > 0$ when it decides to access the channel, the total expected power consumption at each slot is given by $\sum_{i=1}^{m} \mathbb{E}_{h_i} \alpha_i(h_i) p_i$ summing up the transmit power of each system *i* if the system decides to transmit. We pose then the design of the sensor access rates α that minimize the total expected power consumption subject to the desired control performance (5.5) (equivalently (5.6)) for all plants as

$$\begin{array}{ll} \underset{\alpha \in \mathcal{A}}{\text{minimize}} & \sum_{i=1}^{m} \mathbb{E}_{h_{i}} \alpha_{i}(h_{i}) p_{i} \\ \text{subject to} & c_{i} \leq \mathbb{E}_{h_{i}} [\alpha_{i}(h_{i})q(h_{i})] \prod_{j \neq i} \left[1 - \mathbb{E}_{h_{j}} [\alpha_{j}(h_{j})] q_{ji} \right] \\ & i = 1, \dots, m. \end{array}$$
(5.7)

Technically we assume that the problem is strictly feasible, as follows.

Assumption 5.2. There exists $\alpha' \in A$ that satisfies constraints (5.8) with strict inequality.

In the following section we proceed to characterize the optimal access policies α^* . In particular we reveal a simple and intuitive decoupled structure. Each sensor *i* independently accesses the channel in a way that trades off the goal of closed loop *i* with the effect of collisions on all other closed loops $j \neq i$ collectively. Later we develop a procedure to find these optimal access policies.

5.2. Channel-aware Random Access Design

Our main result is the following characterization of the optimal access policies for the sensors.

Theorem 5.1 (Optimal sensor access policies). Consider a random access architecture with *m* control loops of the form (5.1), communication modeled by (5.4), and control performance abstracted by (5.5)-(5.6) for each loop i = 1, ..., m. Consider the design of optimal sensor access policies (5.7)-(5.8), and let Assumptions 5.1, 5.2 hold. Then there exists a matrix of non-negative elements $v^* \in \mathbb{R}^{m \times m}_+$ such that the optimal sensor access policy for each sensor i = 1, ..., m is written as

$$\alpha_i^*(h_i) = \begin{cases} 1 & \text{if } \nu_{ii}^* q(h_i) \ge p_i + \sum_{j \ne i} \nu_{ji}^* q_{ij} \\ 0 & \text{otherwise.} \end{cases}$$
(5.9)

We observe the following interesting facts. First note that the optimal policies are deterministic, that is, given current channel conditions each sensor either transmits or not. Second we note that by the assumed strict monotonicity of the packet success function q(.), the optimal sensor access policies in (5.9) are threshold policies. That is, a sensor transmits only when its corresponding

channel quality is above some threshold. The intuitive interpretation is that a sensor should attempt to close its loop only when its channel is sufficiently favorable.

Third, and more importantly, the optimal policies are decoupled among the sensors. That is because the policy α_i^* (or equivalently the threshold for sensor *i*) in (5.9) only depends on parameters pertinent to system *i*, i.e., its transmit power p_i , and the values ν_{ii}^* and $\sum_{j \neq i} \nu_{ji}^* q_{ij}$ which belong in the *i*th column of matrix ν^* . Hence, as long as the matrix ν^* is available, each sensor can select its optimal channel access policy independently of what the other sensors are trying to achieve. We note that decentralized threshold-based policies have also been shown to be optimal for general wireless communication networks Adireddy and Tong (2005); Qin and Berry (2006); Hu and Ribeiro (2011). The context differs however, since in these works the objective is thoughput-based utility functions in contrast to the packet success rates used for control systems here.

As we explain in the proof, the matrix v^* technically corresponds to the optimal Lagrange multiplier of an appropriately defined problem (cf. (5.25)-(5.28)). An intuitive alternative interpretation is as follows. We can think of each diagonal term v_{ii}^* as the importance of control performance of system *i*, and of each offdiagonal term v_{ji}^* as the collision effect that sensor *i* has on system *j*. The optimal access policy for sensor *i* in (5.9), or equivalently the optimal channel threshold, trades off the requirement on loop *i* and the collective negative effect $(\sum_{j \neq i} v_{ji}^* q_{ij})$ on all other control loops $j \neq i$. That is because a larger value v_{ii}^* corresponds to a lower threshold (sensor transmits more often), while a larger value $\sum_{j \neq i} v_{ji}^* q_{ij}$ corresponds to a higher threshold (sensor transmits less often). Note also that the latter summands are normalized by the parameters q_{ij} , i.e., the probability that sensor *i* collides with link *i* when both sensors transmit. Morever, a high transmit power p_i in (5.9) also implies that sensor *i* should access the channel less often to limit expenditures.

The decoupled structure of the optimal sensor access policies in Theorem 5.1 relies on knowing the values v^* . In the following section we develop a distributed iterative procedure to obtain the desired v^* .

Remark 5.1. In our work in Gatsis et al. (2015c) we consider simpler random access policies for the sensors, not taking into account channel state information. In particular we consider that at every time k each sensor i randomly and independently transmits with some constant probability $\tilde{\alpha}_i \in [0, 1]$ to be designed. Similarly to (5.4) the probability of successfully closing each loop is given by $\mathbb{P}(\gamma_{i,k} = 1) = \tilde{\alpha}_i q_{ii} \prod_{j \neq i} [1 - \tilde{\alpha}_j q_{ji}]$. It turns out (Gatsis et al., 2015c, Theorem 2) that the optimal access rates $\tilde{\alpha}^*$, i.e., the solution to a problem equivalent to the channel-aware setup in (5.7)-(5.8) can be expressed as

$$\tilde{\alpha}_i = \frac{\tilde{\nu}_{ii}}{p_i + \sum_{j \neq i} \tilde{\nu}_{ji} q_{ij}}$$
(5.10)

for each $i \in \{1, ..., m\}$ for some non-negative matrix $\tilde{v} \in \mathbb{R}^{m \times m}_+$. The matrix \tilde{v} here has the same interpretation as the matrix v^* of Theorem 5.1 but the two matrices are different as they correspond to different problems. Hence we see that for the non-channel-aware case the sensors need to randomize $(0 < \tilde{\alpha}_i < 1)$. In contrast, conditioned on channel state information being available the optimal policies for the

sensors are deterministic, exploiting favorable channel conditions to transmit.

We can now capitalize on the fact that the optimal sensor access policies that guarantee control performance of all closed loop systems and minimize power expenditures are characterized in terms of some appropriate matrix of values v^* (Theorem 5.1). We develop an iterative procedure to determine the optimal sensor access policies by computing these values v^* . The procedure is distributed and easily implementable in the architecture of Fig. 15. In particular the common access point/controller is responsible for finding v^* and communicates them to the sensors via the reverse channel, so that the sensors do not need to directly coordinate or communicate among themselves.

Technically as we have argued in the proof of Theorem 5.1 the values v^* are the optimal Lagrange dual variables of an appropriately defined problem (cf.(5.25)-(5.28)). The iterative procedure presented in Algorithm 5.1 corresponds mathematically to a dual subgradient algorithm (Bertsekas et al., 2003, Ch. 8) to find the optimal dual variables v^* . Alternatively we can interpret the procedure as a distributed implementation in the wireless control architecture of Fig. 15 as follows.

At each period *t* the access point/controller of Fig. 15 maintains a tentative matrix of values v(t). At the beginning of each period, the access point (AP) sends to each sensor *i* the values $v_{ii}(t)$ and $\sum_{j \neq i} v_{ji}(t)q_{ij}$ via the reverse channel (Step 3). For the rest of the period *t* each sensor uses a random access policy $\alpha(h_i, ; t)$ as if the received values v(t) corresponded to the optimal v^* (Step 4). Here $\alpha(h_i; t)$ denotes the valuation of the policy during period *t* at any channel state $h_i \in \mathbb{R}_+$. Then the AP measures the gap between desired and current control performance of each system during this period and updates the values v(t) to v(t+1) to prepare for the next period (Step 7). To perform this update the AP needs to compute⁴ the average transmission and packet success rates for each system during this period (Step 5) and keep track of some auxiliary variables (Step 6).

This algorithm is guaranteed to converge to the optimal sensor access policies as we state next.

Theorem 5.2 (Sensor access policy optimization). Consider the setup of Theorem 5.1. The iterations of Algorithm 5.1 with stepsizes in (5.14)-(5.16) satisfying $\sum_{t\geq 0} \varepsilon(t)^2 < \infty$, $\sum_{t\geq 0} \varepsilon(t) = \infty$ converge to the optimal sensor access policies, *i.e.*,

$$c_{i} \leq \lim_{t \to \infty} \mathbb{E}_{h_{i}}[\alpha_{i}(h_{i};t)q(h_{i})] \prod_{j \neq i} \left[1 - \mathbb{E}_{h_{j}}[\alpha_{j}(h_{j};t)]q_{ji}\right],$$
(5.17)

for all $i = 1, \ldots, m$, and

$$\lim_{t \to \infty} \sum_{i=1}^{m} \mathbb{E}_{h_i}[\alpha_i(h_i; t)] p_i = \sum_{i=1}^{m} \mathbb{E}_{h_i}[\alpha_i^*(h_i)] p_i.$$
(5.18)

⁴Here we assume that even when collisions arise the AP can identify which sensor transmits at each time slot. Hence it can measure the average rate $\mathbb{E}_{h_i}[\alpha_i(h_i;t)]$ at which each sensor *i* accesses the channel, as well as the term $\mathbb{E}_{h_i}[\alpha_i(h_i;t) q(h_i)]$ which is the packet success ratio when only sensor *i* transmits.

Algorithm 5.1 Distributed random access computation

- 1: Initialize $\lambda(0) \in \mathbb{R}^m_+, \nu(0) \in \mathbb{R}^{m \times m}_+$ at the AP
- 2: **loop** At period t = 0, 1, ...
- 3: AP sends $\nu_{ii}(t)$, $\sum_{i \neq i} \nu_{ji}(t) q_{ij}$ to each sensor *i*.

 $\lambda_i(t$

4: During the period each sensor *i* accesses the channel according to policy

$$\alpha_i(h_i;t) \leftarrow \begin{cases} 1 & \text{if } \nu_{ii}(t) q(h_i) \ge p_i + \sum_{j \ne i} \nu_{ji}(t) q_{ij} \\ 0 & \text{otherwise.} \end{cases}$$
(5.11)

- 5: AP measures $\mathbb{E}_{h_i}[\alpha_i(h_i;t)q(h_i)]$, $\mathbb{E}_{h_i}[\alpha_i(h_i;t)]$ for all sensors i = 1, ..., m during the period.
- 6: AP computes the auxiliary variables

$$\beta_{ii}(t) \leftarrow \left[\frac{\lambda_i(t)}{\nu_{ii}(t)}\right]_{\mathcal{B}}$$
(5.12)

$$\beta_{ji}(t) \leftarrow \left[1 - \frac{\lambda_i(t)}{\nu_{ij}(t)}\right]_{\mathcal{B}}$$
(5.13)

for all $i \neq j \in \{1, ..., m\}$, where $[]_{\mathcal{B}}$ denotes the projection to the set defined in (5.24). 7: AP updates the new dual variables

$$\nu_{ii}(t+1) \leftarrow \begin{bmatrix} \nu_{ii}(t) + \\ \varepsilon(t) \left(\beta_{ii}(t) - \mathbb{E}_{h_i}[\alpha_i(h_i;t)q(h_i)]\right) \end{bmatrix}_+$$
(5.14)

$$v_{ij}(t+1) \leftarrow \begin{bmatrix} v_{ij}(t) + \\ c(t) \left(\mathbb{E}_{+} \left[v_{i}(h,t) \right] a_{i} - \beta_{ii}(t) \right) \end{bmatrix}$$
(5.15)

$$\varepsilon(t) \left(\mathbb{E}_{h_{j}} [\alpha_{j}(n_{j};t)] q_{ji} - \beta_{ji}(t) \right) \Big|_{+}$$

$$+1) \leftarrow \left[\lambda_{i}(t) + \varepsilon(t) \left(\log(c_{i}) - \log(\beta_{ii}(t)) \right) \right]_{+}$$
(5.15)

$$-\sum_{j\neq i} \log(1-\beta_{ji}(t)) \Big]_+$$
(5.16)

or all $i \neq j \in \{1, ..., m\}$, where $[]_+$ denotes the projection to the non-negatives \mathbb{R}_+ . 8: **end loop**

The caveat of this distributed implementation is that it requires information exchange between sensors and the access point, hence it introduces some communication overhead. This overhead however burdens mainly the access point which is typically a base station with more capabilities compared to the simpler wireless sensors.

5.3. Numerical simulations

We present a numerical example of the random access design. We consider a case with m = 2 scalar control systems of the form (5.1). We assume the first system has open and closed loop dynamics given by $A_{o,1} = 1.1$, $A_{c,1} = 0.5$ respectively, i.e., it is open loop unstable. We assume the second system has integrator open loop dynamics $A_{o,i} = 1$ and stable closed loop dynamics $A_{c,2} = 0.4$. Both systems are perturbed by zero-mean unit-variance Gaussian noises, hence both



Figure 16: Evolution of dual variables during the optimization algorithm. The elements of the matrix v(t) converge to the optimal values v^* required to obtain the optimal sensor access policies.

system states will diverge unless the closed loops are applied appropriately. The systems are asymmetric, but we model a symmetric control performance requirement. The Lyapunov function $V_i(x_i) = x_i^2$ ($P_i = 1$) for both plants i = 1, 2 is required to decrease with expected rate $\rho_1 = \rho_2 = 0.8$ (cf. (5.5)). By Proposition 4.1 these control performance requirements are equivalent to required packet success rates $c_1 \approx 0.43$, $c_2 \approx 0.27$ for the two sensors, computed by (4.45). Hence System 1, which is more unstable, requires a higher packet success rate.

We assume that both channel states $h_{1,k}$, $h_{2,k}$ are i.i.d. exponential with mean 1. In isolation each sensor faces a packet success probability modeled by the function $q(h_{i,k})$, i = 1, 2 shown in Fig. 2. Also when both sensors transmit at the same time, collisions occur with probability $q_{12} = q_{21} = 0.5$. The transmit powers are taken equal $p_1 = p_2 = 1$.

We solve the random access design problem (5.7)-(5.8) by implementing Algorithm 5.1, which as explained in the previous section solves the problem in the dual domain. We note that at each iteration of the algorithm some expectations with respect to the channel state distributions need



Figure 17: Channel thresholds corresponding to the access policies selected by the optimization algorithm. The channel thresholds for both sensors converge to their optimal values in the limit. Sensor 1 has a lower threshold, i.e., transmits more often, since it is required to guarantee control performance for a more demanding (unstable) plant.



Figure 18: Evolution of control systems using the optimal random access policies. Both systems remain stable despite collisions and packet drops. Also their long run average quadratic cost converges to the same value, since be design both systems were required to have the same control performance.

to be computed, in particular in steps (5.14) and (5.15) of Algorithm 5.1. In our simulations we approximate these expectations with averages from a large number of samples, since samples from the exponential channel distributions can be readily simulated. The iterates of the matrix dual variables v(t) during the simulation are shown in Fig. 16 where we observe that they converge to the optimal values v^* , as was also shown in the proof of Theorem 5.2. We also plot the evolution of the sensor access policies $\alpha_i(h_i;t)$, or equivalently the thresholds of these policies during the simulation of the algorithm in Fig. 17. As also established in Theorem 5.2 the channel thresholds converge to their optimal values in the limit. We observe that the Sensor 1 has a lower threshold, meaning that it transmits more often, which is natural since it corresponds to the unstable plant.

After the optimal access policies (equivalently channel thresholds) have been found, we simulate the random access architecture with the obtained. In Fig. 18 we plot the empirical average long term quadratic cost of the systems $1/N\sum_{k=1}^{N} x_{i,k}^2$ for each system i = 1, 2. We first observe that

both systems remain stable despite packet collisions over the shared channel. Moreover, even though the two systems are asymmetric, both long term average costs converge to the same value because we required the same control performance for both systems. More specifically this long term cost equals the value $\text{Tr}(P_iW_i)/(1-\rho_i) = 1/(1-0.8) = 5$ for both systems i = 1, 2, as noted in Section 2.1.1. Hence even though the two plants have different dynamics, the obtained channel access policies provide symmetric performance by design. The empirical rates $1/N\sum_{k=1}^{N} \alpha_{i,k}$ at which each sensor transmits equal 0.51 and 0.32 for i = 1, 2 respectively. As expected, both sensors access the channel at a rate higher than the respective necessary packet success rate on each link, i.e., $\alpha_i^* > c_i$. This happens because the sensors need to counteract the effect of packet collisions, as well as packet drops due to decoding errors.

5.4. Proofs

5.4.1. Proof of Theorem 5.1

The first part of the proof involves converting problem (5.7)-(5.8) into an equivalent auxiliary optimization problem which has zero duality gap. Then in the second part we use Lagrange duality arguments to show that (5.9) describes an optimal solution for the auxiliary problem.

We begin by a modification to remove the product of the expectations appearing in the constraints (5.8). Taking the logarithm at each side of (5.8) preserves the feasible set of variables by monotonicity. Then the logarithm of the product at the right hand side of (5.8) becomes a sum of logarithms, and we can rewrite the optimal random access design problem equivalently as

$$\begin{array}{ll} \underset{\alpha \in \mathcal{A}}{\text{minimize}} & \sum_{i=1}^{m} \mathbb{E}_{h_{i}} \alpha_{i}(h_{i}) p_{i} \\ \text{subject to} & \log(c_{i}) \leq \log(\mathbb{E}_{h_{i}}[\alpha_{i}(h_{i}) q(h_{i})]) \\ & + \sum_{j \neq i} \log(1 - \mathbb{E}_{h_{j}}[\alpha_{j}(h_{j})]q_{ji}), \end{array}$$

$$(5.20)$$

Here we make an implicit technical assumption that the terms
$$\mathbb{E}_{h_i}[\alpha_i(h_i) q(h_i)]$$
 and $\mathbb{E}_{h_i}[\alpha_i(h_i)] q_{ij}$
in (5.20), which in general take values in the unit interval [0,1] as all involved variables belong
there too, are bounded away from 0 and 1. Then the logarithms in (5.20) are well-defined and
finite. This does not restrict the feasible set of solutions, as intuitively each sensor *i* can neither
choose $\alpha_i(h_i)$ too close to 0 otherwise it cannot meet its packet success requirement in (5.8), nor
too close to 1 otherwise it causes significant packet collisions on other sensors.

i = 1, ..., m.

Next, we replace the term $\mathbb{E}_{h_i}[\alpha_i(h_i)q(h_i)]$ in constraint (5.20) by an auxiliary variable β_{ii} for i = 1, ..., m, and the terms $\mathbb{E}_{h_j}[\alpha_j(h_j)]q_{ji}$ in (5.20) by variables β_{ji} for $j \neq i$. Hence we rewrite (5.20) as

$$\log(c_i) \le \log(\beta_{ii}) + \sum_{j \ne i} \log(1 - \beta_{ji}).$$
(5.21)

To force the auxiliary variables to behave like the expectations we introduce additional constraints of the form

$$\beta_{ii} \leq \mathbb{E}_{h_i}[\alpha_i(h_i) q(h_i)] \tag{5.22}$$

$$\beta_{ji} \geq \mathbb{E}_{h_j}[\alpha_j(h_j)] q_{ji}$$
(5.23)

for all $i, j \in \{1, ..., m\}, j \neq i$. Each of these variables are restricted to a subset

$$\beta_{ij} \in \mathcal{B} = [\beta_{\min}, \beta_{\max}] \tag{5.24}$$

of the unit interval [0, 1]. In a matrix form $\beta \in \mathcal{B}^{m \times m}$. These upper and lower bounds guarantee that all logarithms at constraints (5.21) are finite, as we also assumed for constraint (5.20). Overall we formulate the auxiliary optimization problem

$$\min_{\alpha \in \mathcal{A}, \beta \in \mathcal{B}^{m \times m}} \sum_{i=1}^{m} \mathbb{E}_{h_i} \alpha_i(h_i) p_i$$
(5.25)

subject to
$$\log(c_i) \le \log(\beta_{ii}) + \sum_{j \ne i} \log(1 - \beta_{ji})$$
 (5.26)

$$\beta_{ii} \le \mathbb{E}_{h_i}[\alpha_i(h_i) q(h_i)] \tag{5.27}$$

$$\beta_{ji} \ge \mathbb{E}_{h_j}[\alpha_j(h_j)] \, q_{ji} \tag{5.28}$$

$$i, j \in \{1, \ldots, m\}, j \neq i$$

We argue that this auxiliary problem is equivalent to the original one in (5.7)-(5.8), in the sense that a feasible solution of one problem corresponds to a feasible solution with the same objective value for the other problem. Indeed let's start with a feasible solution α for (5.7)-(5.8). Let us define variables β that make (5.27), (5.28) hold with equality for all $i, j \in \{1, ..., m\}, j \neq i$. Then the pair α, β is also feasible for problem (5.25)-(5.28) and has the same objective. Reversely, consider a feasible pair α, β for problem (5.25)-(5.28). Without loss we can assume that all constraints (5.27)-(5.28) hold with equality. Otherwise if, say, an inequality *i* in (5.27) is strict, we can increase the value of variable β_{ii} till equality in (5.27) is reached without loss of feasibility in (5.26) and without changing the objective value in (5.25). A similar procedure can be performed if some inequality (5.28) is strict, leading finally to a new feasible point satisfying (5.27)-(5.28) with equalities. Then it is immediate that α is also feasible for (5.8) and has the same objective.

Based on the established equivalence, in the rest of the proof it suffices to show that (5.9) describes an optimal solution for the auxiliary problem (5.25)-(5.28). The advantage of formulating this auxiliary problem is that it has zero duality gap as can be shown by the results in Ribeiro (2012). To formally state this result, let us denote the optimal value of this problem by P^* (finite by feasibility Assumption 4.3) and let us define the Lagrange dual problem. We associate dual variables $\lambda_i \ge 0$ with inequalities (5.26), $\nu_{ii} \ge 0$ with (5.27), and $\nu_{ij} \ge 0$ with (5.28), for $i, j \in \{1, ..., m\}$. We write the Lagrangian function as

$$L(\alpha, \beta, \lambda, \nu) = \sum_{i=1}^{m} \mathbb{E}_{h_i} \alpha_i(h_i) p_i$$

+
$$\sum_{i=1}^{m} \lambda_i \Big\{ \log(c_i) - \log(\beta_{ii}) - \sum_{j \neq i} \log(1 - \beta_{ji}) \Big\}$$

+
$$\sum_{i=1}^{m} \nu_{ii} (\beta_{ii} - \mathbb{E}_{h_i}[\alpha_i(h_i) q(h_i)])$$

+
$$\sum_{i=1}^{m} \sum_{j \neq i} \nu_{ij} (\mathbb{E}_{h_j}[\alpha_j(h_j)] q_{ji} - \beta_{ji})$$
(5.29)

Here the dual variables take values $\lambda \in \mathbb{R}^m_+$, $\nu \in \mathbb{R}^{m \times m}_+$. We can rearrange the terms of the Lagrangian in the form

$$L(\alpha, \beta, \lambda, \nu) = \sum_{i=1}^{m} \left\{ \mathbb{E}_{h_{i}} \alpha_{i}(h_{i}) \left[p_{i} + \sum_{j \neq i} \nu_{ji} q_{ij} - \nu_{ii} q(h_{i}) \right] + \sum_{j \neq i} \left[-\lambda_{i} \log(1 - \beta_{ji}) - \nu_{ij} \beta_{ji} \right] + \nu_{ii} \beta_{ii} - \lambda_{i} \log(\beta_{ii}) + \lambda_{i} \log(c_{i}) \right\}.$$
(5.30)

This form is useful because each primal variable ($\alpha_i(h_i)$ and β_{ji} for each *i*, *j*) is decoupled from the others, a fact we will exploit next. Then we can define the Lagrange dual function

$$g(\lambda,\nu) = \inf_{\alpha \in \mathcal{A}, \beta \in \mathcal{B}^{m \times m}} L(\alpha,\beta,\lambda,\nu),$$
(5.31)

as well as the Lagrange dual problem whose optimal value we denote by D^* as

$$D^* = \inf_{\lambda \in \mathbb{R}^m_+, \nu \in \mathbb{R}^{m \times m}_+} g(\lambda, \nu).$$
(5.32)

Then we can establish the following zero duality property about the auxiliary problem (5.25)-(5.28).

Proposition 5.1 (Strong Duality). Let Assumptions 5.1 and 5.2 hold. Then the problem (5.25)-(5.28) has zero duality gap, i.e., $P^* = D^*$. Moreover if α^* , β^* are optimal solutions and λ^* , ν^* are optimal solutions for the dual problem (5.32), then

$$\alpha^*, \beta^* \in \operatorname*{argmin}_{\alpha \in \mathcal{A}, \beta \in \mathcal{B}^{m \times m}} L(\alpha, \beta, \lambda^*, \nu^*).$$
(5.33)

The result follows from (Ribeiro, 2012, Theorems 1 and 4) where general stochastic optimization problems of the form (5.25)-(5.28) are examined under absolute continuity (Assumption 5.1) and strict feasibility (Assumption 5.2). The proof is omitted due to space limitations.

The above characterization suggests that we can recover the optimal variables α^* , β^* of our problem by just minimizing the unconstrained Lagrangian function. A technical caveat of (5.33) is that it describes an inclusion only, implying that in general there might be Lagrangian minimizers that are not optimal. The following lemma excludes such cases by establishing that the Lagrangian minimizers α , which are functions, i.e., infinite-dimensional variables, are unique up to a set of measure zero. Moreover the following lemma gives an explicit expression for these minimizers.

Lemma 5.1. Consider any dual variables $\lambda \in \mathbb{R}^m_+$, $\nu \in \mathbb{R}^{m \times m}_+$. Then the functions $\alpha \in \mathcal{A}$ that minimize the Lagrangian $L(\alpha, \beta, \lambda, \nu)$ are uniquely defined except for a set of arguments $h \in \mathbb{R}^m_+$ of measure zero, and are given by

$$\alpha_{i}(h_{i};\lambda,\nu) = \begin{cases} 1 & \text{if } \nu_{ii}q(h_{i}) \geq p_{i} + \sum_{j \neq i} \nu_{ji}q_{ij} \\ 0 & \text{otherwise.} \end{cases}$$
(5.34)

for each i = 1, ..., m and for every value $h_i \in \mathbb{R}_+$.

In (5.34) the term $\alpha_i(h_i; \lambda, \nu)$ denotes the function α_i that minimizes the Lagrangian $L(\alpha, \beta, \lambda, \nu)$ at given dual points λ, ν evaluated at an argument h_i .

To sum up we have shown in (5.33) that the optimal solution $\alpha(.)$ to problem (5.25)-(5.28) belongs in the set of Lagrange minimizers at λ^* , ν^* , and by Lemma 5.1 these minimizers are unique up to a set of measure zero. As a result, all these minimizers will have the same objective and constraint slack in problem (5.25)-(5.28), and they will all be optimal for this problem. In particular, the specific minimizer defined by $\alpha_i(h_i; \lambda^*, \nu^*)$ given in (5.34) will be optimal for the problem, and corresponds exactly to the one given in (5.9) at the statement of the theorem.

5.4.2. Proof of Lemma 5.1

Consider the problem of minimizing the Lagrangian $L(\alpha, \beta, \lambda, \nu)$ over variables $\alpha \in A, \beta \in B^{m \times m}$. Due to the separability of the Lagrangian given in the form (5.30) over variables α, β , we can separate the problem into subproblems

$$\underset{\beta_{ii} \in \mathcal{B}}{\operatorname{argmin}} \qquad -\lambda_i \log(1 - \beta_{ji}) - \nu_{ij} \beta_{ji} \tag{5.35}$$

$$\underset{\beta_{ii} \in \mathcal{B}}{\operatorname{argmin}} \quad \nu_{ii}\beta_{ii} - \lambda_i \log(\beta_{ii}) \tag{5.36}$$

$$\underset{\alpha_i \in \mathcal{A}_i}{\operatorname{argmin}} \qquad \mathbb{E}_{h_i} \alpha_i(h_i) \left[p_i + \sum_{j \neq i} \nu_{ji} q_{ij} - \nu_{ii} q(h_i) \right]$$
(5.37)

for $i, j \in \{1, ..., m\}, i \neq j$. Next we need to verify that (5.34) is optimal for (5.37). Note that without loss of generality we can exchange the expectation operator \mathbb{E}_{h_i} and the minimization over $\alpha_i \in \mathcal{A}_i$, which is a function $\alpha_i : \mathbb{R}_+ \to [0, 1]$ defined for any channel value $h_i \in \mathbb{R}_+$, to

equivalently solve

$$\underset{\alpha_{i}(h_{i})\in[0,1]}{\operatorname{argmin}} \alpha_{i}(h_{i}) \left[p_{i} + \sum_{j\neq i} \nu_{ji}^{*} q_{ij} - \nu_{ii}^{*} q(h_{i}) \right].$$
(5.38)

pointwise at all values $h_i \in \mathbb{R}_+$. This is valid because any function α_i that minimizes (5.37) can differ form the minimizer in (5.38) at a set of values $h_i \in \mathbb{R}_+$ with measure at most zero.

Then we can verify that (5.34) is the minimizer in (5.38). That is because the right hand side in (5.38) is a linear expression of $\alpha_i(h_i) \in [0, 1]$. Hence the minimizer $\alpha_i(h_i)$ is uniquely defined, and takes values either 0 or 1 except for the values h_i where $p_i + \sum_{j \neq i} v_{ji}^* q_{ij} - v_{ii}^* q(h_i) = 0$. In the latter case the minimizer is not uniquely defined. However due to the strict monotonicity assumption for $q(h_i)$ this case occurs for at most one value h_i , hence it is a measure zero event due to the absolute continuity of the measure ϕ_i by Assumption 5.1. This completes the proof.

We also note for future reference the terms β that minimize the Lagrangian. Since (5.35), (5.36) are strongly convex, their minimizers are unique and satisfy the first order conditions $\partial L/\partial \beta = 0$, that is

$$\nu_{ii} - \frac{\lambda_i}{\beta_{ii}} = 0 \tag{5.39}$$

$$\frac{\lambda_i}{1-\beta_{ji}} - \nu_{ij} = 0. \tag{5.40}$$

respectively subject to the box constraints $\beta_{ji} \in \mathcal{B}$ for all $i, j \in \{1, ..., m\}$. As a result the optimal solutions are given by

$$\beta_{ii}(\lambda,\nu) = \left[\frac{\lambda_i}{\nu_{ii}}\right]_{\mathcal{B}}$$
(5.41)

$$\beta_{ji}(\lambda,\nu) = \left[1 - \frac{\lambda_i}{\nu_{ij}}\right]_{\mathcal{B}}$$
(5.42)

for all $i \neq j \in \{1, ..., m\}$, where $[]_{\mathcal{B}}$ denotes the projection to the set defined in (5.24).

5.4.3. Proof of Theorem 5.2

A sufficient condition for (5.17) and (5.18) is that

$$\lim_{t \to \infty} \mathbb{E}_{h_i}[\alpha_i(h_i; t)] = \mathbb{E}_{h_i}[\alpha_i^*(h_i)]$$
(5.43)

$$\lim_{t \to \infty} \mathbb{E}_{h_i}[\alpha_i(h_i; t)q(h_i)] = \mathbb{E}_{h_i}[\alpha_i^*(h_i)q(h_i)]$$
(5.44)

hold for all i = 1, ..., m. Indeed this immediately implies (5.18), while (5.17) is also implied since the optimal policy α^* for problem (5.7)-(5.8) needs to be feasible. In the proof of Theorem 5.1 we argued that problem (5.7)-(5.8) is equivalent to the auxiliary problem (5.25)-(5.28) with variables $\alpha \in \mathcal{A}, \beta \in \mathcal{B}^{m \times m}$. Hence it suffices to show that the algorithm converges to the optimal solution of this auxiliary problem in the sense of (5.43)-(5.43).

Recall that after introducing dual variables $\lambda \in \mathbb{R}^m_+$, $\nu \in \mathbb{R}^{m \times m}_+$, the Lagrange dual function $g(\lambda, \nu)$ of the auxiliary problem (5.25)-(5.28) is defined in (4.17). We begin by arguing that that at each iteration of the algorithm, the dual variables $\lambda(t)$, $\nu(t)$ according to (5.14)-(5.16) move towards a subgradient direction of the dual function. For convenience let us denote the direction of the steps at (5.14)-(5.15) by the matrix $s_{\nu}(t) \in \mathbb{R}^{m \times m}$ defined as

$$s_{\nu,ii}(t) = \beta_{ii}(t) - \mathbb{E}_{h_i}[\alpha_i(h_i; t) q(h_i)]$$
(5.45)

$$s_{\nu,ij}(t) = \mathbb{E}_{h_i}[\alpha_j(h_j;t)]q_{ji} - \beta_{ji}(t)$$
(5.46)

for all $i \neq j \in \{1, ..., m\}$, and the steps at (5.16) by the vector $s_{\lambda}(t) \in \mathbb{R}^m$ defined as

$$s_{\lambda,i}(t) = \log(c_i) - \log(\beta_{ii}(t)) - \sum_{j \neq i} \log(1 - \beta_{ji}(t))$$
(5.47)

for all $i \in \{1, ..., m\}$. We argue that $s_{\nu}(t), s_{\lambda}(t)$ are subgradient directions for the dual function at the point $\lambda(t), \nu(t)$, i.e., that

$$g(\lambda',\nu') - g(\lambda(t),\nu(t)) \leq (\lambda' - \lambda(t))^T s_{\lambda}(t) + \operatorname{Tr}((\nu' - \nu(t)) s_{\nu}(t))$$
(5.48)

for all $\lambda' \in \mathbb{R}^m_+$, $\nu' \in \mathbb{R}^{m \times m}_+$. This can be shown as follows.

Consider an iteration of Algorithm 5.1. The variable $\alpha(t)$ selected by the algorithm at step (5.11) is a variable that minimizes the Lagrangian $L(\alpha, \beta, \lambda(t), \nu(t))$ with respect to the variable $\alpha \in \mathcal{A}$. This follows directly from Lemma 5.1. Similarly the variables $\beta(t)$ at step (5.13) minimize the Lagrangian function $L(\alpha, \beta, \lambda(t), \nu(t))$ with respect to the variable $\beta \in \mathcal{B}^{m \times m}$. This fact is included in the proof of Lemma 5.1 at (5.41)-(5.42). As a result by the definition of the dual function in (4.17) it follows that $g(\lambda(t), \nu(t)) = L(\alpha(t), \beta(t), \lambda(t), \nu(t))$. Additionally we can substitute the Lagrangian at the right with the form given at (4.16) to get

$$g(\lambda(t), \nu(t)) = \sum_{i=1}^{m} \mathbb{E}_{h_i} \alpha_i(h_i; t) p_i + \lambda(t)^T s_\lambda(t) + \operatorname{Tr}(\nu(t) s_\nu(t)).$$
(5.49)

Here for convenience we replaced the lengthy parentheses of (4.16) by the equivalent terms $s_{\lambda}(t)$, $s_{\nu}(t)$ defined in (5.46), (5.47), (5.45).

Next note that at any point λ', ν' the dual function $g(\lambda', \nu')$ is by definition (4.17) the minimum of the Lagrangian $L(\alpha, \beta, \lambda', \nu')$, hence we must have that $g(\lambda', \nu') \leq L(\alpha(t), \beta(t), \lambda', \nu')$. Using again

the notation $s_{\lambda}(t)$, $s_{\nu}(t)$ at the right hand side we get that

$$g(\lambda',\nu') \leq \sum_{i=1}^{m} \mathbb{E}_{h_i} \alpha_i(h_i;t) p_i + \lambda'^T s_\lambda(t) + \operatorname{Tr}(\nu' s_\nu(t)),$$
(5.50)

Subtracting (5.49) from (5.50) by sides yields (5.48).

To sum up, at each iteration of the algorithm the dual variables $\lambda(t)$, $\nu(t)$ move towards a subgradient direction of the dual function. Additionally the subgradients are bounded. That is true for $s_{\nu}(t)$ because all the terms at the right hand side of (5.45)-(5.46) are between 0 and 1. It is also true for $s_{\lambda}(t)$ because the logarithms at the right hand side of (5.47) are finite by the restriction $\beta(t) \in \mathcal{B}^{m \times m}$ defined in (5.24). Under the bounded subgradient condition, convergence of $\lambda(t)$, $\nu(t)$ to the optimal dual variables λ^* , ν^* for stepsizes as in the statement of the theorem relies on standard subgradient method arguments – see, e.g., (Bertsekas et al., 2003, Prop. 8.2.6) for a proof.

In the rest of the proof, based on the established convergence of the dual variables to the optimal ones, we will show that the same holds for the primal variable $\alpha(.;t)$ in the sense of (5.43)-(5.43). Note that at any iteration t the function $\alpha_i(.;t)$ takes the value 1 when $\nu_{ii}(t)q(h_i) \ge p_i + \sum_{j \neq i} \nu_{ji}(t)q_{ij}$ and 0 otherwise. Due to the strict monotonicity of the function q(.) this is a threshold-like function taking the value 1 when $h_i \ge \bar{h}_i(t) = q^{-1}(p_i + \sum_{j \neq i} \nu_{ji}(t)q_{ij})/\nu_{ii}(t))$. Since we have established that $\nu(t) \to \nu^*$, and since the function q(.) is continuous hence its inverse too, we conclude that the threshold $\bar{h}_i(t)$ converges to $\bar{h}_i^* = q^{-1}(p_i + \sum_{j \neq i} \nu_{ji}^*q_{ij})/\nu_{ii}^*)$. By Theorem 5.1 this limit value equals the threshold of the optimal access policy, which is also of a threshold form.

Hence we conclude that $\alpha_i(.;t) \to \alpha_i^*(.)$ pointwise for all $h_i \in \mathbb{R}_+$ except perhaps for the point h_i^* , i.e., the optimal threshold point. By absolute continuity of the probability measure ϕ_i the point h_i^* has a probability measure zero. Hence $\alpha(.;t) \to \alpha^*(.)$ almost everywhere. Also both sequences of functions $\alpha_i(.;t)$ and $\alpha_i(.;t)q(.)$ are uniformly bounded in [0, 1]. By the Bounded Convergence Theorem (Billingsley, 1995, Theorem 16.5) we conclude that convergence in expectation, i.e., (5.43) and (5.43), also holds. This completes the proof.

Chapter 6: Adaptation to Plant States

The allocation of communication resources in wireless control architectures needs to account for the dynamic evolution of the control system over time. For example, in the problem of power management examined in Chapter 3 it is shown that the optimal transmit power for a sensor needs to be dynamically and opportunistically adapted to the control system state and wireless channel state online. Formally it is shown that such communication design problems can be seen as Markov Decision Process (MDP) problems (Bertsekas (2005)), where the state is composed of the plant state x_k and the channel state h_k , while the decided actions are the resource allocation variables at each time step, e.g. the transmit power p_k . The difficulty in solving these MDP instances lies on the fact that the state space takes a continuum of values rendering standard dynamic programming algorithms, e.g., value or policy iteration (cf. Bertsekas (2005)), computationally hard. In Chapters 4, 5 this difficulty is bypassed using a different control performance abstraction (Lyapunov functions), however the developed solutions excluded the adaptation to physical plant states.

In this chapter we alleviate the technical difficulties of the optimal designs by proposing suboptimal yet tractable resource allocation algorithms that benefit from online adaptation to both control system states and channel conditions. The proposed policies have their basis in the approximate dynamic programming notion of rollout algorithms. More importantly, despite their suboptimality, the proposed algorithms provide control performance and resource utilization guarantees. These guarantees are enforced by design, as the proposed algorithms are designed in order to improve upon simple non-adaptive algorithms that are used as a reference. As we see in numerical simulations the proposed policies achieve in practice significant gains in performance.

We begin by introducing the proposed algorithms in the problem of transmit power allocation for control systems, and then in the scheduling problem. Finally, we propose suboptimal algorithms for the more complicated problem of random access design for control systems.

6.1. Rollout Power Allocation Policies

We consider again the problem of a sensor communicating over a wireless fading channel to a controller in order to control a plant. The control system is described by a switched linear time invariant model of the form (cf. Section 2.1)

$$x_{k+1} = \begin{cases} A_c x_k + w_k, & \text{if } \gamma_k = 1\\ A_o x_k + w_k, & \text{if } \gamma_k = 0 \end{cases}$$
(6.1)

depending on whether a successful transmission occurs at time k or not, indicated with variables $\gamma_k \in \{0, 1\}$. Here $x_k \in \mathbb{R}^n$ denotes the state of the overall control system at time k, and $A_c, A_o \in \mathbb{R}^{n \times n}$ are the stable closed loop and unstable open loop dynamics respectively. The plant disturbance $w_k, k \ge 0$ in (6.1) is an i.i.d. process with some given probability distribution ϕ_w with mean zero and positive definite covariance W.

The wireless channel has an i.i.d. fading state $h_k \in \mathbb{R}_+$, $k \ge 0$ with some probability distribution ϕ_h (cf. Section 2.2). The sensor selects a transmit power level $p_k \in [0, p_{\text{max}}]$ at each time step, and the probability of successfully receiving the message at the controller and closing the loop is given by the relationship

$$\mathbb{P}(\gamma_k = 1) = q(h_k, p_k) \tag{6.2}$$

as described in Section 2.2.

We are interested in maintaining good control performance over a long time horizon. In particular we consider quadratic costs on the system state given by

$$\begin{cases} x_k^T Q_c x_k, & \text{if } \gamma_k = 1\\ x_k^T Q_o x_k, & \text{if } \gamma_k = 0 \end{cases}$$
(6.3)

These costs depend on the success of the transmission at each time step, as introduced in Section 2.1.2. Both matrices Q_c , Q_o are assumed to be positive semidefinite.

Our goal is to design transmit power policies, that is, transmission decisions p_k , $k \ge 0$ adapted to the available plant and channel information x_k , h_k at the sensor at each time step k. The objective is to minimize the long time average control performance described as

$$J_{\text{control}}(p_0, p_1, \ldots) = \limsup_{N \to \infty} \frac{1}{N} \sum_{k=0}^{N-1} \mathbb{E}[x_k^T \left(\gamma_k Q_c + (1 - \gamma_k) Q_o\right) x_k].$$
(6.4)

As the transmit power affects the mode of operation (open or closed) by (6.2), the evolution of the system in (6.1) and the cost in (6.3) lead to a different average expected cost in (6.4). Besides, the transmit power selection should make an efficient use of the available power resources accounted by

$$J_{\text{comm}}(p_0, p_1, \ldots) = \limsup_{N \to \infty} \frac{1}{N} \sum_{k=0}^{N-1} \mathbb{E} p_k.$$
 (6.5)

Our approach to designing control-aware and channel-aware power allocation policies begins by considering simpler policies that will be used as a reference. The proposed policies will then emerge as improvements upon the reference ones.

6.1.1. Reference Policies and Relative Value of Transmission

We consider a family of policies adapting transmit power only to the channel state h_k and not to the plant state x_k . Such a policy can be described as $p_k = \tilde{\mathbf{p}}(h_k)$ for some measurable function $\tilde{\mathbf{p}} : \mathbb{R}_+ \to [0, p_{\text{max}}]$ mapping channel states to power values⁵. We will call such a policy a reference policy. Since channel states are independent of the plant states x_k , as well as independent and identically distributed over time, the policy $\tilde{\mathbf{p}}$ results in a successful packet decoding (cf. (6.2)) with a constant probability at each time step *k* denoted here by \tilde{q} defined as

$$\tilde{q} := \mathbb{P}(\gamma_k = 1) = \mathbb{E}_h[q(h, \tilde{\mathbf{p}}(h))].$$
(6.6)

The expectation in this expression is with respect to the channel distribution ϕ_h . Moreover this reference policy has an average power consumption given by

$$J_{\text{comm}}(\tilde{\mathbf{p}}) = \mathbb{E}_h[\tilde{\mathbf{p}}(h)] \tag{6.7}$$

where we used again the fact that the channel states are i.i.d. over time.

Under the considered reference policy for the system (6.1) the closed loop mode A_c is applied with constant probability \tilde{q} in (6.6) while the open loop mode A_o is applied with the complement probability. The stability of the system under such a reference policy can be characterized using well established results in the context of random jump linear systems. In particular, a direct application of (Costa and Fragoso, 1993, Theorem 1) verifies the following stability result.

Proposition 6.1 (Reference policy stability). Consider the wireless control system (6.1) with communication modeled as (6.2). Consider a reference policy of the form $p_k = \tilde{\mathbf{p}}(h_k), k \ge 0$, leading to a packet success probability \tilde{q} given in (6.6). Then the system is mean square stable, i.e., $\limsup_{k\to\infty} \mathbb{E}x_k x_k^T < \infty$, if and only if

$$\max_{i=1,\dots,n^2} |\lambda_i \{ \tilde{q} A_c^T \otimes A_c^T + (1-\tilde{q}) A_o^T \otimes A_o^T \}| < 1$$

$$(6.8)$$

where \otimes denotes the Kronecker product.

After having established the stability properties of the reference policy, the following proposition characterizes the quadratic control performance of such a reference policy, denoted by $J_{\text{control}}(\mathbf{\tilde{p}})$.

Proposition 6.2 (Reference Policy Performance). Consider the wireless control system (6.1) with communication modeled as (6.2) and control costs described by (6.3). Consider a reference policy of the form $p_k = \tilde{\mathbf{p}}(h_k), k \ge 0$, leading to mean square stability with a packet success probability \tilde{q} given in (6.6). Then there exists a positive semidefinite matrix $P \in \mathbb{S}^n_+$ satisfying the linear matrix equality

$$P = \tilde{q} \left(Q_c + A_c^T P A_c \right) + (1 - \tilde{q}) (Q_o + A_o^T P A_o),$$
(6.9)

⁵Throughout the chapter we use the boldface notation to denote the functions/policies we are interested in designing.

and the average control cost (6.4) equals

$$J_{control}(\tilde{\mathbf{p}}) = Tr(PW). \tag{6.10}$$

The interpretation of the matrix *P* in this proposition is as follows. The quadratic function $x_k^T P x_k$ models the future expected control cost if the sensor follows the reference policy starting from system state x_k at time *k*, i.e., the cost-to-go function of this policy (Bertsekas (2005)). This characterization of the reference policy is important as it leads to an easily computable cost-to-go function by solving (6.9) – see Remark 6.2.

Given a reference policy and the associated cost-to-go function, we will develop policies that adapt to plant and channel states online while they improve upon the control and communication performance of the reference policies. To prepare for the proposed policies we define the matrix

$$M := Q_o + A_o^T P A_o - Q_c - A_c^T P A_c.$$
(6.11)

We interpret the quadratic form $x_k^T M x_k$ as the *relative value of transmission* at the current control system state x_k . Indeed if the sensor successfully transmits ($\gamma_k = 1$), the current control cost is $x_k Q_c x_k$ and the system evolves according to A_c meaning that the future expected costs becomes $x_k^T A_c^T P A_c x_k$ assuming the reference model for the future. Alternatively if no transmission occurs ($\gamma_k = 1$), the current and future control cost becomes $x_k(Q_o + A_o^T P A_o) x_k$. Hence the relative gains if the sensor transmits as compared to not transmitting involve the matrix difference in (6.11).

6.1.2. Rollout policy

We propose then the following power allocation policy $\mathbf{p}_{\nu}^{\text{roll}}$: $\mathbb{R}^n \times \mathbb{R}_+ \rightarrow [0, p_{\max}]$ mapping the current plant and channel conditions x_k, h_k to a transmit power p_k . In particular let the sensor choose the transmit power according to the following minimization

$$p_k = \mathbf{p}_{\nu}^{\text{roll}}(x_k, h_k) := \underset{p \in [0, p_{\text{max}}]}{\operatorname{argmin}} \quad \nu p - q(h_k, p) x_k^T M x_k.$$
(6.12)

Here $\nu \ge 0$ is a non-negative constant. The expression at the right hand side of (6.12) represents a tradeoff between the allocated power (measured by νp) and the expected gains in control performance (given by $x_k^T M x_k$) if the transmission is successful (which happens with probability $q(h_k, p)$).

We call the proposed policy in (6.12) a *rollout* policy. We note that a rollout policy depends on the selected reference policy $\tilde{\mathbf{p}}$ via the matrix *M* appearing in (6.12). One may think of the reference policy as a knob or a tunable parameter that gives different rollout policies. We will not actively design or tune here the reference policy, but will assume one is given to us. However we note that the approaches in Chapters 4, 5 may be useful to come up with good channel-aware reference policies.



Figure 19: The control and communication performance of a selected reference policy is depicted, along with the selected tradeoff line $\nu \ge 0$ between the two objectives. By Theorem 6.1 the control and communication performance resulting from the rollout policy is guaranteed to lie left of the selected tradeoff (shaded area). The actual rollout performance for this example is also depicted. The set of reference policies with constant power allocation $\tilde{p}(.)$ are also depicted for comparison.

The following theorem is the main result of this section and characterizes the performance of the rollout policy.

Theorem 6.1 (Rollout Power Policy Performance). Consider the system described by (6.1) with communication modeled as (6.2) and control costs described by (6.3). Let $\tilde{\mathbf{p}}$ be a given reference policy leading to mean square stability, associated with a matrix M defined in (6.11). For any non-negative constant $v \ge 0$, suppose the sensor follows the rollout policy $p_k = \mathbf{p}_v^{roll}(x_k, h_k)$ given in (6.12). Then the average control and communication costs of this policy satisfy

$$J_{control}(\mathbf{p}_{\nu}^{roll}) + \nu J_{comm}(\mathbf{p}_{\nu}^{roll}) \le J_{control}(\tilde{\mathbf{p}}) + \nu J_{comm}(\tilde{\mathbf{p}}).$$
(6.13)

This theorem provides a guarantee on the joint control performance and power consumption of the proposed rollout policy. In particular it states that the rollout policy performs better that the reference policy with respect to a linear combination of the two objectives. It is worth noting that even though an optimal power policy might be hard to find, the proposed suboptimal policy is given explicitly in (6.12). By tuning the weight $v \ge 0$, i.e., by a simple modification of the proposed policy in (6.12), one can get a more improved control performance or power utilization. A geometric interpretation is given in Fig. 19. As we will also see in numerical simulations this results in significant performance improvements in practice.

An example is presented next.

Example 6.1. Consider the setup of Chapter 3. In that case a specific controller is employed in order to control a linear plant (cf. Theorem 3.1). This leads to a problem where the sensors tries to regulate the

estimation error ε_k evolving according to (3.17). This is special case of the general switched linear system (6.1). For this special case the dynamics are given by $A_o = A$, $A_c = 0$, and the estimation cost is given by $Q_c = 0$, $Q_o = \tilde{P}$ (cf.(3.27)). For a channel-adapted reference policy of the form $\tilde{\mathbf{p}}$, with an average packet success \tilde{q} given by (6.6), one first solves for the reference matrix P in (6.9), which now becomes a Lyapunov equation of the form

$$P = (1 - \tilde{q}) \left(APA^T + \tilde{P} \right). \tag{6.14}$$

The resulting rollout power allocation policy (6.12) as a function of the system state, which here is the estimation error ε_k , and the channel state h_k becomes

$$\mathbf{p}_{\nu}^{roll}(\varepsilon_k, h_k) := \underset{p \in [0, p_{\max}]}{\operatorname{argmin}} \quad \nu p - q(h_k, p) \varepsilon_k^T (APA^T + \tilde{P}) \varepsilon_k.$$
(6.15)

Observe that (6.15) is of the same form as the optimal communication policy (3.32) except that the optimal unknown function $R(\varepsilon_k)$ in the latter is replaced by an easily computed quadratic form. Since the rollout policy is suboptimal the quadratic can be viewed as an approximation of the function $R(\varepsilon_k)$.

For the particular case of capacity achieving codes repeating the analysis of Section 3.4.1 we can modify (6.15) to obtain the suboptimal policy

$$\mathbf{p}_{\nu}^{roll,CA}(\varepsilon_k,h_k) := \begin{cases} 0 & \text{if } h_k \, \varepsilon_k^T (APA^T + \tilde{P}) \varepsilon_k \le \nu \, p_0 \text{ or } h_k \le \frac{p_0}{p_{max}} \\ \frac{p_0}{h_k} & \text{otherwise} \end{cases}$$
(6.16)

Again the unknown function $R(\varepsilon_k)$ in (3.36) is approximated by a quadratic that can be easily computed. This is an explicit event-triggered communication policy, where the events depend on the current channel state h_k and error ε_k . For more details see Gatsis et al. (2014c).

After the following remarks, we extend the rollout policy development to the case of state-aware scheduling of multiple control systems over a shared wireless medium.

Remark 6.1 (Rollout Algorithm Interpretation). *The motivation behind rollout policies is as follows. In a Markov Decision Process there is a state space, e.g., composed of plant and channel states x, h in our case, and an action space, e.g., power p in our case, which affects the evolution of the states. The goal is to find an action policy that minimizes a long term cost, where at each time step a function c(x, h, p) penalizes current states and actions. The optimal action at each step arises by solving a problem of the form*

$$\underset{p \in [0, p_{\max}]}{\operatorname{argmin}} c(x, h, q) + \mathbb{E} \left[V^*(x_+, h_+) \, \big| \, x, h, q \right].$$
(6.17)

for some function V^* called the optimal cost-to-go function, and where the expectation is over the state evolution. In cases where the optimal cost-to-go function V^* is not easily computed, rollout policies arise as suboptimal solutions (Bertsekas, 2005, Vol. I). One needs to find a reference policy (or a class of reference policies) for which the cost-to-go function, which is suboptimal in general, is easily computed. Say the policy is π and the corresponding function is V^{π} . Then the rollout policy is given by solving

$$\underset{p \in [0, p_{\max}]}{\operatorname{argmin}} c(x, h, q) + \mathbb{E} \left[V^{\pi}(x_+, h_+) \, \big| \, x, h, q \right].$$
(6.18)

in lieu of the optimal policy (6.17). In other words, the rollout policy finds the optimal current action assuming that the reference policy will be used from the next time step and onwards. At the next time step a rollout action is selected as well, and so on in a receding horizon fashion. The specialization of (6.18) in our setting yields (6.12).

The rollout policy (6.18) is a heuristic. Intuitively if the reference policy π is close to the optimal policy then the rollout is close to the optimal as well. Even though it is not easy to characterize how worse the rollout performs compared to the optimal policy, in practice it usually performs well as noted by (Bertsekas, 2005, Vol. I).

Rollout sensor transmit policies have also been used by Antunes and Heemels (2014) however without account of power resources but based on deterministic periodic transmission schedules as references. As a result the transmission decisions in that work are updated once every period, unlike our policies in (6.12) which allow the sensor to continuously exploit online at each time step the stochastic plant state and channel state processes.

Remark 6.2. The matrix equation (6.9) is linear in the unknown variable P hence can be readily solved. A specific solution approach, often used in the context of jump linear systems (Costa and Fragoso (1993)) or of Lyapunov equations (Laub, 2005, Ch. 13), is to vectorize the matrix equation to convert it to a vector of linear equations. That is,

$$\operatorname{vec}(P) = \left(\tilde{q} \, A_c^T \otimes A_c^T + (1 - \tilde{q}) \, A_o^T \otimes A_o^T\right) \operatorname{vec}(P) + \operatorname{vec}(\tilde{q} \, Q_c + (1 - \tilde{q}) \, Q_o) \tag{6.19}$$

where \otimes denotes the Kronecker product. Here we employed the identity $vec(ABC) = C^T \otimes A vec(B)$ (Laub, 2005, Ch. 13). The vectorized matrix P that solves (6.19) is written as

$$\operatorname{vec}(P) = \left[I - \left(\tilde{q} \, A_c^T \otimes A_c^T - (1 - \tilde{q}) \, A_o^T \otimes A_o^T\right)\right]^{-1} \operatorname{vec}(\tilde{q} \, Q_c + (1 - \tilde{q}) \, Q_o). \tag{6.20}$$

This way the cost-to-go matrix *P* is described explicitly as a function of the system dynamics, costs, and packet success of the reference policy. We also note that mean square stability of the reference policy implies the existence of such a solution (cf. Prop. 6.2). Indeed (6.8) implies that the matrix in (6.20) is invertible.

6.1.3. Numerical examples

Consider the control of a plant of the form (2.1) with dynamics

$$A = \begin{bmatrix} 1.05 & 0 \\ 0 & 0.6 \end{bmatrix}, B = \begin{bmatrix} 1 \\ 4 \end{bmatrix}.$$
 (6.21)



Figure 20: Contour plot of the indefinite value-of-transmission function example.

Suppose first we employ the controller introduced in Example 2.1 which applies a state feedback whenever a message is received and zero input otherwise, i.e., $A_c = A + BK$, $A_o = A$. Moreover let the feedback gain *K* be chosen as the optimal LQR controller with respect to quadratic state and control costs given by identity matrices $Q = I_2$, $R = I_1$. That would be the best feedback gain if communication was perfect.

Suppose we pick a reference power allocation policy $\tilde{\mathbf{p}}$ which is constant, resulting in an average packet success rate $\tilde{q} = 0.3$ (cf. (6.6)), meaning that the sensor would close the loop with the controller over the channel 30% of the time. To implement the rollout policy of the form (6.12) we solve for the reference matrix *P* by (6.9) and for the value-of-transmission matrix *M* in (6.11). We observe that the resulting matrix *M* is indefinite as it has one positive and one negative eigenvalue. In Fig. 20 we illustrate the contour plot of the quadratic $x^T M x$. The plot has a hyperbolic form, and sometimes the quadratic function takes negative values. Intuitively this negative value of transmission exists because the feedback gain of the controller is designed assuming a perfect communication link. The intermittent communication however degrades the value of sensor information to the controller. This numerical example serves to point out that communication and control policies are in general coupled leading to perhaps counter-intuitive results.

Alternatively for the same plant we consider the more sophisticated controller of Example 2.3 (or Chapter 3) which keeps a local plant state estimate \hat{x}_k and applies the control input $u_k = K\hat{x}_k$ with *K* being the standard LQR feedback gain. For the same reference policy \tilde{p} and average packet success rate $\tilde{q} = 0.3$ we solve for the matrices *P*, *M* in (6.9) and (6.11) respectively. We observe then that the matrix *M* is positive semi-definite and exactly half of its eigenvalues are zero. Actually, we can prove theoretically that for any plant combined with a controller of this form the quadratic form is exactly equal to a positive definite quadratic function of just the terms $\varepsilon_k = x_k - A\hat{x}_{k-1}$. The latter terms can be thought of as the innovation process of the plant – see also Chapter 3. This contours of this value-of transmission function are plotted in Fig. 21 which in contrast to the previous case exhibits an ellipsoid form.



Figure 21: Contour plot of the positive definite value-of-transmission function example.



Figure 22: Control performance and average power consumption of proposed rollout policies, in comparison with the reference policies.

We assume now the plant is perturbed by zero mean Gaussian noise with covariance W = 1. We implement the rollout power allocation policy of the form (6.12) for the controller with the local estimator. The resulting average control performance and power consumption after varying the parameter ν are shown in Fig. 22. We observe significant performance improvements compared to the reference policies.

These performance improvements can be understood if we look at an example of the schedule of selected powers during operation Fig. 23. The sensor often times does not transmit at all, while when it does it adapts its transmit power. Typically higher system states require higher transmit power, and better channel conditions indicate opportunities for communication with low amounts of power.



Figure 23: Schedule of selected transmit powers by the rollout algorithm.

6.2. Rollout Multiple Access Policies

In this section we discuss how the rollout power allocation procedure can be extended for the case where multiple independent plants need to communicate over the same wireless channel. In that case we design an online scheduler who decides at each time step which sensor gets access to the channel based on the plant and channel conditions experienced by all systems. We revisit the setup of Chapter 4. There are *m* plants of the form (6.1) indexed by i = 1, ..., m, i.e.,

$$x_{k+1}^{i} = \begin{cases} A_{c}^{i} x_{k}^{i} + w_{k}^{i}, & \text{if } \gamma_{k}^{i} = 1\\ A_{o}^{i} x_{k}^{i} + w_{k}^{i}, & \text{if } \gamma_{k}^{i} = 0 \end{cases}$$
(6.22)

and corresponding quadratic costs of the form (6.3) given by

$$\begin{cases} x_k^{i}{}^T Q_c^i x_k, & \text{if } \gamma_k^i = 1\\ x_k^{i}{}^T Q_o^i x_k, & \text{if } \gamma_k^i = 0 \end{cases}.$$
(6.23)

for positive semidefinite matrices $Q_o^i, Q_c^i, i = 1, ..., m$.

The channel fading state for each system *i* at time *k* is denoted by h_k^i . Channel conditions for all sensor-controller pairs are grouped in a vector h_k assumed to be i.i.d. over time with distribution ϕ_h . When system *i* is scheduled it decides upon a transmit power p_k^i and this results in a transmission success with probability given by a function $q(h_k^i, p_k^i)$ of the current channel and power. System *i* is scheduled with probability $\alpha_k^i \in [0, 1]$ at time *k*. Since at most one system can be scheduled at each time step, the scheduling vector α_k takes values in the probability simplex $\mathcal{A}_m = \{\alpha \in [0, 1]^m : \sum_{i=1}^m \alpha_i \leq 1\}$, where the inequality means that there is a probability that no plant is scheduled.
To follow the rollout procedure introduced in the previous section, we need to pick reference policies. Once again we consider reference scheduling and power allocation policies that are functions of the channel conditions but independent of the plant states. Let $\tilde{\alpha}(h)$ and $\tilde{\mathbf{p}}(h)$ be the reference scheduling and power allocation policies. Under the reference policies, each system *i* closes its loop, i.e., successfully transmits, with probability

$$\tilde{q}^{i} := \mathbb{P}(\gamma_{k}^{i} = 1) = \mathbb{E}_{h}[\tilde{\boldsymbol{\alpha}}^{i}(h)q(h^{i}, \tilde{\mathbf{p}}^{i}(h^{i}))]$$
(6.24)

at each time step k, where the expectation is with respect to the joint channel distribution. The average power consumption for each system is given by

$$J_{\text{comm}}^{i}(\tilde{\boldsymbol{\alpha}}, \tilde{\mathbf{p}}) = \limsup_{N \to \infty} \frac{1}{N} \sum_{k=0}^{N-1} \mathbb{E} \, \alpha_{k}^{i} \, p_{k}^{i} = \mathbb{E}_{h}[\tilde{\boldsymbol{\alpha}}^{i}(h) \, \tilde{\mathbf{p}}^{i}(h^{i})].$$
(6.25)

As in (6.9) in the previous section, we can compute the control cost-to-go function for each system i = 1, ..., m under this reference policy by solving for matrices P^i satisfying

$$P^{i} = \tilde{q}^{i} \left(Q_{c}^{i} + A_{c}^{i}{}^{T} P^{i} A_{c}^{i} \right) + (1 - \tilde{q}^{i}) \left(Q_{o}^{i} + A_{o}^{i}{}^{T} P^{i} A_{o}^{i} \right).$$
(6.26)

From these matrices we can deduce the relative value of transmission matrices

$$M^{i} := Q_{o}^{i} + A_{o}^{i}{}^{T}P^{i}A_{o}^{i} - Q_{c}^{i} - A_{c}^{i}{}^{T}P^{i}A_{c}^{i}.$$
(6.27)

The rollout scheduling and power allocation policy is then implemented as follows. Fix some vectors of non-negative weights $\mu \ge 0, \nu \ge 0$. Given the current channel conditions h_k^i and plant state x_k^i for all systems i = 1, ..., m, the current scheduling and power allocation are selected as the solutions to the optimization

$$\min_{\alpha \in \mathcal{A}_m} \left\{ \sum_{i=1}^m \alpha^i \min_{p_i \in [0, p_{\max}]} \left[\nu^i p^i - \mu^i q(h_k^i, p^i) x_k^{i T} M^i x_k^i \right] \right\}$$
(6.28)

The structure of this rollout policy resembles the structure of the channel-only-aware scheduling in Chapter 4 but also involves plant states. In particular each sensor selects its transmit power in a decentralized fashion based only on its corresponding local plant and channel conditions, by solving the inner minimization problem in (6.28). We point out that this power optimization problem has the same structure as in the single loop case in (6.12). The scheduler decision then is as follows. Since (6.28) involves a sum over the scheduling probability vector $\alpha \in A_m$, the scheduler will deterministically select to schedule the system currently resulting in the lowest power-control tradeoff. A similar opportunistic scheduling structure was also found in Chapter 4.

Moreover we can extend Theorem 6.1 in the current setup. By utilizing the rollout policy (6.28)

the total average control and communication costs satisfy by design

$$\sum_{i=1}^{m} \mu^{i} J_{\text{control}}^{i} + \sum_{i=1}^{m} \nu^{i} J_{\text{comm}}^{i} \le \sum_{i=1}^{m} \mu^{i} J_{\text{control}}^{i}(\tilde{\boldsymbol{\alpha}}, \tilde{\mathbf{p}}) + \sum_{i=1}^{m} \nu^{i} J_{\text{comm}}^{i}(\tilde{\boldsymbol{\alpha}}, \tilde{\mathbf{p}}).$$
(6.29)

The weights $\nu \ge 0$, $\mu \ge 0$ are parameters that can be tuned to penalize more or less the control and/or communication costs for different systems independently.

We note that in order to implement this state-aware policy, the scheduler needs to have access to the actual plant states. This might be unrealistic in the uplink case of, e.g., Fig. 10, where the access point receives sensor measurements and needs to schedule one without knowing the exact plant states. The setup becomes more practical in the downlink case where an access point has all the sensor data available and decides to communicate to one of many different receivers/controllers. We also note that rollout algorithms have been used for the problem of scheduling control systems elsewhere in the literature (Antunes et al. (2012)) however without consideration of power resources or wireless fading channels.

6.3. Rollout Random Access Policies

In this section we revisit the setup of Chapter 5 where multiple sensors decide in a decentralized manner whether to transmit or not over a shared wireless channel in order to close their loops. In this section we will design sensor policies adapted online to their local plant state as well as channel state conditions. That is in contrast to the policies designed in Chapter 5 which did not involve plant adaptation. As in the previous section there are *m* independent systems and each one has its own plant states x_k^i and dynamics of the form

$$x_{k+1}^{i} = \begin{cases} A_{c}^{i} x_{k}^{i} + w_{k}^{i}, & \text{if } \gamma_{k}^{i} = 1\\ A_{o}^{i} x_{k}^{i} + w_{k}^{i}, & \text{if } \gamma_{k}^{i} = 0 \end{cases}$$
(6.30)

for i = 1, ..., m. Here $w_k^i, k \ge 0$ denotes an i.i.d. noise process with some known distribution ϕ_{w^i} with zero mean and covariance W^i . We emphasize that knowledge of the distribution is important in the development that follows, however our results hold regardless of the type of distribution. As in the previous section the systems have their own control costs (cf. (6.23)) and i.i.d. channel states h_k^i with some distribution ϕ_{h^i} .

According to the decentralized mechanism each sensor i = 1, ..., m transmits at each time step k over the shared channel with probability $\alpha_k^i \in [0, 1]$. We suppose that a collision occurs if more than one sensors transmit at the same time slot, and a collision-free transmission is subject to packet decoding errors depending on channel fading conditions and transmit power. In this section we suppose each sensor's i transmit power is fixed to some value p^i , not a design variable. As a result, the probability that sensor i successfully closes its loop at time k equals

$$\mathbb{P}(\gamma_k^i = 1 \mid \alpha_k, h_k) = \alpha_k^i q(h_k^i, p^i) \prod_{j \neq i} \left[1 - \alpha_k^i \right].$$
(6.31)

This expression was also introduced in Chapter 5 (cf. (5.2)) and states that the probability of system *i* closing its loop at time *k* equals the probability that transmission *i* is successfully decoded at the receiver, multiplied by the probability that no other sensor $j \neq i$ is causing collisions on *i*th transmission. Similar to Chapter 5 we will assume the success function $q(h_k^i, p^i)$ is strictly increasing in the channel conditions h_k^i .

Consider now reference channel access policies for the sensors independent of the plant states, but potentially adapting to local channel states. In particular we consider policies of the form $\alpha_k^i = \tilde{\alpha}^i(h_k^i)$ for each i = 1, ..., m, where $\tilde{\alpha}^i : \mathbb{R}_+ \to [0, 1]$ is a mapping of current local channel states to the probability of accessing the shared channel. The vector of all sensor reference policies is denoted by $\tilde{\alpha}$. Under these reference policies, a sensor *j* is accessing the channel with probability that we denote as

$$\tilde{\alpha}^j := \mathbb{E}[\tilde{\alpha}^j(h_k^j)]. \tag{6.32}$$

Here the expectation is with respect to the *j*th channel distribution. Substituting these reference policies in (6.31), the probability of system *i* closing its loop becomes constant at each time step that we denote as

$$\tilde{q}^i := \mathbb{P}(\gamma_k^i = 1) = \mathbb{E}\left[\tilde{\boldsymbol{\alpha}}^i(h_k^i) q(h_k^i, p^i)\right] \prod_{j \neq i} [1 - \tilde{\alpha}^j].$$
(6.33)

Here we used the fact that channel distributions are independent among systems.

Assuming all systems are mean square stable under the chosen reference policies $\tilde{\alpha}$ (cf. Proposition 6.1), then according to Proposition 6.2 the control performance of each system *i* under this policy is given by $J_{\text{control}}^i(\tilde{\alpha}) = \text{Tr}(P^i W^i)$ where the positive semidefinite matrix P^i solves

$$P^{i} = \tilde{q}^{i} \left(Q_{c}^{i} + A_{c}^{i}^{T} P^{i} A_{c}^{i} \right) + \left(1 - \tilde{q}^{i} \right) \left(Q_{o}^{i} + A_{o}^{i}^{T} P^{i} A_{o}^{i} \right)$$
(6.34)

for the term \tilde{q}^i defined in (6.33).

Moreover we can define once again the relative value of transmitting for each system *i* under the current plant conditions as the quadratic term $x_k^i{}^T M^i x_k^i$, with the matrix M^i defined as in (6.27). In this section we also make the assumption that this value is always non-negative.

Assumption 6.1. For each system *i* the matrix $M^i = Q_o^i + A_o^i{}^T P^i A_o^i - Q_c^i - A_c^i{}^T P^i A_c^i$ is positive semidefinite.

Unlike the previous cases where the decision at each time step was centralized, here each sensor makes their own decision in a decentralized fashion based on their own local information. Unfortunately it is not obvious how to generalize the previous centralized rollout policies (Sections 6.1, 6.2) in this setting. We propose instead a different methodology for improving upon the reference policies.

According to the reference policy each sensor j accesses the channel on average with a constant

rate $\tilde{\alpha}^{j}$ given in (6.32) and is mean square stable. Intuitively this rate is low enough so that the resulting collisions on all other communication links do not deteriorate the control performance of the rest of the systems. This rate can be thought of as a *safe* access rate. Our approach in the current section is to allow the sensors to adapt to their local plant and channel states, but explicitly require that the rate at which each sensor accesses the shared channel at every time step prescribes to the reference safe access rate. We term this approach *constrained rollout*.

6.3.1. Constrained Rollout Policies

To formalize our approach we first characterize the information available to each sensor at the beginning of a time step k. Before measuring the channel state h_k^i , the sensor knows that its distribution is given by ϕ_{h^i} . Additionally before measuring the plant state x_k^i , sensor i knows by (6.30) that x_k^i has a distribution ϕ_{w^i} centered at some current mean value denoted here by \bar{x}_k^i . In other words $x_k^i = \bar{x}_k^i + w_{k-1}^i$. Similarly to (6.30) the mean \bar{x}_k^i evolves according to

$$\bar{x}_{k}^{i} = \begin{cases} A_{c}^{i} x_{k-1}^{i}, & \text{if } \gamma_{k-1}^{i} = 1\\ A_{o}^{i} x_{k-1}^{i}, & \text{if } \gamma_{k-1}^{i} = 0 \end{cases}$$
(6.35)

That is, if a transmission occurred at the last time step the sensor knows that the plant state evolves according to the closed loop mode A_c^i and expects a measurement x_k^i with mean $\bar{x}_k^i = A_c^i x_{k-1}^i$. The case of no transmission is similar. The sensor can keep track of this evolving mean value assuming it knows all past measurements and transmission successes by acknowledgments.

Our approach is to guarantee that, conditioned on the current locally available information, sensor *i* transmits with the prescribed access rate *on average*, that is

$$\mathbb{E}[\alpha_k^i \mid \bar{x}_k^i] \le \tilde{\alpha}^i. \tag{6.36}$$

At time *k* the sensor may adapt its channel access decision α_k^i to the measured plant and channel states x_k^i, h_k^i respectively. Hence we allow in general a policy of the form $\alpha_k^i = \alpha^i(x_k^i, h_k^i)$, where the set of all such transmission functions at any given mean state value \bar{x}_k is denoted by

$$\mathcal{A} = \{ \boldsymbol{\alpha}^i : \mathbb{R}^{n_i} \times \mathbb{R}_+ \to [0, 1] \}.$$
(6.37)

We propose then to select *dynamically* at each time step *k* the transmission functions α^i , i = 1, ..., m as a solution to a constrained optimization problem. The sensor would like to maximize the expected gains in control performance due to transmission, while at the same time adheres to the prescribed access rates according to (6.36). Formally this optimization problem is

$$\underset{\boldsymbol{\alpha}^{i}\in\mathcal{A}}{\text{minimize}} \quad \mathbb{E}\left[-\boldsymbol{\alpha}^{i}(x_{k}^{i},h_{k}^{i})\,q(h_{k}^{i},p^{i})\,\prod_{j\neq i}(1-\tilde{\alpha}_{j})\,x_{k}^{i}{}^{T}M^{i}x_{k}^{i}\mid\bar{x}_{k}^{i}\right] \tag{6.38}$$

subject to
$$\mathbb{E}[\boldsymbol{\alpha}^{i}(\boldsymbol{x}_{k}^{i},h_{k}^{i}) \mid \bar{\boldsymbol{x}}_{k}^{i}] \leq \tilde{\boldsymbol{\alpha}}^{i}.$$
 (6.39)

In this problem the expectations are with respect to the current channel distribution of h_k^i , as well as the plant state distribution x_k^i which conditioned on the current mean value \bar{x}_k^i just depends on the current noise w_{k-1}^i . We denote the optimal solutions, i.e., the optimal transmission functions as a^{*i} for each system *i*.

At each time step the sensor selects the current transmit function as a solution to the constrained optimization problem (6.38)-(6.39) where both the objective and the constraint are expressed *in expectation over the current plant and channel conditions* x_k , h_k to be measured. The expected objective to be minimized in (6.38) entails the quadratic form $x_k^{i}{}^T M^i x_k^i$ of the local plant state x_k^i . As explained in Section 6.1 this represents the relative gains in control performance if a successful transmission occurs. The objective also involves the probability of successful transmission $q(h_k^i, p^i)$. So the overall objective indeed represents the expected gains in control performance of the system.

For the proposed policy we can establish the following result which characterizes the control performance using the proposed constrained rollout policies.

Theorem 6.2 (Rollout Random Access Policy Performance). Let Assumption 6.1 hold. Suppose every sensor uses the policy α^{*i} optimizing (6.38)-(6.39) at each time step. Then the control cost of each system i = 1, ..., m satisfies

$$J_{control}^{i}(\boldsymbol{\alpha}^{*}) \leq J_{control}^{i}(\tilde{\boldsymbol{\alpha}}).$$
(6.40)

where $J_{\text{control}}^{i}(\tilde{\alpha})$ is the control cost using the reference policies.

This result is important as it demonstrates that despite the collisions arising in the shared medium between transmissions, each sensor can adapt to their own local plant state measurements and channel conditions and guarantee an improved control performance (compared to the reference). As a side remark we note that by construction the sensors transmit at the same rate as the reference policies. This is because (6.39) is enforced at each time step, hence it holds in the long run too. The average rate of collisions remains the same as for the reference policy as well. What changes is that transmissions occur when it is opportunistically more important for the system, hence successful transmissions also occur when it is important.

To find the transmission function α^{*i} at each time step given its current mean state $\bar{x}_k^i \in \mathbb{R}^{n_i}$, the sensor needs to solve problem (6.38)-(6.39). This is an optimization problem over the space of transmission functions, i.e., an infinite-dimensional optimization problem. However as we show in the following result it enjoys simple threshold-based optimal solutions.

Proposition 6.3. Consider the transmission function optimization problem in (6.38)-(6.39) given any fixed $\bar{x}_k^i \in \mathbb{R}^{n_i}$. Suppose the distributions ϕ_{h^i} of channel states $\{h_k^i, k \ge 0\}$ and ϕ_{w^i} of plant disturbances $\{w_k^i, k \ge 0\}$ of system *i* are absolutely continuous, *i.e.*, have probability density functions. Then there exists a non-negative constant $v^i \ge 0$, depending on \bar{x}_k^i , such that the function

$$\boldsymbol{\alpha}^{*i}(x_k^i, h_k^i) = \begin{cases} 1, & \text{if } q(h_k^i, p^i) \prod_{j \neq i} (1 - \tilde{\alpha}_j) x_k^{i^T} M^i x_k^i \ge \nu^i \\ 0 & \text{otherwise.} \end{cases}$$
(6.41)

is an optimal solution.

The above proposition shows first that the proposed decision is not randomized but deterministic. Given the current plant and channel conditions the sensor will either transmit or not. Furthermore it is of a threshold form. The sensor transmits if the product between the current channel success $q(h_k^i, p^i) \prod (1 - \tilde{\alpha}_j)$ and value of transmission $x_k^{i} M^i x_k^i$ exceeds some threshold. Here the packet success assumes that all other sensors $j \neq i$ will adhere to their reference access rates $\tilde{\alpha}^j$.

We note that threshold policies with respect to the plant state frequently appear in the eventbased control framework of, e.g., Xu and Hespanha (2004); Molin and Hirche (2009); Heemels et al. (2012). The threshold value here however is not a free parameter, but it depends on the current mean value \bar{x}_k^i which dynamically varies at each time step (cf. (6.35)). As a result the transmission decision α_k^i is not just a time-invariant function of states x_k^i , h_k^i but it also depends on the mean value \bar{x}_k^i . This feature distinguishes our policies from other event-based control policies and is by construction added to ensure that the sensor adheres to the prescribed safe access rate $\tilde{\alpha}^i$.

Since Prop. 6.3 simplifies the search for general transmission functions in Theorem 6.2 to the search for threshold functions, we describe next a computationally efficient procedure to find the appropriate threshold. We note that in order to implement the proposed policy, a transmission function solving problem (6.38)-(6.39) is needed for any possible mean value $\bar{x}_k^i \in \mathbb{R}^{n_i}$. This requires the solution of an infinite number of such optimization problems. In practice, as well as in the numerical simulations that follow, the space of mean values $\bar{x}_k^i \in \mathbb{R}^n$ can be discretized and we can solve instead a large number of optimization problems at the discrete points.

6.3.2. Computing Transmission Functions

For notational simplicity we now drop the time index k. From the proof of Prop. 6.3 we see that the threshold v^i , as a function of the current mean \bar{x}^i , corresponds to the optimal Lagrange dual variable of problem (6.38)-(6.39). We briefly describe a dual subgradient algorithm to find this optimal dual point. A subgradient direction for the dual problem is typically given by the constraint slack of the primal problem evaluated at a primal Lagrangian minimizer (cf. (Bertsekas et al., 2003, Ch.8)). More precisely, given some dual variable $v_t \ge 0$ at iteration t, a corresponding primal solution is given by substituting v_t in place of v^i in (6.41).

The constraint slack of this solution with respect to (6.39) is given by the difference

$$s_t := \mathbb{P}\Big[q(h^i, p^i) \prod_{j \neq i} (1 - \tilde{\alpha}_j) x^{i^T} M^i x^i \ge v^t \mid \bar{x}^i\Big] - \tilde{\alpha}^i.$$
(6.42)

Hence a dual subgradient ascent algorithm to compute the point v_{t+1} for the next iteration is given by

$$\nu_{t+1} = \max\left\{0, \, \nu_t + \varepsilon_t \, s_t\right\} \tag{6.43}$$

where $\varepsilon_t \ge 0$ is a stepsize, and the maximum is taken so that the new dual variable is projected

to the non-negatives. Iterating (6.43), the dual variable v_t converges to the optimal v^i – see, e.g., (Bertsekas et al., 2003, Ch. 8.2).

6.3.3. Numerical Simulations

We consider a random access architecture involving the control of two (m = 2) scalar systems of the form (6.30). The first system is described by

$$x_{k+1}^{1} = \begin{cases} 0.6 \, x_{k}^{1} + w_{k}^{1}, & \text{if } \gamma_{k}^{1} = 1\\ 1.1 \, x_{k}^{1} + w_{k}^{1}, & \text{if } \gamma_{k}^{1} = 0 \end{cases}$$
(6.44)

i.e., its open loop is unstable. The second system is assumed to have open loop integrator dynamics and a controller resetting the state to zero, i.e.,

$$x_{k+1}^{2} = \begin{cases} 0 x_{k}^{2} + w_{k}^{2}, & \text{if } \gamma_{k}^{2} = 1\\ 1 x_{k}^{2} + w_{k}^{2}, & \text{if } \gamma_{k}^{2} = 0 \end{cases}$$
(6.45)

Due to plant noise, taken here to be standard Gaussian, both system states will grow unbounded unless communication is performed often enough to close the loops. Moreover we assume the same quadratic state costs for both systems $Q_o^i = Q_c^1 = 1$, i = 1, 2 (cf. (6.23)). We also assume the channel fading states for the two systems are identically distributed, $h_k^i \sim \exp(1)$, $i = 1, 2, k \ge 0$, while the probability of success is modeled as in Fig. 2.

We begin our approach by considering reference policies. For simplicity we assume both sensors access the shared wireless channel with a constant probability under the reference policies. In particular suppose system 1 which is the unstable one accesses the channel with probability $\tilde{\alpha}^1 = 0.40$, while system 2 which is the marginally stable one with probability $\tilde{\alpha}^2 = 0.15$. Despite the collisions arising when both sensors transmit simultaneously the two systems remain stable. Their control performances can be evaluated either theoretically following Section 6.3, i.e., by $J_{\text{control}}^i(\tilde{\alpha}) = \text{Tr}(P^i W^i)$ where P^i satisfies (6.34), or in simulations. In Fig. 24 we plot the empirical control cost from simulations

$$J_{\text{control}}^{1}(\tilde{\boldsymbol{\alpha}}) = \lim_{N \to \infty} \frac{1}{N} \sum_{k=0}^{N-1} (x_{k}^{1})^{2} \approx 32,$$

$$J_{\text{control}}^{2}(\tilde{\boldsymbol{\alpha}}) = \lim_{N \to \infty} \frac{1}{N} \sum_{k=0}^{N-1} (x_{k}^{2})^{2} \approx 13.$$
 (6.46)

In other words using the reference policies the more unstable system experiences worse control performance.

We now employ the proposed state-aware policies described in the previous section. In particular each sensor i = 1, 2 keeps track of its own plant state statistics, and applies the transmission rule α^{*i} solving problem (6.38)-(6.39). This is a threshold-based policy as shown in (6.41) which can be computed following Section 6.3.2. Sensor *i* has to solve this problem for all possible current mean



Figure 24: Simulation of reference random access policies and proposed state-aware random access policies. By opportunistically adapting to system states the proposed policies exhibit a significantly improved control performance.

values $\bar{x}_k^i \in \mathbb{R}$ of the plant state. In simulations we discretize the space $\bar{x}_k^i \in \mathbb{R}$, find the policies at the discrete points, and during run-time the sensor selects the policy of the closest discrete point.

We simulate the random access wireless architecture using the proposed state-aware policies and in Fig. 24 we present the average empirical control cost, i.e.,

$$J_{\text{control}}^{1}(\boldsymbol{\alpha}^{*}) = \lim_{N \to \infty} \frac{1}{N} \sum_{k=0}^{N-1} (x_{k}^{1})^{2} \approx 6.3,$$

$$J_{\text{control}}^{2}(\boldsymbol{\alpha}^{*}) = \lim_{N \to \infty} \frac{1}{N} \sum_{k=0}^{N-1} (x_{k}^{2})^{2} \approx 6.6$$
(6.47)

during simulation. As verified by Theorem 6.2 the performance using the proposed state-aware policies is improved compared to the reference. The theoretical result claims just an improvement, however the simulations illustrate significant improvements at the order of 80% for plant 1 and 50% for system 2. The reason for these improvements is that the proposed policies adapt online in an opportunistic fashion to both plant and channel conditions. This way a sensor accesses the shared channel only when favorable channel states or necessary control requirements appear. We note that it seems that the more unstable system has higher performance improvements. We also note that control performances are improved even though collisions still occur, in fact at the same



Figure 25: Reference random access. Green stars indicate successful transmissions for each system. Red circles indicate collisions between the two systems. Successful transmissions happen in a random fashion independent of the system states.



Figure 26: Constrained rollout random access. Green stars indicate successful transmissions for each system. Red circles indicate collisions between the two systems. Successful transmissions typically occur when the systems states are large, meaning that control performance is at stake.

rate as in the reference policies.

The large performance improvements can also be understood by looking at sample system traces (Fig. 25, Fig. 26. In the reference policy successful transmissions happen in a random fashion, independent of system states. In the constrained rollout policies successful transmissions are be design correlated with times when the system states are large, i.e., control regulation is needed.

Finally, we would also like to point out that the plant adaptation of the proposed rollout policy is so significant that in many cases it can outperform non-state-aware centralized policies. In other words it overcomes the performance losses due to collisions. For example, consider the two systems given above and consider a centralized scheduler that gives access to system 1 with probability 0.4 at each time step, and gives access to system 2 with probability 0.15 at each time step. These are the same rates as the reference random access rates, but in a centralized scheduling setup. We can compute the control performances for systems 1 and 2 to be respectively 13.7,8

which are worse than the control performances achieved by the proposed state-aware random access policy.

6.4. Proofs

6.4.1. Notation used for the proofs of Proposition 6.2 and Theorem 6.1

We introduce some notation that will be used within the proofs of this chapter. When the current plant and channel states take some values $x \in \mathbb{R}^n$, $h \in \mathbb{R}_+$ and the sensor takes a decision $p \in [0, p_{\text{max}}]$, let $\gamma \in \{0, 1\}$ denote the random success of the current transmission modeled by (6.2). Also let $x_+ \in \mathbb{R}^n$, $h_+ \in \mathbb{R}_+$ denote the random value of the plant and channel states at the next time step. The distribution of the former depends on γ and the current plant disturbance $w \sim \phi_w$ according to $x_+ = \gamma A_c x + (1 - \gamma)A_o x + w$ as follows from (6.1). The next channel state is independent by assumption. We denote the integration with respect to the distribution of γ and x_+ , h_+ given the values of x, h, p as $\mathbb{E}[. | x, h, p]$. This expectation depends just on the control and communication system model, not on the employed policy.

Moreover consider the function

$$V(x,h) := x^T P(h)x \tag{6.48}$$

where P(h) is a symmetric positive semidefinite matrix defined as

$$P(h) := q(h, \tilde{\mathbf{p}}(h)) (Q_c + A_c^T P A_c) + (1 - q(h, \tilde{\mathbf{p}}(h))) (Q_o + A_o^T P A_o),$$
(6.49)

where *P* is defined in (6.9). It is also easy to see from (6.49) that the expected value of the matrix *P*(*h*) with respect to the channel distribution ϕ_h equals the matrix *P*, i.e.,

$$\mathbb{E}_h P(h) = P. \tag{6.50}$$

Define also for any $x \in \mathbb{R}^n$, $h \in \mathbb{R}_+$ and $p \in [0, p_{max}]$ the function

$$F(x,h,p) := \mathbb{E}[x^T (\gamma Q_c + (1-\gamma)Q_o) x + V(x_+,h_+) | x,h,p] - V(x,h) - Tr(PW)$$
(6.51)

Substituting $x_+ = \gamma A_c x + (1 - \gamma)A_o x + w$ in $V(x_+, h_+)$ at the right hand side and by expanding the expectation term we have

$$\mathbb{E}[x^{T}(\gamma Q_{c} + (1 - \gamma)Q_{o})x + V(x_{+}, h_{+}) | x, h, p] = q(h, p)x^{T}(Q_{c} + A_{c}^{T}PA_{c})x + (1 - q(h, p))x^{T}(Q_{o} + A_{o}^{T}PA_{o})x + \operatorname{Tr}(PW)$$
(6.52)

where the latter term Tr(PW) is due to the variance of the system disturbance and (6.50). Substituting (6.52) and (6.48)-(6.49) in the definition of the function F(x, h, p) in (6.51) we derive an equivalent expression for the function as

$$F(x,h,p) = -(q(h,p) - q(h,\tilde{\mathbf{p}}(h))) x^{T} M x$$
(6.53)

All the above expressions will be used for the proofs of our results. In particular we will make use of both expressions (6.51) and (6.53). We also note for future reference that substituting the reference policy function $\tilde{\mathbf{p}}(h)$ in (6.53) we directly get

$$F(x,h,\tilde{\mathbf{p}}(h)) = 0 \tag{6.54}$$

6.4.2. Proof of Proposition 6.2

We are now ready to prove the result of this Proposition. First we will show that there is a unique symmetric positive semidefinite matrix solution P of (6.9). To do this we can describe the solution P as the limit of the matrix recursion

$$P_{k+1} = \tilde{q} \left(Q_c + A_c^T P_k A_c \right) + (1 - \tilde{q}) (Q_o + A_o^T P_k A_o),$$
(6.55)

for some initial matrix condition P_0 . This recursion is convergent. To see this, we can perform the vectorization operation (see also Remark 6.2) to rewrite the recursion as

$$\operatorname{vec}(P_{k+1}) = \left(\tilde{q} \, A_c^T \otimes A_c^T - (1 - \tilde{q}) \, A_o^T \otimes A_o^T\right) \operatorname{vec}(P_k) + \operatorname{vec}(\tilde{q} \, Q_c + (1 - \tilde{q}) \, Q_o). \tag{6.56}$$

By assumption the employed reference policy $\tilde{\mathbf{p}}$ leads to mean square stability, which implies (6.8), i.e., that the matrix in (6.56) is asymptotically stable. Hence vec(P_k) converges to a unique limit point, and the matrix P_k converges to a unique matrix P solving equation (6.9).

To show that *P* is also a symmetric positive semidefinite matrix, let us pick the initial condition $P_0 = \tilde{q} Q_c + (1 - \tilde{q})Q_o$. It can be easily argued by induction that $P_{k+1} \succeq P_k \succeq 0$ for all $k \ge 0$. Hence the limit also satisfies $P = \lim_{k\to\infty} P_k \succeq 0$.

We are now ready to prove the second part of the proposition. That is, if x_k , h_k denote the random variables describing the plant and channel states of the system at time k when using the reference policy $p_k = \tilde{\mathbf{p}}(h_k)$, we will show that

$$\lim_{N \to \infty} \frac{1}{N} \sum_{k=0}^{N-1} \mathbb{E}[x_k^T \left(\gamma_k Q_c + (1 - \gamma_k) Q_o\right) x_k] = \operatorname{Tr}(PW).$$
(6.57)

If x_k , h_k are the random variables denoting the plant and channel states of the system at time k using the reference policy $p_k = \tilde{\mathbf{p}}(h_k)$, then by (6.54) we have that

$$F(x_k, h_k, p_k) = F(x_k, h_k, \tilde{\mathbf{p}}(h_k)) = 0$$
(6.58)

holds almost surely for all $k \ge 0$. Since the function *F* is equal to zero, by its definition (6.51) we conclude that

$$\mathbb{E}[x_k^T(\gamma_k Q_c + (1 - \gamma_k)Q_o)x_k + V(x_{k+1}, h_{k+1}) | x_k, h_k, p_k] = V(x_k, h_k) + \operatorname{Tr}(PW)$$
(6.59)

holds almost surely for all $k \ge 0$.

Let us take the expectation of (6.59) with respect to the distribution of x_k , h_k to conclude that

$$\mathbb{E}[x_k^T(\gamma_k Q_c + (1 - \gamma_k) Q_o) x_k + V(x_{k+1}, h_{k+1})] = \mathbb{E}[V(x_k, h_k)] + \operatorname{Tr}(PW)$$
(6.60)

holds for all $k \ge 0$. Summing up (6.60) for k = 0, ..., N - 1, removing the telescoping terms $\sum_{k=0}^{N-1} \mathbb{E}V(x_k, h_k)$, and dividing by *N* we get that

$$\frac{1}{N}\sum_{k=0}^{N-1}\mathbb{E}[x_k^T(\gamma_k Q_c + (1-\gamma_k)Q_o)x_k] + \frac{1}{N}\mathbb{E}[V(x_N, h_N)] = \frac{1}{N}\mathbb{E}V(x_0, h_0) + \operatorname{Tr}(PW)$$
(6.61)

Consider now the terms $\mathbb{E}V(x_k, h_k) = \mathbb{E} x_k^T P(h_k) x_k$ for any time $k \ge 0$. Since the channel state h_k at time k is by the model independent of the plant state x_k and distributed by ϕ_h , we have that

$$\mathbb{E}V(x_k, h_k) = \operatorname{Tr}(\mathbb{E}P(h_k) \mathbb{E} x_k x_k^T) = \operatorname{Tr}(P \mathbb{E} x_k x_k^T) < \infty$$
(6.62)

Here in the second equality we used (6.50) and in the inequality we used the mean square stability of the system by assumption ($\limsup_{k\to\infty} \mathbb{E}x_k x_k^T < \infty$). Hence the terms $\mathbb{E}V(x_0, h_0)$, $\mathbb{E}[V(x_N, h_N)]$ in (6.61) are uniformly bounded over *N*. Hence taking the limit in (6.61) as $N \to \infty$ verifies (6.57) and completes the proof.

6.4.3. Proof of Theorem 6.1

We use the notation introduced in the proof of Prop. 6.2. For any constant $\nu \ge 0$, and any plant and channel state value $x \in \mathbb{R}^n$, $h \in \mathbb{R}_+$ consider the minimization of the expression $F(x, h, p) + \nu p$ over power choices $p \in [0, p_{\text{max}}]$. Since the reference choice $\tilde{\mathbf{p}}(h)$ is one feasible solution we have

$$\min_{p \in [0, p_{\max}]} F(x, h, p) + \nu p \leq F(x, h, \tilde{\mathbf{p}}(h)) + \nu \tilde{\mathbf{p}}(h) = \nu \tilde{\mathbf{p}}(h).$$
(6.63)

where the last equality follows from (6.54). We then analyze the left hand side of the above inequality (6.63). By the expression for the function F given in (6.53) we have that

$$\min_{p \in [0, p_{\max}]} F(x, h, p) + \nu p = \min_{p \in [0, p_{\max}]} \nu p - (q(h, p) - q(h, \tilde{\mathbf{p}}(h))) x^T M x$$
(6.64)

This minimization is exactly the same as the one appearing in (6.12), hence the optimal solution to this minimization problem is $\mathbf{p}_{\nu}^{\text{roll}}(x, h)$. As a result (6.63) implies the inequality

$$F(x,h,\mathbf{p}_{\nu}^{\text{roll}}(x,h)) + \nu \, \mathbf{p}_{\nu}^{\text{roll}}(x,h) \le \nu \, \tilde{\mathbf{p}}(h).$$
(6.65)

We will use this inequality to prove the Theorem.

Suppose x_k , h_k are the random variables representing the plant and channel states at the *k*th time step after employing the policy $p_k = \mathbf{p}_{\nu}^{\text{roll}}(x_k, h_k)$ given in (6.12). By (6.65) we get then that

$$F(x_k, h_k, p_k) + \nu p_k \le \nu \,\tilde{\mathbf{p}}(h_k). \tag{6.66}$$

holds almost surely for all $k \ge 0$. Substituting then the definition of the function *F* by (6.51) in (6.66) we have that

$$\mathbb{E}[x_k^T \left(\gamma_k Q_c + (1 - \gamma_k) Q_o\right) x_k + V(x_{k+1}, h_{k+1}) \mid x_k, h_k, p_k] + \nu p_k$$

$$\leq V(x_k, h_k) + Tr(PW) + \nu \tilde{\mathbf{p}}(h_k)$$
(6.67)

holds almost surely for all $k \ge 0$. By taking the expectation of the above inequality with respect to the processes x_k , h_k , p_k we have that

$$\mathbb{E}[x_k^T \left(\gamma_k Q_c + (1 - \gamma_k) Q_o\right) x_k + \nu p_k + V(x_{k+1}, h_{k+1})] \\ \leq \mathbb{E}V(x_k, h_k) + Tr(PW) + \nu \mathbb{E}\tilde{\mathbf{p}}(h_k)$$
(6.68)

holds for all $k \ge 0$. Then summing up all the above inequalities (6.68) for k = 0, ..., N - 1, removing the telescoping terms $\sum_{k=0}^{N-1} \mathbb{E}V(x_k, h_k)$, and dividing by N we get

$$\frac{1}{N} \sum_{k=0}^{N-1} \mathbb{E}[x_k^T (\gamma_k Q_c + (1 - \gamma_k) Q_o) x_k + \nu p_k] + \frac{1}{N} \mathbb{E}[V(x_N, h_N)] \leq \frac{1}{N} \mathbb{E}V(x_0, h_0) + \operatorname{Tr}(PW) + \nu \frac{1}{N} \sum_{k=0}^{N-1} \mathbb{E}\tilde{\mathbf{p}}(h_k)$$
(6.69)

Now since the function *V* is non-negative (cf.(6.49)), the term $\mathbb{E}[V(x_N, h_N)]$ is also non-negative and can be omitted from (6.69). Moreover $\mathbb{E}V(x_0, h_0)$ is bounded as the initial plant state x_0 has finite variance by assumption. Hence by taking the limit in (6.69) as $N \to \infty$ we conclude that

$$\limsup_{N \to \infty} \frac{1}{N} \sum_{k=0}^{N-1} \mathbb{E}[x_k^T \left(\gamma_k Q_c + (1 - \gamma_k) Q_o\right) x_k + \nu p_k] \le \operatorname{Tr}(P W) + \nu \limsup_{N \to \infty} \frac{1}{N} \sum_{k=0}^{N-1} \mathbb{E}\tilde{\mathbf{p}}(h_k)$$
(6.70)

As established in Proposition 6.2 the value Tr(PW) is equal to the average control performance $J_{\text{control}}(\tilde{\mathbf{p}})$ of the reference policy $\tilde{\mathbf{p}}$. Also the average sum at the right hand side of (6.70) is equal to the average power consumption of the reference policy $\tilde{\mathbf{p}}$ given in (6.7). Hence (6.70) verifies (6.13) and completes the proof.

For simplicity the arguments to follow prove (6.40) for some fixed system *i*. A symmetric argument can be performed for each system i = 1, ..., m. We will use notation similar to the one introduced in the proof of Proposition 6.2, however the present setup is more complex as there are many systems.

We will refer to the current mean value for each system i by $\bar{x}^i \in \mathbb{R}^{n_i}$ as introduced in (6.35). Then the current plant state $x^i \in \mathbb{R}^{n_i}$ at system i depends on the current mean \bar{x}^i and the current random plant disturbance $w^i \in \mathbb{R}^{n_i}$ with distribution ϕ_{w^i} , according to $\bar{x}^i = x^i + w^i$. When the current channel states for each system i take some values $h^i \in \mathbb{R}_+$ and the sensors access the channel with probabilities $\alpha^i \in [0, 1]$, let $\gamma^i \in \{0, 1\}$ denote the random transmission success on link i modeled by (6.31). Also let $x^i_+ \in \mathbb{R}^{n_i}, h^i_+ \in \mathbb{R}_+$ denote the random value of the plant and channel states for system i at the next time step. The distribution of the former depends on γ^i and the next plant disturbance $w^i_+ \sim \phi_{w^i}$ according to (6.30). The next channel state h^i_+ is independent of all other variables by assumption. In a vector form we will denote the above variables as $\bar{x} \in \mathbb{R}^n, x \in \mathbb{R}^n, h \in \mathbb{R}^m_+, \alpha \in [0,1]^m, \gamma \in 0, 1^m, x_+ \in \mathbb{R}^n, h_+\mathbb{R}^m_+$. We denote the integration with respect to the distributions of these variables given the values of x, h, α as $\mathbb{E}[. | x, h, \alpha]$. This expectation depends just on the control and communication system model, not on the employed policy. The expectation of any function of just the current plant and channel states x, h given the current mean values \bar{x} will be denoted by $\mathbb{E}[. | \bar{x}]$.

Moreover, similar to the proof of Proposition 6.2, define the functions

$$V^{i}(x^{i}, h^{i}) := x^{i^{T}} P^{i}(h^{i}) x^{i}$$
(6.71)

for i = 1, ..., *m*, where $P^i(h^i)$ are symmetric positive semidefinite matrices defined as ⁶

$$P^{i}(h^{i}) := \tilde{\boldsymbol{\alpha}}^{i}(h^{i}) q(h^{i}) \prod_{j \neq i} [1 - \tilde{\boldsymbol{\alpha}}^{j}(h^{j})] (Q_{c}^{i} + A_{c}^{i}{}^{T}P^{i}A_{c}^{i}) + \left(1 - \tilde{\boldsymbol{\alpha}}^{i}(h^{i}) q(h^{i}) \prod_{j \neq i} [1 - \tilde{\boldsymbol{\alpha}}^{j}(h^{j})]\right) (Q_{o}^{i} + A_{o}^{i}{}^{T}P^{i}A_{o}^{i}),$$
(6.72)

where P^i is defined in (6.34). It is also easy to see from (6.72) and the definition of P^i in (6.34) that the expected value of the matrix $P^i(h^i)$ with respect to the channel distribution ϕ_h equals the matrix P^i , i.e., $\mathbb{E}_h P^i(h^i) = P^i$.

Similar to the proof of Proposition 6.2, define also for any $x^i \in \mathbb{R}^{n_i}$, $h^i \in \mathbb{R}_+$ and $\alpha \in [0, 1]^m$ the function

$$F(x^{i},h^{i},\alpha) := \mathbb{E}[x^{i^{T}}(\gamma^{i}Q_{c}^{i} + (1-\gamma^{i})Q_{o}^{i})x^{i} + V^{i}(x_{+}^{i},h_{+}^{i}) | x^{i},h^{i},\alpha] - V^{i}(x^{i},h^{i}) - Tr(P^{i}W^{i})$$
(6.73)

Note that the function depends on the whole vector α , i.e., the transmissions of all sensors. Fol-

⁶As the powers p^i are fixed for all systems, for brevity we will denote the function $q(h_k^i, p^i)$ as $q(h_k^i)$.

lowing arguments similar to the ones leading from (6.51) to (6.53) we can equivalently write the function F as

$$F(x^{i},h^{i},\alpha) = -\left[\alpha^{i} \prod_{j \neq i} [1-\alpha^{j}] - \tilde{\alpha}^{i}(h^{i}) \prod_{j \neq i} [1-\tilde{\alpha}^{j}(h^{j})]\right] q(h^{i}) x^{i^{T}} M^{i} x^{i}$$
(6.74)

Now substituting α^i in (6.74) from any local transmission function $\alpha^i(x^i, h^i)$ for each system i = 1, ..., m we get an expression $F(x^i, h^i, \alpha(x, h))$ which depends only on the current plant and channel conditions x, h of all systems, where with a slight abuse of notation here we use

$$\boldsymbol{\alpha}(x,h) := \begin{bmatrix} \vdots \\ \boldsymbol{\alpha}^{i}(x^{i},h^{i}) \\ \vdots \end{bmatrix}.$$
(6.75)

We also note for future reference that substituting the reference transmission functions $\tilde{\alpha}^i(h^i)$ in (6.74) we directly get

$$F(x^{i}, h^{i}, \tilde{\boldsymbol{\alpha}}(h)) = 0 \tag{6.76}$$

for any values of $x \in \mathbb{R}^n$, $h \in \mathbb{R}^m_+$.

The proof of the theorem relies on the following technical lemma.

Lemma 6.1. Let Assumption 6.1 hold. Then for any value of $\bar{x} \in \mathbb{R}^n$ it holds that

$$\mathbb{E}\left[F\left(x^{i},h^{i},\boldsymbol{\alpha}^{*}(x,h)\right)\mid\bar{x}\right]\leq0$$
(6.77)

where α^* are the optimal solutions of problems (6.38)-(6.39) for all systems i = 1, ..., m.

Before proceeding to the proof of the lemma let us show how it can establish the desired result. Suppose x_k , h_k are the random variables representing the plant and channel states of all systems at the *k*th time step after each system j = 1, ..., m employs the proposed policy $\alpha_k^j = \alpha^{*j}(x_k^j, h_k^j)$ solving (6.38)-(6.39). By (6.77) we get then that

$$\mathbb{E}\left[F\left(x_{k}^{i},h_{k}^{i},\alpha_{k}\right)\mid\bar{x}_{k}\right]\leq0.$$
(6.78)

holds almost surely for all $k \ge 0$. Substituting then the definition of the function *F* by (6.73) in (6.78) we have that

$$\mathbb{E}[x_k^{iT}(\gamma_k^i Q_c^i + (1 - \gamma_k^i) Q_o^i) x_k^i + V^i(x_{k+1}^i, h_{k+1}^i) - V^i(x_k^i, h_k^i) \mid \bar{x}_k] \le Tr(P^i W^i)$$
(6.79)
(6.80)

holds almost surely for all $k \ge 0$. By taking the expectation of the above inequality with respect

to the processes x_k , h_k , α_k we have that

$$\mathbb{E}[x_k^{i^T}(\gamma_k^i Q_c^i + (1 - \gamma_k^i) Q_o^i) x_k^i] + \mathbb{E}V^i(x_{k+1}^i, h_{k+1}^i) - \mathbb{E}V^i(x_k^i, h_k^i) \le Tr(P^i W^i)$$
(6.81)

holds for all $k \ge 0$. Then summing up all the above inequalities (6.81) for k = 0, ..., N - 1, removing the telescoping terms $\sum_{k=0}^{N-1} \mathbb{E}V^i(x_k^i, h_k^i)$, and dividing by N we get

$$\frac{1}{N}\sum_{k=0}^{N-1}\mathbb{E}[x_k^{i^T}(\gamma_k^i Q_c^i + (1-\gamma_k^i) Q_o^i) x_k^i] + \frac{1}{N}\mathbb{E}[V^i(x_N^i, h_N^i)] \le \frac{1}{N}\mathbb{E}V(x_0^i, h_0^i) + \operatorname{Tr}(P^i W^i)$$
(6.82)

Now since the function *V* is non-negative (cf.(6.72)), the term $\mathbb{E}[V^i(x_N^i, h_N^i)]$ is also non-negative and can be omitted from (6.82). Moreover $\mathbb{E}V(x_0^i, h_0^i)$ is bounded as the initial plant states have finite variance by assumption. Hence by taking the limit in (6.82) as $N \to \infty$ we conclude that

$$\limsup_{N \to \infty} \frac{1}{N} \sum_{k=0}^{N-1} \mathbb{E}[x_k^{i^T} (\gamma_k^i Q_c^i + (1 - \gamma_k^i) Q_o^i) x_k^i] \le \operatorname{Tr}(P^i W^i)$$
(6.83)

As established in Proposition 6.2 the value $\text{Tr}(P^i W^i)$ is equal to the average control performance $J^i_{\text{control}}(\tilde{\alpha})$ of the reference policy $\tilde{\alpha}$. Hence (6.83) verifies (6.40) and completes the proof.

Proof of Lemma 6.1. First for notational convenience for any transmission functions $\alpha^1, \ldots, \alpha^m$ let us define the function

$$\mathbf{F}(\boldsymbol{\alpha}^{i},\boldsymbol{\alpha}^{-i}) := \mathbb{E}\left[F\left(x^{i},h^{i},\boldsymbol{\alpha}(x,h)\right) \mid \bar{x}\right]$$
(6.84)

where α^{-i} is the set of all transmission functions for systems $j \neq i$. To prove the lemma we need to show that

$$\mathbf{F}(\boldsymbol{\alpha}^{*i}, \boldsymbol{\alpha}^{*-i}) \le 0. \tag{6.85}$$

Let us substitute the expression for the function F given in (6.74) in (6.84) to rewrite

$$\mathbf{F}(\boldsymbol{\alpha}^{i},\boldsymbol{\alpha}^{-i}) = \mathbb{E}\left[-\left[\boldsymbol{\alpha}^{i}(x^{i},h^{i})\prod_{j\neq i}\left[1-\boldsymbol{\alpha}^{j}(x^{j},h^{j})\right]-\tilde{\boldsymbol{\alpha}}^{i}(h^{i})\prod_{j\neq i}\left[1-\tilde{\boldsymbol{\alpha}}^{j}(h^{j})\right]\right]q(h^{i})x^{i^{T}}M^{i}x^{i} \mid \bar{x}\right]$$
(6.86)

Now note that the plant and channel states for all systems $j \neq i$ conditioned on the current mean values \bar{x} become independent among systems, because transmission functions are localized on each system. As a result we have

$$\mathbf{F}(\boldsymbol{\alpha}^{i},\boldsymbol{\alpha}^{-i}) = -\mathbb{E}\left[\boldsymbol{\alpha}^{i}(x^{i},h^{i}) q(h^{i}) x^{i^{T}} M^{i} x^{i} \mid \bar{x}^{i}\right] \prod_{j \neq i} \mathbb{E}\left[1 - \boldsymbol{\alpha}^{j}(x^{j},h^{j}) \mid \bar{x}^{j}\right] \\ + \mathbb{E}\left[\tilde{\boldsymbol{\alpha}}^{i}(h^{i}) q(h^{i}) x^{i^{T}} M^{i} x^{i} \mid \bar{x}^{i}\right] \prod_{j \neq i} \mathbb{E}\left[1 - \tilde{\boldsymbol{\alpha}}^{j}(h^{j}) \mid \bar{x}^{j}\right]\right]$$
(6.87)

Then note that by construction, the policies α^{*j} for all systems $j \neq i$ satisfy the constraint (6.39),

i.e.,

$$\mathbb{E}[\boldsymbol{\alpha}^{*j}(\boldsymbol{x}^{j},\boldsymbol{h}^{j}) \mid \bar{\boldsymbol{x}}^{j}] \leq \tilde{\boldsymbol{\alpha}}^{j} = \mathbb{E}[\tilde{\boldsymbol{\alpha}}^{j}(\boldsymbol{h}^{j})]$$
(6.88)

where the last equality follows from (6.32). Since all the above quantities are probabilities, i.e., take values in [0, 1] we conclude that

$$-\prod_{j\neq i} \mathbb{E}\left[1 - \boldsymbol{\alpha}^{*j}(x^j, h^j) \,\big|\, \bar{x}\right] \le -\prod_{j\neq i} \mathbb{E}\left[1 - \tilde{\boldsymbol{\alpha}}^j(h^j) \,\big|\, \bar{x}\right].$$
(6.89)

Applying this inequality to (6.87), and since the matrix M^i is positive semidefinite by Assumption 6.1, we have that

$$\mathbf{F}(\boldsymbol{\alpha}^{i},\boldsymbol{\alpha}^{*-i}) \leq \mathbf{F}(\boldsymbol{\alpha}^{i},\tilde{\boldsymbol{\alpha}}^{-i})$$
(6.90)

holds for any transmission function α^i .

Now consider the problem of minimizing $\mathbf{F}(\boldsymbol{\alpha}^{i}, \tilde{\boldsymbol{\alpha}}^{-i})$ over functions $\boldsymbol{\alpha}^{i}$ such that

$$\mathbb{E}[\boldsymbol{\alpha}^{i}(\boldsymbol{x}^{i},\boldsymbol{h}^{i}) \mid \bar{\boldsymbol{x}}^{i}] \leq \tilde{\boldsymbol{\alpha}}^{i} \tag{6.91}$$

holds. By the expression (6.87) we have that the objective equals

$$\mathbf{F}(\boldsymbol{\alpha}^{i}, \tilde{\boldsymbol{\alpha}}^{-i}) = -\mathbb{E}\left[\left(\boldsymbol{\alpha}^{i}(x^{i}, h^{i}) - \tilde{\boldsymbol{\alpha}}^{i}(h^{i})\right) q(h^{i}) x^{i^{T}} M^{i} x^{i} \mid \bar{x}\right] \prod_{j \neq i} \left[1 - \tilde{\alpha}^{j}\right]$$
(6.92)

This is exactly the same objective as the one appearing in the optimization problem (6.38)-(6.39), whose optimal solution is α^{*i} . Hence the optimal solution is the same for both problems and it satisfies

$$\mathbf{F}(\boldsymbol{\alpha}^{*i}, \tilde{\boldsymbol{\alpha}}^{-i}) = \min_{\boldsymbol{\alpha}^{i} \text{ s.t. (6.91)}} \mathbf{F}(\boldsymbol{\alpha}^{i}, \tilde{\boldsymbol{\alpha}}^{-i}).$$
(6.93)

Now note that the reference transmission function $\tilde{\alpha}^i$ is by construction a feasible solution to this problem, in general suboptimal, hence it must be that

$$\mathbf{F}(\boldsymbol{\alpha}^{*i}, \tilde{\boldsymbol{\alpha}}^{-i}) \le \mathbf{F}(\tilde{\boldsymbol{\alpha}}^{i}, \tilde{\boldsymbol{\alpha}}^{-i})$$
(6.94)

Recall however that $\mathbf{F}(\tilde{\alpha}^{i}, \tilde{\alpha}^{-i}) = 0$ due to (6.76). Hence

$$\mathbf{F}(\boldsymbol{\alpha}^{*i}, \tilde{\boldsymbol{\alpha}}^{-i}) \le 0 \tag{6.95}$$

Combining (6.90) with (6.95) verifies (6.85) and completes the proof of the lemma.

6.4.5. Proof of Proposition 6.3

To simplify notation we will drop the index *i* within the proof and we will also omit the constant term $\prod_{j \neq i} (1 - \tilde{\alpha}_j)$. More specifically we will prove under the same assumptions that the following optimization problem

$$\min_{\boldsymbol{\alpha}(.)\in\mathcal{A}} \mathbb{E}\left[-\boldsymbol{\alpha}(x,h)\,q(h,p)\,x^{T}Mx\;\big|\,\bar{x}\right]$$
(6.96)

subject to
$$\mathbb{E}[\boldsymbol{\alpha}(x,h) \mid \bar{x}] \leq \tilde{\alpha}.$$
 (6.97)

for any fixed value $\bar{x} \in \mathbb{R}^n$ has a solution of the form

$$\boldsymbol{\alpha}^{*}(x,h) = \begin{cases} 1, & \text{if } q(h,p) x^{T} M x \ge \nu \\ 0 & \text{otherwise.} \end{cases}$$
(6.98)

for some non-negative $\nu \ge 0$ which depends on \bar{x} .

We will show that problem (6.96)-(6.97) has zero duality gap and that ν corresponds to the optimal Lagrange dual variable. This is based on the results of Ribeiro (2012) where stochastic optimization problems of a similar mathematical structure are considered.

First, note that the expectations in (6.96)-(6.97) are well defined for any measurable function $\alpha \in A$. This is true for (6.97) as α takes values in the finite interval [0, 1], and for (6.96) as the function q(.) is bounded and the distribution of x has finite second moments by assumption. Second, the problem is strictly feasible. For example a constant policy $\alpha(x, h) = \hat{\alpha}$ for all $x \in \mathbb{R}^n$, $h \in \mathbb{R}_+$ for some $\hat{\alpha} \in [0, \tilde{\alpha})$ strictly satisfies (5.21). Third, note that the random variables x, h have an absolutely continuous probability distribution given by the product of the plant disturbance distribution ϕ_w and the channel distribution ϕ_h which are abs. continuous by assumption.

Let us denote the optimal value of (6.96)-(6.97) by P^* and let us define the Lagrange dual problem. Consider a dual variable $\nu \in \mathbb{R}_+$. Then the Lagrangian function can be written as

$$L(\boldsymbol{\alpha}, \boldsymbol{\nu}) = \mathbb{E}[-\boldsymbol{\alpha}(\boldsymbol{x}, \boldsymbol{h}) q(\boldsymbol{h}, \boldsymbol{p}) \boldsymbol{x}^{T} \boldsymbol{M} \boldsymbol{x}] + \boldsymbol{\nu}(\mathbb{E}[\boldsymbol{\alpha}(\boldsymbol{x}, \boldsymbol{h})] - \tilde{\boldsymbol{\alpha}})$$
(6.99)

where for simplicity we dropped the conditioned expectation notation as \bar{x} is fixed. Note then that by additivity of expectation we can rewrite the Lagrangian as

$$L(\boldsymbol{\alpha}, \boldsymbol{\nu}) = \mathbb{E}[-\boldsymbol{\alpha}(x, h) \left(q(h, p) x^T M x - \boldsymbol{\nu}\right)] - \boldsymbol{\nu} \tilde{\boldsymbol{\alpha}}$$
(6.100)

The Lagrange dual problem whose optimal value we denote by D^* is posed as

$$D^* = \underset{\nu \in \mathbb{R}_+}{\operatorname{maximize}} \min_{\boldsymbol{\alpha} \in \mathcal{A}} L(\boldsymbol{\alpha}, \nu).$$
(6.101)

Then we can establish the following results.

By (Ribeiro, 2012, Theorem 1) the problem (6.96)-(5.21) has zero duality gap, i.e., $P^* = D^*$. Moreover by (Ribeiro, 2012, Theorem 4), if α^* is an optimal primal solution and ν^* is an optimal solution for the dual problem (6.101), then

$$\boldsymbol{\alpha}^* \in \operatorname*{argmin}_{\boldsymbol{\alpha} \in \mathcal{A}} L(\boldsymbol{\alpha}, \boldsymbol{\nu}^*). \tag{6.102}$$

These results hold under the abs. continuous probability measures and strict feasibility which we verified above.

The above characterization (6.102) suggests that we can recover the optimal variables α^* by just minimizing the unconstrained Lagrangian function. Given the form of the Lagrangian given in (6.100) we rewrite (6.102) as

$$\boldsymbol{\alpha}^* \in \operatorname*{argmin}_{\boldsymbol{\alpha} \in \mathcal{A}} \mathbb{E}[-\boldsymbol{\alpha}(x,h) \left(q(h,p) \, x^T M x - \nu^*\right)]. \tag{6.103}$$

Moreover minimizing over the function α defined for all arguments $x \in \mathbb{R}^n$, $h \in \mathbb{R}_+$ above is equivalent to minimizing at each argument separately, that is

$$\boldsymbol{\alpha}^{*}(x,h) \in \operatorname*{argmin}_{\boldsymbol{\alpha}(x,h) \in [0,1]} - \boldsymbol{\alpha}(x,h) \left(q(h,p) \, x^{T} M x - \nu^{*} \right). \tag{6.104}$$

In other words we have exchanged the expectation and the minimization. This is done almost surely without loss of optimality, i.e., an optimal solution α^* needs to satisfy (6.104) at all points x, h except perhaps for a measure zero set. The optimal value of $\alpha(x, h) \in [0, 1]$ at each point x, h takes value either 0 or 1 according to (6.98) given in the statement. Technically, since (6.104) is an inclusion we have to argue that the minimizers at the right hand side are almost surely unique. Indeed the set of points x, h where the minimizer is not unique is the set defined as

$$S = \{x, h : q(h, p) \ x^T M x = \nu^*\}.$$
(6.105)

Note then that

$$\mathbb{P}(S) = \mathbb{E}[\mathbb{P}[q(h, p) = \nu^* / x^T M x \mid x]] = \mathbb{E}[0] = 0.$$
(6.106)

That is because the function q(h, p) is assumed strictly increasing in h, hence there is at most a single point h meeting the equality, and since the distribution of h is abs. continuous the integral over that set must be zero.

To sum up, we have shown that for each value \bar{x} there exists a constant $\nu \ge 0$ such that the optimal solution is given by (6.98). This concludes the proof.

Chapter 7: Conclusion

The goal of this work was to design resource allocation policies for control systems implemented over wireless channels. Towards this goal, appropriate abstractions of wireless control systems were developed in Chapter 2. On the one hand this includes the mathematical description of dynamical systems involving sensors and actuators communicating between different physical locations. The main feature of these systems is that they operate on different modes, open and closed loop, depending on whether transmissions occur successfully or not. Appropriate abstractions of long term control performance are also introduced. On the other hand, the uncertainties in wireless communication are also explicitly modeled with wireless channel models. These models not only account for the randomly varying fading channel conditions, but also explicitly capture the allocation of transmit power resources. This model facilitates the analysis and design of resource allocation algorithms in wireless control systems.

The fundamental problem of co-designing transmit powers and control inputs for a wireless sensor-actuator system is considered in Chapter 3. The goal of the design is to minimize jointly average (linear quadratic) control performance and average power utilization. Separation of the two designs is made possible by a suboptimal decoupled information structure. The resulting optimal power allocation is characterized qualitatively and an opportunistic adaptation nature is revealed. Transmission should be avoided under adverse channel conditions or when the control system is under desirable conditions, while power allocation should increase as the system deviates. This introduces a novel paradigm for operating power-constrained wireless control systems. Related publications include Gatsis et al. (2013b, 2014c). Preliminary extensions to the problem of managing receiver power resources are presented in Gatsis et al. (2013a).

Channel-aware schedulers for wireless control systems with multiple loops closing over a shared wireless medium are considered in Chapter 4. This is posed as a constrained stochastic optimization problem where desired control performance requirements need to be guaranteed for each system, while the overall transmit power utilization is to be minimized. The structure of the optimal solution is characterized. The scheduler opportunistically decides one plant to close the loop at each time step, while transmit powers can be decoupled among the different sensors. We develop an offline optimization algorithm to solve the problem, as well as an online communication algorithm that converges to the optimal performance using only random observed channel sequences. Related publications include Gatsis et al. (2014a, 2015a). An extension of the approach to scheduling inter-dependent control tasks is considered in Gatsis et al. (2014b).

With decentralized channel access algorithms, sensors can independently decide whether to access the shared medium, without the need of a centralized scheduler. However the emerging wireless interferences cause control performance degradation, hence need to be mitigated. In Chapter 5 we develop a random access mechanism for control systems. The proposed policies are decoupled among systems and balance the control performance gains from transmissions with the losses due to interferences. Related publications include Gatsis et al. (2015c, 2016a). A different algorithm based on a game-theoretic approach is considered in Gatsis et al. (2015b) allowing the sensors to learn their policies in a decentralized fashion.

Finally in Chapter 6 we design dynamic sensor transmission policies adapting online to the varying control system state as well as the wireless channel state. Based on approximate dynamic programming the proposed policies come with improved control performance and resource utilization, as compared to simple policies used as a reference. The approach is illustrated in the problem of transmit power allocation, in the problem of scheduling, as well as in the problem of decentralized channel access. The latter case is particularly complex as the sensor policies are coupled globally over the shared medium but are allowed to adapt only to locally available information. Related publications include Gatsis et al. (2016b,c).

7.1. Open Problems and Future Research Directions

The future of modern smart infrastructures entails large numbers of interconnected devices in interaction with the physical world. As the control inputs applied to the system by actuators rely on wirelessly received data collected by sensors, the joint design of efficient control and wireless communication policies remains an important challenge. Despite recent advances towards this end (e.g. Gatsis et al. (2014c); Nayyar et al. (2013)), the determination of appropriate information structures as well as computationally tractable policy design algorithms requires further consideration.

The efficient allocation of communication resources, as in the problems examined in this work, needs to account for a large number of interconnected closed loop systems. To enable such a large scale design appropriate optimization procedures need to be developed, for example, distributed optimization algorithms and efficient spectrum management. As wireless sensors and actuators become available to the rest of the system in a time varying fashion, i.e., randomly connect or disconnect, the communication algorithms need to respond and adapt online to these changes and maintain efficiency.

As control systems change dynamically over time, opportunistically adapted mechanisms such as those presented in Chapter 6 become apparent. These mechanisms can respond to sensor measurements that are critical for the performance of the system, however determining when such critical measurements take place is challenging because of the large scale nature of the systems and the fact that information is only locally available. Moreover, to achieve efficient utilization of communication resources over a long planning horizon, appropriate abstractions of control performance need to be taken into account.

Furthermore, with increased connectivity an increased number of security concerns arise. Sen-

sor measurements may not always be desired to be shared with other connected systems due to privacy concerns. In that case wireless mechanisms need not only account for reliable communication but also safe communication, e.g., in the presence of eavesdroppers. Alternatively if malicious agents are able to intrude and inject false measurements over the communication channel, it is essential to determine defense/controller strategies that would maintain control system integrity.

Appendix A: Imperfect Channel State Information

According to the wireless communication model of Chapter 2 the sensor/transmitter relies on channel state information to decide whether to access the channel to transmit and close the loop. Perfect channel state information is difficult to acquire at the transmitter side in practice, hence some channel estimation procedure needs to be followed – see, e.g., Hassibi and Hochwald (2003) for a discussion. In this section we illustrate how our developments can be adapted in this case.

A common model for imperfect channel state information in the literature is that *channel outage* occurs when the attempted transmission rate is higher than the maximum rate supported by the current channel conditions (Vakili et al. (2006); Zheng et al. (2008); Ouyang et al. (2010); Hu and Ribeiro (2013)). More specifically, suppose at some time step the channel state is given by $h_k \in \mathbb{R}_+$ while the sensor only knows a channel estimate $\hat{h}_k \in \mathbb{R}_+$. We do not examine a specific channel estimation procedure but instead characterize the estimation quality by a conditional probability distribution $\mathbb{P}(h | \hat{h})$.

The current channel state h_k can support a packet success rate described by the function $q(h_k p)$ depending on received power level $h_k p$ as described in Section 2.2. To achieve this rate however the transmitter needs to select a code appropriately adapted to the channel state h_k which is not perfectly known. Suppose then that the sensor attempts to transmit at the *estimated packet success rate* $q(\hat{h}_k p)$ instead. The probability of successfully receiving the message is then modeled as

$$\mathbb{P}(\gamma_k = 1 \mid h_k, \hat{h}_k) = \begin{cases} q(\hat{h}_k p) & \text{if } \hat{h}_k \le h_k, \\ 0 & \text{otherwise} \end{cases}$$
(A.1)

That is, the rate equals the attempted rate if the channel fading gain is higher than the estimated gain, otherwise a channel outage occurs and the rate is zero. Depending on how reliable the channel estimation procedure is, the channel outage may result in significant performance losses.

A commonly proposed solution to mitigate this effect of channel outage is to employ a *backoff* rate. Given the current channel estimate the transmitter can attempt to communicate at a lower rate than the estimated rate to protect against the case where the actual channel state is worse than the estimated one. Instead of transmitting at the estimate rate $q(\hat{h}_k p)$ the sensor can alternatively select a backoff rate, or equivalently a backoff channel gain $b(\hat{h}_k)$ depending on the current channel estimate. The transmitter then attempts the backoff packet success rate $q(b(\hat{h}_k) p)$. The resulting probability of successful transmission then becomes

$$\mathbb{P}(\gamma_k = 1 \mid h_k, \hat{h}_k) = \begin{cases} q(b(\hat{h}_k) p) & \text{if } b(\hat{h}_k) \le h_k, \\ 0 & \text{otherwise} \end{cases}$$
(A.2)

Integrating (A.2) with respect to the estimation quality model $\mathbb{P}(h_k | \hat{h}_k)$ we conclude that

$$\mathbb{P}(\gamma_k = 1 \mid \hat{h}_k) = q(b(\hat{h}_k)) \mathbb{P}(b(\hat{h}_k) \le h_k \mid \hat{h}_k).$$
(A.3)

The backoff channel rate $b(\hat{h}_k)$ can then be selected at each channel estimate \hat{h}_k in order to max-

imize the probability of success given in the above expression. There is a tradeoff in this design. Low backoff rates imply fewer outages, as the second term in the product above increases, but also lower achieved rates, as the first term decreases. Typically backoff channels/rates are selected lower than the channel estimates, $b(\hat{h}_k) \leq \hat{h}_k$, so that the outage event $b(\hat{h}_k) > h_k$ happens with lower probability (Hu and Ribeiro (2013)). After a backoff mechanism is selected, we can denote (A.3) as a general function of the form

$$\mathbb{P}(\gamma_k = 1 \mid \hat{h}_k) = \hat{q}(\hat{h}_k). \tag{A.4}$$

The policies based on channel states presented in this work are still valid by replacing the general channel model (2.17) with the modified one in (A.4) expressed with respect to channel estimates. For example, consider a sensor obtaining channel state estimates \hat{h}_k , $k \ge 0$ which are i.i.d. over time and deciding whether to communicate over the channel in order to close its loop. Let us indicate this choice by a variable $\alpha_k \in [0, 1]$. Similar to the development of rollout policies (6.12), (6.28) in Sections 6.1, 6.2 we can propose the following policy

$$\alpha_k = \underset{\alpha \in [0,1]}{\operatorname{argmin}} -\alpha \,\hat{q}(\hat{h}_k) \, x_k^T M x_k + \nu \, \alpha. \tag{A.5}$$

Here $\nu \ge 0$ is a non-negative constant parameter and the matrix *M* corresponds to the relative value-of-transmitting the current state x_k , as introduced in (6.11). Due to linearity of the objective the solution α_k to (A.5) takes values either 1 or 0, transmit or not. The sensor transmits when the expected value of transmission $\hat{q}(\hat{h}_k) x_k^T M x_k$ exceeds the parameter ν . We note that here the expected value of transmission takes into account the possible current channel state values given the current channel state estimate.

Appendix B: Control over Markov Fading Channels

In this section we briefly present how our resource allocation approach needs to be modified in cases where the channel fading conditions between sensors and actuators are not independent over time but follow instead a Markov process (Mushkin and Bar-David (1989); Ahmad et al. (2009); Ouyang et al. (2012)). To simplify the exposition suppose the channel can take one of two states, h_0 , h_1 and the transition probability between states h_i and h_j is denoted as p_{ij} , for $i, j \in \{1, 2\}$. The resulting Markov chain is assumed to be irreducible and aperiodic so that it possesses a stationary distribution. As introduced in 2.2 the probability of successful transmission at a channel state h_i when transmitting with power p is modeled by the relationship $q(h_i, p)$.

Consider now the problem of optimal power allocation according to Chapter 3 where we are interested in minimizing the objective

$$\limsup_{N \to +\infty} \frac{1}{N} \mathbb{E} \sum_{k=0}^{N-1} x_k^T Q x_k + u_k^T R u_k + \lambda p_k.$$
(B.1)

Assuming that an optimal solution exists and that it technically satisfies the Bellman equation, as is the case in Theorem 3.2, then we can characterize its form as follows. When the current plant conditions are given by $\varepsilon_k \in \mathbb{R}^n$ (the innovation term in (3.11)) and the current channel conditions are given by $h_k \in \{h_0, h_1\}$, the optimal transmit power at the sensor is given by

$$p_k^* = \underset{p \in [0, p_{\max}]}{\operatorname{argmin}} \lambda p - q(h_k, p) R(\varepsilon_k, h_k)$$
(B.2)

for some function $R(\varepsilon, h)$. Comparing this with the optimal power allocation for the i.i.d. channel case presented in (3.34) we see that here the channel h_k structurally plays a dual role. The current channel state h_k not only shows how easy it is to transmit at the current time step (by the term $q(h_k, p)$) but also reveals information about future gains (by the term $R(\varepsilon_k, h_k)$) since the channel state at the next time step is correlated.

We also briefly present how the Markovian channel structure affects the design of rollout power allocation policies as presented in Section 6.1. To begin with, suppose a reference policy is available which selects a power level $\tilde{p}_i \in [0, p_{\text{max}}]$ under channel conditions h_i , for i = 0, 1. We denote the corresponding packet success probabilities at the two possible channel states according to the reference policy as

$$\tilde{q}_i := q(h_i, \tilde{p}_i), \quad i = 0, 1.$$
 (B.3)

Suppose then the control system dynamics are given by

$$x_{k+1} = \begin{cases} A_c x_k + w_k, & \text{if } \gamma_k = 1\\ A_o x_k + w_k, & \text{if } \gamma_k = 0 \end{cases}$$
(B.4)

as in Section 6.1. Also for simplicity suppose control system states are penalized at each time step by a quadratic cost of the form $x_k^T Q x_k$, i.e., the cost does not depend on the current packet success (cf. (6.3)). Similar to (6.9), we first solve for matrices corresponding to the future control cost-to-go of the reference policy. Due to the Markovian structure, here we have to actually solve for a set of matrices $\{P_i, i = 0, 1\}$, one for each possible channel state value. Intuitively that is because the future expected control cost depends the current channel state. In particular one needs to solve

$$P_{i} = Q + \sum_{j=0,1} p_{ij} \left\{ \tilde{q}_{j} A_{c}^{T} P_{j} A_{c} + (1 - \tilde{q}_{j}) A_{o}^{T} P_{j} A_{o} \right\}, \text{ for } i = 0, 1.$$
(B.5)

These are a set of coupled linear matrix equalities that have to be jointly solved, e.g., via the vectorization approach in Remark 6.2.

Given the above reference policy, we can construct rollout power allocation policies that improve upon the reference performance, similar to the approach in Section 6.1. It turns out that the power allocation takes the following form. If the current plant state is $h_k = h_i$, for i = 0, 1 and current control system state is x_k , the transmit power according to the rollout policy becomes

$$p_{k}^{\text{roll}} = \underset{p \in [0, p_{\text{max}}]}{\operatorname{argmin}} \nu p - q(h_{i}, p) x_{k}^{T} (A_{o}^{T} P_{i} A_{o} - A_{c}^{T} P_{i} A_{c}) x_{k}$$
(B.6)

where $\nu \ge 0$ is a scalar parameter. This power choice is in general of the same form as in the i.i.d. channel state (cf. (6.12)). A closer look reveals again the dual role of the current channel state as it appears both in the term $q(h_i, p)$ and in the quadratic form $x_k^T (A_o^T P_i A_o - A_c^T P_i A_c) x_k$. The latter quadratic form can be interpreted as the value of transmitting under the current control state x_k and the current channel state $h_k = h_i$. The reason for the latter is that the current channel states are coupled over time in a Markovian fashion. Hence current channel conditions are taken into account in evaluating future control performance gains.

BIBLIOGRAPHY

- S. Adireddy and L. Tong. Exploiting decentralized channel state information for random access. *IEEE Transactions on Information Theory*, 51(2):537–561, 2005.
- S. H. A. Ahmad, M. Liu, T. Javidi, Q. Zhao, and B. Krishnamachari. Optimality of myopic sensing in multichannel opportunistic access. *IEEE Transactions on Information Theory*, 55(9):4040–4050, 2009.
- A. Anta and P. Tabuada. To sample or not to sample: Self-triggered control for nonlinear systems. *IEEE Transactions on Automatic Control*, 55(9):2030–2042, 2010.
- D. Antunes and W. Heemels. Rollout event-triggered control: beyond periodic control performance. *IEEE Transactions on Automatic Control*, 59(12):3296–3311, 2014.
- D. J. Antunes, W. Heemels, J. P. Hespanha, and C. Silvestre. Scheduling measurements and controls over networkspart i: Rollout strategies for protocol design. In 2012 American Control Conference (ACC), pages 2036–2041. IEEE, 2012.
- K. J. Astrom and B. M. Bernhardsson. Comparison of Riemann and Lebesgue sampling for first order stochastic systems. In Proc. of the 41st IEEE Conference on Decision and Control, volume 2, pages 2011–2016, 2002.
- D. P. Bertsekas. Dynamic Programming and Optimal Control. Athena Scientific, 2005.
- D. P. Bertsekas, A. Nedić, and A. E. Ozdaglar. *Convex analysis and optimization*. Athena Scientific, 2003.
- D. Bertsimas and J. N. Tsitsiklis. Introduction to linear optimization. Athena Scientific Belmont, 1997.
- P. Billingsley. Probability and measure. John Wiley & Sons, 1995.
- R. Blind and F. Allgöwer. Analysis of networked event-based control with a shared communication medium: Part I-Pure ALOHA. In *IFAC World Congress*, 2011.
- V. S. Borkar and S. K. Mitter. LQG control with communication constraints. Springer, 1997.
- S. Boyd and L. Vandenberghe. Convex Optimization. Cambridge University Press, 2009.
- M. S. Branicky, S. M. Phillips, and W. Zhang. Scheduling and feedback co-design for networked control systems. In *Proc.of the 41st IEEE Conf. on Dec. and Control (CDC)*, volume 2, pages 1211– 1217, 2002.
- J. H. Braslavsky, R. H. Middleton, and J. S. Freudenberg. Feedback stabilization over signal-tonoise ratio constrained channels. *IEEE Transactions on Automatic Control*, 52(8):1391–1403, 2007.
- R. W. Brockett and D. Liberzon. Quantized feedback stabilization of linear systems. *IEEE Transac*tions on Automatic Control, 45(7):1279–1289, 2000.
- R. Cardell-Oliver, K. Smettem, M. Kranz, and K. Mayer. Field testing a wireless sensor network for reactive environmental monitoring [soil moisture measurement]. In *Proceedings of the 2004 Intelligent Sensors, Sensor Networks and Information Processing Conference, 2004.*, pages 7–12. IEEE, 2004.
- A. Cervin and T. Henningsson. Scheduling of event-triggered controllers on a shared network. In *Proc. of the 47th IEEE Conf. on Dec. and Control (CDC)*, pages 3601–3606, 2008.
- R. Cogill. Event-based control using quadratic approximate value functions. In *Proc. of the 48th IEEE Conference on Decision and Control, 2009 held jointly with the 2009 28th Chinese Control Conference,* pages 5883–5888, 2009.

- R. Cogill, S. Lall, and J. Hespanha. A constant factor approximation algorithm for event-based sampling. In Proc. of the 2007 American Control Conference, pages 305–311, 2007.
- O. Costa and M. Fragoso. Stability results for discrete-time linear systems with markovian jumping parameters. *Journal of mathematical analysis and applications*, 179(1):154–178, 1993.
- S. De Vito, P. D. Palma, C. Ambrosino, E. Massera, G. Burrasca, M. L. Miglietta, and G. D. Francia. Wireless sensor networks for distributed chemical sensing: Addressing power consumption limits with on-board intelligence. *IEEE Sensors Journal*, 11(4):947–955, 2011.
- M. Donkers, W. Heemels, N. Van De Wouw, and L. Hetel. Stability analysis of networked control systems using a switched linear systems approach. *IEEE Transactions on Automatic Control*, 56 (9):2101–2115, 2011.
- R. Durrett. Probability: theory and examples. Cambridge University Press, 2010.
- M. Egerstedt and Y. Wardi. Multi-process control using queuing theory. In *Proc. of the 41st IEEE Conf. on Dec. and Control (CDC)*, volume 2, pages 1991–1996, 2002.
- N. Elia. Remote stabilization over fading channels. Systems & Control Letters, 54(3):237–249, 2005.
- M. Franceschetti and P. Minero. Elements of information theory for networked control systems. In *Information and Control in Networks*, pages 3–37. Springer, 2014.
- K. Gatsis, M. Pajic, A. Ribeiro, and G. J. Pappas. Power-aware communication for wireless sensoractuator systems. In *Proc. of the 52nd IEEE Conf. on Dec. and Control (CDC)*, pages 4006–4011, 2013a.
- K. Gatsis, A. Ribeiro, and G. J. Pappas. Optimal power management in wireless control systems. In *Proc. of the 2013 American Control Conference (ACC 2013)*, pages 1562–1569, 2013b.
- K. Gatsis, M. Pajic, A. Ribeiro, and G. J. Pappas. Opportunistic scheduling of control tasks over shared wireless channels. In ACM/IEEE 5th Int. Conf. on Cyber-Physical Systems (ICCPS), pages 48–59, 2014a.
- K. Gatsis, M. Pajic, A. Ribeiro, and G. J. Pappas. Opportunistic sensor scheduling in wireless control systems. In 53rd IEEE Conf. on Dec. and Control (CDC), pages 3777–3782, 2014b.
- K. Gatsis, A. Ribeiro, and G. J. Pappas. Optimal power management in wireless control systems. *IEEE Transactions on Automatic Control*, 59(6):1495–1510, 2014c.
- K. Gatsis, M. Pajic, A. Ribeiro, and G. J. Pappas. Opportunistic control over shared wireless channels. *IEEE Transactions on Automatic Control*, 60(12):3140–3155, 2015a.
- K. Gatsis, A. Ribeiro, and G. J. Pappas. Decentralized channel access for wireless control systems. 5th IFAC Workshop on Distributed Estimation and Control in Networked Systems (NecSys), IFAC-PapersOnLine, 48(22):209–214, 2015b.
- K. Gatsis, A. Ribeiro, and G. J. Pappas. Control with random access wireless sensors. In 54th IEEE Conference on Decision and Control, pages 318–323, 2015c.
- K. Gatsis, A. Ribeiro, and G. Pappas. Random access design for wireless control systems. *Automatica*, (submitted), April 2016a. Available at http://arxiv.org/pdf/1605.00627v1.pdf.
- K. Gatsis, A. Ribeiro, and G. J. Pappas. Control-aware random access communication. In Proc. of the ACM/IEEE 7th International Conference on Cyber-Physical Systems. Austria Vienna, April 11-14 2016b.

- K. Gatsis, A. Ribeiro, and G. J. Pappas. State-based communication design for wireless control systems. In *Proc. Conf. on Decision Control*, volume (submitted). Las Vegas, December 12-14 2016c.
- L. Georgiadis, M. J. Neely, and L. Tassiulas. *Resource allocation and cross-layer control in wireless networks*. Now Publishers Inc, 2006.
- A. Goldsmith. Wireless communications. Cambr. Univ. Press, 2005.
- V. Gupta, T. H. Chung, B. Hassibi, and R. M. Murray. On a stochastic sensor selection algorithm with applications in sensor scheduling and sensor coverage. *Automatica*, 42(2):251–260, 2006.
- V. Gupta, B. Hassibi, and R. Murray. Optimal LQG control across packet-dropping links. Systems & Control Letters, 56(6):439–446, 2007.
- C. N. Hadjicostis and R. Touri. Feedback control utilizing packet dropping network links. In Decision and Control, 2002, Proceedings of the 41st IEEE Conference on, volume 2, pages 1205–1210. IEEE, 2002.
- D. Han, Y. Mo, J. Wu, S. Weerakkody, B. Sinopoli, and L. Shi. Stochastic event-triggered sensor schedule for remote state estimation. *IEEE Transactions on Automatic Control*, 60(10):2661–2675, 2015.
- B. Hassibi and B. M. Hochwald. How much training is needed in multiple-antenna wireless links? *Information Theory, IEEE Transactions on,* 49(4):951–963, 2003.
- W. Heemels, K. H. Johansson, and P. Tabuada. An introduction to event-triggered and self-triggered control. In *51st Annual Conference on Decision and Control*, pages 3270–3285, 2012.
- W. Heemels, M. Donkers, and A. R. Teel. Periodic event-triggered control for linear systems. *IEEE Transactions on Automatic Control*, 58(4):847–861, 2013.
- W. M. H. Heemels, A. R. Teel, N. van de Wouw, and D. Nesic. Networked control systems with communication constraints: Tradeoffs between transmission intervals, delays and performance. *IEEE Transactions on Automatic control*, 55(8):1781–1796, 2010.
- O. Hernández-Lerma and J. Lasserre. *Discrete-time Markov control processes: basic optimality criteria*. Springer New York, 1996.
- J. Hespanha, P. Naghshtabrizi, and Y. Xu. A survey of recent results in networked control systems. *Proceedings of the IEEE*, 95(1):138–162, 2007.
- E. J. Horvitz, J. Apacible, R. Sarin, and L. Liao. Prediction, expectation, and surprise: Methods, designs, and study of a deployed traffic forecasting service. arXiv preprint arXiv:1207.1352, 2012.
- D. Hristu-Varsakelis. Feedback control systems as users of a shared network: Communication sequences that guarantee stability. In *Proc. of the 40th IEEE Conf. on Dec. and Control (CDC)*, volume 4, pages 3631–3636, 2001.
- D. Hristu-Varsakelis and P. R. Kumar. Interrupt-based feedback control over a shared communication medium. In *Proc. of the 41st IEEE Conf. on Dec. and Control (CDC)*, volume 3, pages 3223–3228, 2002.
- Y. Hu and A. Ribeiro. Adaptive distributed algorithms for optimal random access channels. *IEEE Transactions on Wireless Communications*, 10(8):2703–2715, 2011.
- Y. Hu and A. Ribeiro. Optimal wireless communications with imperfect channel state information. *Signal Processing, IEEE Transactions on*, 61(11):2751–2766, 2013.

- O. C. Imer and T. Başar. Optimal estimation with limited measurements. In 44th IEEE Conference on Decision and Control, 2005 and 2005 European Control Conference. CDC-ECC'05., pages 1029–1034. IEEE, 2005.
- O. C. Imer and T. Başar. Optimal estimation with limited measurements. *International Journal of Systems, Control and Communications,* 2(1-3):5–29, 2010.
- O. C. Imer, S. Yüksel, and T. Başar. Optimal control of LTI systems over unreliable communication links. *Automatica*, 42(9):1429–1439, 2006.
- Y. W. Kim, S. J. Lee, G. H. Kim, and G. J. Jeon. Wireless electronic nose network for real-time gas monitoring system. In *IEEE International Workshop on Robotic and Sensors Environments*, 2009. ROSE 2009., pages 169–172. IEEE, 2009.
- A. J. Laub. Matrix analysis for scientists and engineers. SIAM, 2005.
- J. Le Ny, E. Feron, and G. J. Pappas. Resource constrained lqr control under fast sampling. In *Proc.* of the 14th International Conference on Hybrid Systems: Computation and Control, pages 271–280. ACM, 2011.
- A. S. Leong and S. Dey. Power allocation for error covariance minimization in Kalman filtering over packet dropping links. In Proc. of the 51st IEEE Conference on Decision and Control, pages 3335–3340, 2012.
- L. Li and M. Lemmon. Performance and average sampling period of sub-optimal triggering event in event triggered state estimation. In *Proc. of the 50th IEEE Conference on Decision and Control and European Control Conference (CDC-ECC 2011)*, pages 1656–1661, 2011.
- L. Li, M. Lemmon, and X. Wang. Event-triggered state estimation in vector linear processes. In *American Control Conference (ACC)*, pages 2138–2143, 2010.
- G. M. Lipsa and N. C. Martins. Remote state estimation with communication costs for first-order lti systems. *IEEE Transactions on Automatic Control*, 56(9):2013–2025, 2011.
- J. W. Liu. Real-Time Systems. Prentice-Hall, Inc, 2000.
- X. Liu and A. Goldsmith. Wireless medium access control in networked control systems. In *Proc.* of the 2004 American Control Conference (ACC), volume 4, pages 3605–3610, 2004.
- X. Liu, E. K. Chong, and N. B. Shroff. A framework for opportunistic scheduling in wireless networks. *Computer Networks*, 41(4):451–474, 2003.
- W. Luo and A. Ephremides. Power levels and packet lengths in random multiple access. *IEEE Transactions on Information Theory*, 48(1):46–58, 2002.
- Y. Ma, A. Kelman, A. Daly, and F. Borrelli. Predictive control for energy efficient buildings with thermal storage. *IEEE Control System Magazine*, 32(1):44–64, 2012.
- A. Mahajan. Sequential decomposition of sequential dynamic teams: applications to real-time communication and networked control systems. ProQuest, Ph.D. dissertation, University of Michigan, 2008.
- N. C. Martins and M. A. Dahleh. Feedback control in the presence of noisy channels:bode-like fundamental limitations of performance. *IEEE Transactions on Automatic Control*, 53(7):1604–1615, 2008.
- L. Meier, J. Peschon, and R. M. Dressler. Optimal control of measurement subsystems. *IEEE Transactions on Automatic Control*, 12(5):528–536, 1967.
- A. Mesquita, J. Hespanha, and G. Nair. Redundant data transmission in control/estimation over lossy networks. *Automatica*, 48:1020–1027, 2012.

- A. Molin and S. Hirche. On LQG joint optimal scheduling and control under communication constraints. In *Proc. of the 48th IEEE Conference on Decision and Control,* pages 5832–5838, 2009.
- A. Molin and S. Hirche. Price-based adaptive scheduling in multi-loop control systems with resource constraints. *IEEE Transactions on Automatic Control*, 59(12):3282–3295, 2014.
- T. K. Moon. *Error correction coding: mathematical methods and algorithms.* Wiley, 2005.
- M. Mushkin and I. Bar-David. Capacity and coding for the gilbert-elliott channels. IEEE Transactions on Information Theory, 35(6):1277–1290, 1989.
- G. Nair, F. Fagnani, S. Zampieri, and R. Evans. Feedback control under data rate constraints: An overview. *Proceedings of the IEEE*, 95(1):108–137, 2007.
- A. Nayyar, T. Başar, D. Teneketzis, and V. V. Veeravalli. Optimal strategies for communication and remote estimation with an energy harvesting sensor. *IEEE Transactions on Automatic Control*, 58 (9):2246–2260, 2013.
- T. X. Nghiem and G. J. Pappas. Receding-horizon supervisory control of green buildings. In *American Control Conference (ACC)*, 2011, pages 4416–4421. IEEE, 2011.
- C. Nowzari and J. Cortés. Self-triggered coordination of robotic networks for optimal deployment. *Automatica*, 48(6):1077–1087, 2012.
- C. Nowzari, V. M. Preciado, and G. J. Pappas. Analysis and control of epidemics: A survey of spreading processes on complex networks. *IEEE Control Systems*, 36(1):26–46, 2016.
- F. Oldewurtel, A. Parisio, C. N. Jones, D. Gyalistras, M. Gwerder, V. Stauch, B. Lehmann, and M. Morari. Use of model predictive control and weather forecasts for energy efficient building climate control. *Energy and Buildings*, 45:15–27, 2012.
- R. Ostrovsky, Y. Rabani, and L. J. Schulman. Error-correcting codes for automatic control. *IEEE Transactions on Information Theory*, 55(7):2931–2941, 2009.
- W. Ouyang, S. Murugesan, A. Eryilmaz, and N. B. Shroff. Scheduling with rate adaptation under incomplete knowledge of channel/estimator statistics. In *Communication, Control, and Computing* (Allerton), 2010 48th Annual Allerton Conference on, pages 670–677. IEEE, 2010.
- W. Ouyang, A. Eryilmaz, and N. B. Shroff. Asymptotically optimal downlink scheduling over markovian fading channels. In *Proceedings IEEE INFOCOM*, pages 1224–1232. IEEE, 2012.
- N. Ploplys, P. Kawka, and A. Alleyne. Closed-loop control over wireless networks. *IEEE Control Systems Magazine*, 24(3):58–71, 2004.
- X. Qin and R. Berry. Distributed approaches for exploiting multiuser diversity in wireless networks. *IEEE Transactions on Information Theory*, 52(2):392–413, 2006.
- D. E. Quevedo, A. Ahlén, A. S. Leong, and S. Dey. On Kalman filtering over fading wireless channels with controlled transmission powers. *Automatica*, 48(7):1306–1316, 2012.
- M. Rabi, K. H. Johansson, and M. Johansson. Optimal stopping for event-triggered sensing and actuation. In *Proc. of the 47th IEEE Conference on Decision and Control.*, pages 3607–3612, 2008.
- M. Rabi, L. Stabellini, A. Proutiere, and M. Johansson. Networked estimation under contentionbased medium access. *Int. Journal of Robust and Nonlinear Control*, 20(2):140–155, 2010.
- M. Rabi, G. V. Moustakides, and J. S. Baras. Adaptive sampling for linear state estimation. *SIAM Journal on Control and Optimization*, 50(2):672–702, 2012.
- C. Ramesh, H. Sandberg, and K. H. Johansson. Design of state-based schedulers for a network of control loops. *IEEE Transactions on Automatic Control*, 58(8):1962–1975, 2013.

- H. Rehbinder and M. Sanfridson. Scheduling of a limited communication channel for optimal control. *Automatica*, 40(3):491–500, 2004.
- A. Ribeiro. Optimal resource allocation in wireless communication and networking. *EURASIP Journal on Wireless Communications and Networking*, 2012(1):1–19, 2012.
- L. Ruiz-Garcia, L. Lunadei, P. Barreiro, and I. Robla. A review of wireless sensor technologies and applications in agriculture and food industry: state of the art and current trends. *Sensors*, 9(6): 4728–4750, 2009.
- A. Sadilek, H. A. Kautz, and V. Silenzio. Modeling spread of disease from social interactions. In *ICWSM*, 2012.
- L. Schenato, B. Sinopoli, M. Franceschetti, K. Poolla, and S. Sastry. Foundations of control and estimation over lossy networks. *Proceedings of the IEEE*, 95(1):163–187, 2007.
- P. Seiler and R. Sengupta. Analysis of communication losses in vehicle control problems. In *Proceedings of the 2001 American Control Conference*, volume 2, pages 1491–1496. IEEE, 2001.
- B. Sinopoli, L. Schenato, M. Franceschetti, K. Poolla, M. Jordan, and S. Sastry. Kalman filtering with intermittent observations. *IEEE Transactions on Automatic Control*, 49(9):1453–1464, 2004.
- S. C. Smith and P. Seiler. Estimation with lossy measurements: jump estimators for jump systems. *IEEE Transactions on Automatic Control*, 48(12):2163–2171, 2003.
- R. T. Sukhavasi and B. Hassibi. Linear time-invariant anytime codes for control over noisy channels. *IEEE Transactions on Automatic Control (to appear)*, 2016.
- M. Tabbara and D. Nesic. Input–output stability of networked control systems with stochastic protocols and channels. *IEEE Transactions on Automatic Control*, 53(5):1160–1175, 2008.
- P. Tabuada. Event-triggered real-time scheduling of stabilizing control tasks. *IEEE Transactions on Automatic Control*, 52(9):1680–1685, 2007.
- T. Tanaka, P. M. Esfahani, and S. K. Mitter. Lqg control with minimal information: Three-stage separation principle and sdp-based solution synthesis. *arXiv preprint arXiv:1510.04214*, 2015.
- S. Tatikonda and S. Mitter. Control under communication constraints. IEEE Transactions on Automatic Control, 49(7):1056–1068, 2004.
- A. Vakili, M. Sharif, and B. Hassibi. The effect of channel estimation error on the throughput of broadcast channels. In Acoustics, Speech and Signal Processing, 2006. ICASSP 2006 Proceedings. 2006 IEEE International Conference on, volume 4, pages IV–IV. IEEE, 2006.
- P. Varaiya and J. Walrand. Causal coding and control for markov chains. *Systems & control letters*, 3(4):189–192, 1983.
- M. Vasconcelos and N. Martins. Remote estimation games over shared networks. In 52nd Annual Allerton Conf. on Communication, Control, and Computing, pages 12–18, 2014.
- O. Vega-Amaya. The average cost optimality equation: a fixed point approach. *Bol. Soc. Mat. Mexicana*, 9(1):185–195, 2003.
- J. C. Walrand and P. Varaiya. Optimal causal coding-decoding problems. *IEEE Transactions on Information Theory*, 29(6):814–820, 1983.
- G. C. Walsh, H. Ye, and L. G. Bushnell. Stability analysis of networked control systems. *IEEE Transactions on Control Systems Technology*, 10(3):438–446, 2002.

- A. Whittle, M. Allen, A. Preis, and M. Iqbal. Sensor networks for monitoring and control of water distribution systems. *Proceedings of the 6th International Conference on Structural Health Monitoring* of Intelligent Infrastructure (SHMII 2013), 2013.
- W. S. Wong and R. W. Brockett. Systems with finite communication bandwidth constraints. i. state estimation problems. *IEEE Transactions on Automatic Control*, 42(9):1294–1299, 1997.
- Y. Xu and J. Hespanha. Optimal communication logics in networked control systems. In *Proc. of the 43rd IEEE Conference on Decision and Control*, volume 4, pages 3527–3532, 2004.
- Y. Xu and J. P. Hespanha. Estimation under uncontrolled and controlled communications in networked control systems. In *Proceedings of the 44th IEEE Conference on Decision and Control*, pages 842–847. IEEE, 2005.
- Z. Yiming, Y. Xianglong, G. Xishan, Z. Mingang, and W. Liren. A design of greenhouse monitoring & control system based on zigbee wireless sensor network. In *International Conference on Wireless Communications, Networking and Mobile Computing, 2007. WiCom 2007.*, pages 2563–2567. IEEE, 2007.
- S. Yuksel and T. Başar. Minimum rate coding for lti systems over noiseless channels. IEEE Transactions on Automatic Control, 51(12):1878–1887, 2006.
- J. Zhang, K. Tan, J. Zhao, H. Wu, and Y. Zhang. A practical SNR-guided rate adaptation. In *INFOCOM 2008. The 27th Conference on Computer Communications.*, pages 2083–2091, 2008.
- L. Zhang and D. Hristu-Varsakelis. Communication and control co-design for networked control systems. Automatica, 42(6):953–958, 2006.
- W. Zhang. Stabilization of networked control systems over a sharing link using ALOHA. In *Proc.* of the 42nd IEEE Conference on Decision and Control (CDC), volume 1, pages 204–209, 2003.
- W. Zhang, M. Branicky, and S. Phillips. Stability of networked control systems. *IEEE Control Systems Magazine*, 21(1):84–99, 2001.
- D. Zheng, M.-O. Pun, W. Ge, J. Zhang, and H. V. Poor. Distributed opportunistic scheduling for ad hoc communications with imperfect channel information. *Wireless Communications, IEEE Transactions on*, 7(12):5450–5460, 2008.